VICTORY ENERGY CORP Form PRER14A October 13, 2017

SCHEDULE 14A INFORMATION

Proxy Statement Pursuant to Section 14(a) of the Securities Exchange Act of 1934 (Amendment No. 1)

Filed by the Registrant x
Filed by a Party other than the Registrant

Check the appropriate box:

x Preliminary Proxy Statement

- "Confidential, for Use of the Commission Only (as Permitted by Rule 14a-6(e)(2))
- " Definitive Proxy Statement
- " Definitive Additional Materials
- "Solicitation Material Pursuant to Rule 14a-11(c) or rule 14a-12

VICTORY ENERGY CORPORATION

(Name of Registrant as Specified in its Charter)

(Name of Person(s) Filing Proxy Statement, if Other Than the Registrant)

Payment of Filing Fee (Check the appropriate box):

- x No fee required.
- "Fee computed on table below per Exchange Act Rules 14a-6(i)(1) and 0-11.
- 1) Title of each class of securities to which transaction applies:
- 2) Aggregate number of securities to which transaction applies:
- 3) Per unit price or other underlying value of transaction computed pursuant to Exchange Act Rule 0-11:
- 4) Proposed maximum aggregate value of transaction
- 5) Total fee paid:
- .. Fee paid previously with preliminary materials.

Check box if any part of the fee is offset as provided by Exchange Act Rule 0-11(a)(2) and identify the filing for "which the offsetting fee was paid previously. Identify the previous filing by registration statement number, or the Form or Schedule and the date of its filing.

- (1) Amount Previously Paid:
- (2) Form, Schedule or Registration Statement No.:
- (3) Filing Party:
- (4) Date Filed:

VICTORY ENERGY CORPORATION 3355 Bee Caves Road, Suite 608 Austin, Texas 78746

NOTICE OF SPECIAL MEETING OF STOCKHOLDERS

To be held on [], 2017

To the Stockholders of Victory Energy Corporation:

Notice is hereby given that a special meeting of the stockholders of Victory Energy Corporation (the "Company") will be held on [], 2017 at 10:30 a.m. Central Standard Time at the Company's office located at 3355 Bee Caves Road, Suite 608, Austin, Texas 78746. At the special meeting you will be asked to consider and vote upon the following proposals:

To approve an amendment and restatement of the Company's Amended and Restated Articles of Incorporation to, among other things, (i) increase the number of authorized shares of the common stock, \$0.001 par value per share,

- 1. of the Company from 47,500,000 shares to 300,000,000 shares and increase the number of authorized shares of the preferred stock, \$0.001 par value per share, of the Company from 2,500,000 shares to 10,000,000 shares; and (ii) implement a 1-for-38 reverse split of the Company's outstanding common stock.
- To approve the divestiture of all of the Company's partnership interests in Aurora Energy Partners, pursuant to the
- 2. terms and conditions of a divestiture agreement, dated August 21, 2017, between the Company and Navitus Energy Group, as amended.
- $3.\,\mathrm{To}$ approve the Victory Energy Corporation 2017 Equity Incentive Plan.
 - To elect seven (7) director nominees to the Company's board of directors to serve until the next annual meeting of
- 4. the Company's stockholders or until their successors are elected and qualify, subject to their prior death, resignation or removal.
- 5. To approve the adjournment of the special meeting for any purpose, including to solicit additional proxies if there are insufficient votes at the time of the special meeting to approve the proposals described above.

The board unanimously recommends a vote "FOR" the amendment and restatement of the Company's Amended and Restated Articles of Incorporation, "FOR" the divesture under the divestiture agreement, "FOR" approval of the Victory Energy Corporation 2017 Equity Incentive Plan, "FOR" each of the director nominees, and "FOR" the proposal to approve the adjournment of the special meeting for any purpose, including to solicit additional proxies if there are insufficient votes at the time of the meeting to approve the proposals mentioned above.

Stockholders of record of the Company's common stock at the close of business on [], 2017 will be entitled to notice of, and are cordially invited to attend, the special meeting and to attend any adjournment or postponement thereof. However, to assure your representation at the meeting, please vote your proxy via the internet, by telephone, or by completing, dating, signing and returning the enclosed proxy. Even if you have previously submitted your proxy, you may choose to vote in person at the meeting. Whether or not you expect to attend the meeting, please read the proxy statement and then promptly vote your proxy in order to ensure your representation at the meeting.

You may cast your vote by visiting www.proxyvote.com. You may also have access to the materials for the annual meeting by visiting the website: www.proxyvote.com.

Each share of common stock entitles the holder thereof to one vote.

You are urged to review carefully the information contained in the enclosed proxy statement prior to deciding how to vote your shares. This notice and the attached proxy statement are first being disseminated to stockholders on or about [], 2017.

BY ORDER OF THE BOARD OF DIRECTORS,

/s/ Kenneth Hill Kenneth Hill Chief Executive Officer

IF YOU RETURN YOUR PROXY CARD WITHOUT AN INDICATION OF HOW YOU WISH TO VOTE, YOUR SHARES WILL BE VOTED IN FAVOR OF EACH PROPOSAL.

Important Notice Regarding the Availability of Proxy Materials for the Special Meeting to be Held on [], 2017: This Proxy Statement is available at: www.proxyvote.com.

TABLE OF CONTENTS

INTRODUCTION1

QUESTIONS AND ANSWERS ABOUT THE SPECIAL MEETING AND THE PROPOSALS3

When and where will the Meeting take place?

What proposals are the stockholders being asked to consider?3

What are the recommendations of the Board?3

What is the Record Date for the Meeting?3

Who can vote at the Meeting?3

What is the proxy card?3

What is the difference between holding shares as a stockholder of record and as a beneficial owner?3

What is the quorum required for the Meeting?4

Assuming that a quorum is present, what vote is required to approve the proposals to be voted upon at the Meeting?4

How do I vote?4

What are the effects of not voting or abstaining? What are the effects of broker non-votes?5

What does it mean if I received more than one proxy card?5

What happens if I don't indicate how to vote my proxy?5

What happens if I sell my shares after the record date but before the Meeting?5

What if I change my mind after I return my proxy?5

Who can help answer my other questions?6

SECURITY OWNERSHIP OF CERTAIN BENEFICIAL OWNERS AND MANAGEMENT7

DIRECTORS AND EXECUTIVE OFFICERS9

Directors, Director Nominees and Executive Officers9

Family Relationships 11

Involvement in Certain Legal Proceedings11

CORPORATE GOVERNANCE13

Governance Structure 13

The Board's Role in Risk Oversight13

Independent Directors13

Audit Committee 13

Compensation Committee 14

Director Nominations15

Stockholder Communications with the Board of Directors 15

Code of Ethics 15

EXECUTIVE COMPENSATION16

Summary Compensation Table - Fiscal Years Ended December 31, 2016 and 201516

Outstanding Equity Awards Value at Fiscal Year-End Table 17

<u>Director Compensation</u>17

TRANSACTIONS WITH RELATED PERSONS 18

SECTION 16(A) BENEFICIAL OWNERSHIP REPORTING COMPLIANCE19

PROPOSAL NO. 1 – CHARTER PROPOSAL 20

Overview20

Reasons for the Amendments to our Current Charter 20

Potential Effects of the Proposed Reverse Stock Split23

Other Possible Effects of the Amendments to our Current Charter 23

No Dissenters' Rights24

Vote Required24

PROPOSAL NO. 2 – DIVESTITURE PROPOSAL25

Introduction25

Summary of Terms of the Divestiture 25

Contact Information of the Parties 26

Business Conducted by the Parties26

Background of the Divestiture 27

Reasons for the Divestiture 28

Activities of our Company Following the Divestiture 28

Terms of the Divestiture Agreement 29

Tax Treatment30

Regulatory Approvals 30

Reports, Opinions and Appraisals 30

Pro Forma Information31

Requirements for Stockholder Approval31

Interests of Certain Persons in the Divestiture Proposal31

No Dissenters' Rights31

Vote Required31

PROPOSAL NO. 3 – 2017 EQUITY INCENTIVE PLAN32

Overview32

Significant Features of the Plan33

New Plan Benefits37

Vote Required37

PROPOSAL NO. 4 - DIRECTOR PROPOSAL 38

Director Nominees38

Vote Required38

PROPOSAL NO. 5 – ADJOURNMENT PROPOSAL39

TRANSACTION OF OTHER BUSINESS40

HOUSEHOLDING OF PROXY STATEMENT40

WHERE YOU CAN FIND MORE INFORMATION40

ANNEX A - FORM OF AMENDED AND RESTATED ARTICLES OF INCORPORATION

ANNEX B - DIVESTITURE AGREEMENT

ANNEX C - VICTORY ENERGY CORPORATION 2017 EQUITY INCENTIVE PLAN

ANNEX D - UNAUDITED PRO FORMA FINANCIAL STATEMENTS

ANNEX E - FINANCIAL STATEMENTS

VICTORY ENERGY CORPORATION

3355 Bee Caves Road, Suite 608

Austin, Texas 78746

PROXY STATEMENT

FOR

SPECIAL MEETING OF STOCKHOLDERS

To be held on [], 2017

INTRODUCTION

This proxy statement is furnished in connection with the solicitation of proxies by the Board of Directors (the "Board") of Victory Energy Corporation ("we," "us," "our" or "our company") for use at the special meeting of stockholders of the Company and at all adjournments and postponements thereof (the "Meeting"). The Meeting will be held on [], 2017 at 10:30 a.m. Central Standard Time at our office located at 3355 Bee Caves Road, Suite 608, Austin, Texas 78746 for the following purposes:

1. To approve an amendment and restatement of our Amended and Restated Articles of Incorporation to, among other things, (i) increase the number of authorized shares of our common stock, \$0.001 par value per share, from

47,500,000 shares to 300,000,000 shares and increase the number of authorized shares of our preferred stock, \$0.001 par value per share, from 2,500,000 shares to 10,000,000 shares; and (ii) implement a 1-for-38 reverse split of our outstanding common stock (the "Charter Proposal").

To approve the divestiture of all of our partnership interests in Aurora Energy Partners, a Texas general partnership

- 2. ("Aurora"), pursuant to the terms and conditions of a divestiture agreement, dated August 21, 2017 (the "Divestiture Agreement"), between Navitus Energy Group ("Navitus") and our company, as amended (the "Divestiture Proposal").
- 3. To approve the Victory Energy Corporation 2017 Equity Incentive Plan (the "Plan").
- To elect seven (7) director nominees to our board of directors to serve until the next annual meeting of our stockholders or until their successors are elected and qualify, subject to their prior death, resignation or removal.
- 5. To approve the adjournment of the Meeting for any purpose, including to solicit additional proxies if there are insufficient votes at the time of the Meeting to approve the proposals described above (the "Adjournment Proposal"). Our board unanimously recommends a vote "FOR" the Charter Proposal, "FOR" the Divestiture Proposal, "FOR" approval of the Plan, "FOR" each of the director nominees, and "FOR" the Adjournment Proposal.

Stockholders of record of our common stock at the close of business on [], 2017 (the "Record Date") will be entitled to notice of, and are cordially invited to attend, the Meeting and to attend any adjournment or postponement thereof. However, to assure your representation at the Meeting, please vote your proxy via the internet, by telephone, or by completing, dating, signing and returning the enclosed proxy. Even if you have previously submitted your proxy, you may choose to vote in person at the Meeting. Whether or not you expect to attend the Meeting, please read the proxy statement and then promptly vote your proxy in order to ensure your representation at the Meeting.

This proxy solicitation is being made and paid for by our company on behalf of our board. In addition, we may retain a third party proxy solicitor for which we may incur fees. Our directors, officers and employees may also solicit proxies by personal interview, mail, e-mail, telephone, facsimile or other means of communication. These persons will not be paid any additional compensation for their efforts. We will also request brokers and other fiduciaries to forward proxy solicitation material to the beneficial owners of shares of our common stock that the brokers and fiduciaries hold of record. Upon request, we will reimburse them for their reasonable out-of-pocket expenses. In addition, we will indemnify our proxy solicitor against any losses arising out of that firm's proxy soliciting services on our behalf. None of the proposals included in this proxy statement has been approved or disapproved by the Securities and Exchange Commission (the "SEC"), and the SEC has not passed upon the fairness or merits of any proposals nor upon the accuracy or adequacy of the information contained in this proxy statement. Any representation to the contrary is unlawful.

OUESTIONS AND ANSWERS ABOUT THE SPECIAL MEETING AND THE PROPOSALS

The following are some questions that you, as a stockholder of our company, may have regarding the Meeting and the proposals and brief answers to such questions. We urge you to carefully read this entire proxy statement, the annexes to this proxy statement and the documents referred to or incorporated by reference in this proxy statement because the information in this section does not provide all the information that may be important to you as a stockholder of our company with respect to the proposals. See "Where You Can Find More Information."

When and where will the Meeting take place?

The Meeting will be held on [], 2017 at 10:30 a.m. Central Standard Time at our office located at 3355 Bee Caves Road, Suite 608, Austin, Texas 78746.

What proposals are the stockholders being asked to consider?

At the Meeting, you will be asked to vote upon the Charter Proposal, the Divestiture Proposal, the Plan, the director nominees, and the Adjournment Proposal.

What are the recommendations of the Board?

Our Board has approved the five proposals and unanimously recommends that the stockholders vote "FOR" each proposal, including "FOR" each director nominee.

What is the Record Date for the Meeting?

Holders of our common stock as of the close of business on [], 2017, the Record Date for the Meeting, are entitled to notice of, and to vote at, the Meeting and any postponements or adjournments of the Meeting.

Who can vote at the Meeting?

Stockholders who owned shares of our common stock, Series C Preferred Stock and Series D Preferred Stock on the Record Date may attend and vote at the Meeting. There were 31,220,326 shares of common stock, 800,000 shares of our non-voting Series B Convertible Preferred Stock, 180,000 shares of our Series C Preferred Stock and 20,000 shares of our Series D Preferred Stock outstanding on the Record Date. All shares of common stock have one vote per share and vote together as a single class. No shares of the Series B Convertible Preferred Stock may vote upon any proposal. Shares of our Series C Preferred Stock and Series D Preferred Stock vote with the shares of common stock as a single class on an as-converted to common stock basis. As of the Record Date, the outstanding shares of Series C Preferred Stock were convertible, subject to approval of the Charter Proposal, into 35,730,241 shares of common stock and the outstanding shares of Series D Preferred Stock were convertible, subject to approval of the Charter Proposal, into 9,508,075 shares of common stock.

Information about the stockholdings of our directors and executive officers is contained in the section of this proxy statement entitled "Security Ownership of Certain Beneficial Owners and Management."

What is the proxy card?

The proxy card enables you to appoint Mr. Kenneth Hill, our Chief Executive Officer, as your representative at the Meeting. By completing and returning the proxy card as described herein, you are authorizing this person to vote your shares at the Meeting in accordance with your instructions on the proxy card. This way, your shares will be voted whether or not you attend the Meeting. Even if you plan to attend the Meeting, we think that it is a good idea to complete and return your proxy card before the Meeting date just in case your plans change. If a proposal comes up for vote at the Meeting that is not on the proxy card, the proxies will vote your shares, under your proxy, according to Mr. Hill's best judgment.

What is the difference between holding shares as a stockholder of record and as a beneficial owner? Most of our stockholders hold their shares in an account at a brokerage firm, bank or other nominee holder, rather than holding share certificates in their own name. As summarized below, there are some distinctions between shares held of record and those owned beneficially.

Stockholder of Record

If, on the Record Date, your shares were registered directly in your name with our transfer agent, Transfer Online, Inc., you are a "stockholder of record" who may vote at the Meeting, and we are sending these proxy materials directly to you. As the stockholder of record, you have the right to direct the voting of your shares by returning the enclosed proxy card to us, by voting online or to vote in person at the Meeting. Whether or not you plan to attend the Meeting, please complete, date and sign the enclosed proxy card or vote online to ensure that your vote is counted. Beneficial Owner

If, on the Record Date, your shares were held in an account at a brokerage firm or at a bank or other nominee holder, you are considered the beneficial owner of shares held "in street name," and these proxy materials are being forwarded to you by your broker or nominee who is considered the stockholder of record for purposes of voting at the Meeting. As the beneficial owner, you have the right to direct your broker on how to vote your shares and to attend the Meeting. However, since you are not the stockholder of record, you may not vote these shares in person at the Meeting unless you receive a valid proxy from your brokerage firm, bank or other nominee holder. To obtain a valid proxy, you must make a special request of your brokerage firm, bank or other nominee holder. If you do not make this request, you can still vote by using the voting instruction card enclosed with this proxy statement; however, you will not be able to vote in person at the Meeting.

What is the quorum required for the Meeting?

The representation in person or by proxy of holders of at least a majority of the issued and outstanding shares of our stock entitled to vote at the Meeting is necessary to constitute a quorum for the transaction of business at the Meeting. Assuming that a quorum is present, what vote is required to approve the proposals to be voted upon at the Meeting? The approval of the Charter Proposal requires the affirmative vote of the holders of a majority of the issued and outstanding shares of our capital stock entitled to vote at the Meeting.

The approval of the Divestiture Proposal requires the affirmative vote of a majority of the shares present in person or represented by proxy at the Meeting.

The approval of the Plan requires the affirmative vote of a majority of the shares present in person or represented by proxy at the Meeting.

The election of each director nominee requires the affirmative vote of a plurality of votes of the shares cast at the election.

The Meeting may be adjourned by the affirmative vote of a majority of the shares present in person or represented by proxy at the Meeting.

How do I vote?

Stockholders have four voting options. You may vote using one of the following methods:

Internet. You can vote over the Internet by accessing the website at www.proxyvote.com, and following the instructions on the website. Internet voting is available 24 hours a day until 11:59 p.m. Eastern Time on the day before the Meeting. If you vote over the Internet, do not return your proxy card.

Telephone. If you hold shares directly in your own name and are the holder of record, you can vote by telephone by calling the toll-free number 800-690-6903 in the United States, Canada or Puerto Rico on a touch-tone phone. You will then be prompted to enter the control number printed on your proxy card and to follow the subsequent instructions. Telephone voting is available 24 hours a day until 11:59 p.m. Eastern Time on the day before the Meeting. If, however, you hold the shares through a broker and not in your own name, then follow the specific instructions included in your proxy materials, including the specific phone number to use to vote your shares by phone.

Mail. You can vote by mail by simply completing, signing, dating and mailing your proxy card in the postage-paid envelope included with this proxy statement.

In Person. You may come to the Meeting and cast your vote there. The Board recommends that you vote by proxy even if you plan to attend the Meeting. If your shares of common stock are held in a stock brokerage account or through a bank, broker or other nominee, or, in other words, in "street name", and you wish to vote in person at the Meeting, you must bring a letter from your bank, broker or nominee identifying you as the beneficial owner of the shares and authorizing you to vote such shares at the Meeting.

What are the effects of not voting or abstaining? What are the effects of broker non-votes?

If you do not vote by virtue of not being present in person or by proxy at the Meeting, your shares will not be counted for purposes of determining the existence of a quorum.

Abstentions will be counted for the purpose of determining the existence of a quorum, however, they will not be considered in determining the number of votes cast. Accordingly, an abstention will have no effect on the Divestiture Proposal, approval of the Plan, the election of directors, or the Adjournment Proposal, but will be treated in the same manner as a vote against the Charter Proposal.

Broker non-votes occur on a matter when a broker is not permitted to vote on that matter without instructions from the beneficial owner and instructions are not given. These matters are referred to as "non-routine" matters. Broker non-votes will be counted for the purpose of determining the existence of a quorum, however the Charter Proposal, the Divestiture Proposal, approval of the Plan, and the election of directors are "non-routine" matters. Thus, in tabulating the voting result for these proposals, shares that constitute broker non-votes are not considered votes cast on those proposals.

What does it mean if I received more than one proxy card?

If your shares are registered differently or in more than one account, you will receive more than one proxy card. Sign and return all proxy cards to ensure that all of your shares are voted.

What happens if I don't indicate how to vote my proxy?

If you just sign your proxy card without providing further instructions, your shares will be counted as a vote "for" each of the proposals.

What happens if I sell my shares after the record date but before the Meeting?

The Record Date for the Meeting is earlier than the date of the Meeting. If you transfer your shares after the Record Date but before the date of the Meeting, you will retain your right to vote at the Meeting (provided that such shares remain outstanding on the date of the Meeting).

What if I change my mind after I return my proxy?

You may revoke your proxy and change your vote at any time before the polls close at the Meeting. You may do this by:

sending a written notice to our Corporate Secretary, Kenneth Hill, stating that you would like to revoke your proxy of a particular date;

signing another proxy card with a later date and returning it before the polls close at the Meeting; voting again at a later time, but prior to the date of the Meeting, via the Internet or telephone; attending the Meeting and voting in person.

Please note, however, that if your shares are held of record by a brokerage firm, bank or other nominee, you must instruct your broker, bank or other nominee that you wish to change your vote by following the procedures on the voting form provided to you by the broker, bank or other nominee. If your shares are held in street name, and you wish to attend and vote at the Meeting, you must bring to the Meeting a legal proxy from the broker, bank or other nominee holding your shares, confirming your beneficial ownership of the shares and giving you the right to vote your shares. Simply attending the Meeting will not constitute a revocation of your proxy.

Who can help answer my other questions?

If you have more questions about the proposals or how to submit your proxy, or if you need additional copies of this proxy statement or the enclosed proxy card or voting instructions, please contact Victory Energy Corporation, Attn: Corporate Secretary, Kenneth Hill, 3355 Bee Caves Road, Suite 608, Austin, Texas 78746, telephone number (512) 347-7300.

SECURITY OWNERSHIP OF CERTAIN BENEFICIAL OWNERS AND MANAGEMENT

The following table sets forth information regarding beneficial ownership of each class of our voting stock as of the Record Date (i) by each person who is known by us to beneficially own more than 5% of our voting stock; (ii) by each of our officers, directors and director nominees; and (iii) by all of our officers and directors as a group. Unless otherwise specified, the address of each of the persons set forth below is in care of our company, 3355 Bee Caves Road, Suite 608, Austin, Texas 78746.

	Amount of Beneficial Ownership ⁽¹⁾			Percent		Percent			Percent	
Name and Address of Beneficial Owner	Common Stock	Series C Preferred Stock	Series D Preferred Stock	of Comm Stock ⁽²	2)	C	ed	of Series D Preferred Stock ⁽⁴⁾	of To	g
Kenneth Hill, CEO and Director (6)	2,420,020	2,519	0	7.43	%	1.39	%	*	3.75	%
Ronald Zamber, Director (7)	23,361,928	158,590	0	48.59	%	88.11	%	*	58.77	%
Robert Grenley, Director (8)	863,434	0	0	2.73	%	*		*	1.12	%
Ricardo A. Salas, Director	0	0	0	*		*		*	*	
Kevin DeLeon, Director (9)	400,000	0	0	1.27	%	*		*	*	
Julio C. Herrera, Director	0	0	0	*		*		*	*	
Eric Eilertsen, Director Nominee	0	0	0	*		*		*	*	
All directors and officers as a group (7 persons named above)	27,045,382	161,108	0	53.80	%	89.50	%	*	61.80	%
David McCall (10)	1,262,253	0	20,000	3.99	%	*		100 %	14.02	%
Greg Johnson (11)	874,986	18,892	0	2.80	%	10.50	%	*	6.05	%
* I ass then 10%										

^{*} Less than 1%

Beneficial Ownership is determined in accordance with the rules of the SEC and generally includes voting or investment power with respect to securities. Each of the beneficial owners listed above has direct ownership of and sole voting power and investment power with respect to the shares of our common stock. For each beneficial owner above, any options exercisable within 60 days have been included in the denominator.

- (2) Based on 31,220,326 shares of our common stock outstanding as of the Record Date.
- (3) Based on 180,000 shares of our Series C Preferred Stock outstanding as of the Record Date. Shares of Series C Preferred Stock are, upon the occurrence of certain events, convertible into shares of common stock on the basis of 198.501338888889 shares of common stock for each share of Series C Preferred Stock (subject to adjustment), which will be adjusted to a conversion ratio of 5.22371944444444 shares of common stock for each share of Series

C Preferred Stock upon the effectiveness of a planned 1-for-38 reverse split of our outstanding common stock. Holders of Series C Preferred Stock vote with the holders of common stock on all matters on an as-converted to common stock basis.

Based on 20,000 shares of our Series D Preferred Stock outstanding as of the Record Date. Shares of Series D Preferred Stock are, upon the occurrence of certain events, convertible into shares of common stock on the basis of

- 475.40375 shares of common stock for each share of Series D Preferred Stock (subject to adjustment), which will be adjusted to a conversion ratio of 12.510625 shares of common stock for each share of Series D Preferred Stock upon the effectiveness of a planned 1-for-38 reverse split of our outstanding common stock. Holders of Series D Preferred Stock vote with the holders of common stock on all matters on an as-converted to common stock basis.
- (5) Percentage of Total Voting Stock represents total ownership with respect to all shares of our common stock, Series C Preferred Stock and Series D Preferred Stock, as a single class and on an as-converted to common stock basis. Includes 1,053,020 shares of common stock, options for the purchase of 1,025,000 shares of common stock exercisable within 60 days, warrants for the purchase of 342,000 shares of common stock exercisable within 60
- (6) days and 2,518.87469776652 shares of Series C Preferred Stock held by Mrs. Hill, wife of Mr. Hill. Does not include options for the purchase of 7,000,000 shares of common stock which may vest more than 60 days after the Record Date.

Includes 1,489,850 shares of common stock, 48,589.0929199162 shares of Series C Preferred Stock and warrants for the purchase of 922,000 shares of common stock exercisable within 60 days held by Dr. Zamber; 2,468,138 shares of common stock owned by Visionary Investments, LLC, of which Dr. Zamber is sole member; 2,437,481 shares of common stock, 110,000.472149068 shares of Series C Preferred Stock and warrants for the purchase of 5,203,252 shares of common stock exercisable within 60 days owned by Visionary Private Equity Group I, LP, of which Dr. Zamber is senior managing director of its general partner, Visionary PE GP I, LLC; 105,860 shares of common stock and warrants for the purchase of 104,846 shares of common stock exercisable within 60 days owned by James Capital Consulting, LLC, of which Dr. Zamber is the managing member; and warrants for the purchase of 10,630,501 shares of common stock exercisable within 60 days owned by Navitus Energy Group, LP, of which James Capital Consulting, LLC is the managing partner.

Includes 441,434 shares of common stock and warrants for the purchase of 422,000 shares of common stock exercisable within 60 days.

(7)

(9) Includes options for the purchase of 400,000 shares of common stock exercisable within 60 days.

Includes 735,020 shares of common stock and warrants for the purchase of 382,000 shares of common stock exercisable within 60 days owned by Mr. McCall, 145,233 shares of common stock owned by 1519 Partners

(10) LLC, of which Mr. McCall is the controlling partner, and 20,000 shares of Series D Preferred Stock held by McCall Law Firm, of which Mr. McCall is the managing partner.

[11] Includes 874,986 shares of common stock and 18,891.5602332489 shares of Series C Preferred Stock held by Mr. Johnson.

On August 21, 2017, we entered into a transaction agreement (the "Transaction Agreement") with Armacor Victory Ventures, LLC, a Delaware limited liability company ("Armacor"), pursuant to which Armacor (i) granted to us a worldwide, perpetual, royalty free, fully paid up and exclusive sublicense to all of Armacor's owned and licensed intellectual property for use in the Oilfield Services Business (as defined in the Transaction Agreement), and (ii) agreed to contribute \$5,000,000 to us, in exchange for which we issued 800,000 shares of our non-voting Series B Convertible Preferred Stock. Upon completion of certain funding conditions set forth in the Transaction Agreement, including stockholder approval of the Charter Proposal, and payment of the aforementioned cash contribution, each share of Series B Convertible Preferred Stock plus accrued, but unpaid, dividends thereon shall be automatically converted into such number of fully paid and non-assessable shares of our common stock as is determined by dividing the Stated Value by the Conversion Price in effect on such conversion date. The "Stated Value" is initially \$122.6628243 per share, subject to appropriate adjustment in the event of any stock dividend, stock split, combination or other similar recapitalization with respect to the Series B Convertible Preferred Stock, and the "Conversion Price" is initially equal to \$0.04, subject to adjustment. Such shares are therefore convertible into 2,453,256,486 shares of common stock, or 64,559,381 shares of common stock following the planned 1-for-38 reverse stock split. The conversion of such shares will result in a change of control of our company.

DIRECTORS AND EXECUTIVE OFFICERS

Directors, Director Nominees and Executive Officers

Following is information about our directors, director nominees and executive officers as of the date of this proxy statement.

NAME AGE POSITION

Kenneth Hill 53 Chief Executive Officer, President, Chief Financial Officer and Director

Ronald W. Zamber 56 Chairman of the Board of Directors

Robert Grenley 59 Director Ricardo A. Salas 53 Director Kevin DeLeon 50 Director Julio C. Herrera 50 Director

Eric Eilertsen 60 Director Nominee

Pursuant to the Transaction Agreement that we entered into with Armacor on August 21, 2017, we agreed to include in this proxy statement a proposal for the election of four (4) directors designated by Armacor, including Ricardo A. Salas, Kevin DeLeon, Julio C. Herrera and Eric Eilertsen. We also agreed that until the next annual meeting of stockholders following this Meeting, the board shall remain at seven (7) persons and Navitus shall have the right to appoint three (3) members to the board, who shall initially be Ronald W. Zamber, Robert Grenley and Kenneth Hill. Kenneth Hill – Chief Executive Officer, President, Chief Financial Officer and Director.

Mr. Hill has served as our Chief Executive Officer since January 2012 and as our Chief Financial Officer since April 2016. He has been a member of our board of directors since April 2011. Mr. Hill previously served as our Vice President and Chief Operating Officer from January 2011 to January 2012. Prior to joining our company, Mr. Hill held titles of Interim CEO, VP of Operations and VP of Investor Relations for the U.S. subsidiary of Austin Exploration, Ltd, a publicly traded oil and gas company on the Australian Stock Exchange. Since 2001, through his private company, Mr. Hill has raised several million dollars of venture capital and he personally invested in and consulted for a number of successful entrepreneurial ventures across a variety of industries, including oil and gas. Prior to 2001, Mr. Hill was employed for 16 years at Dell, Inc. As one of the first 20 employees at Dell he served in a variety of management positions including manufacturing, sales, marketing, and business development. Prior to joining Dell, Mr. Hill studied Business Management and Business Marketing at Southwest Texas State University (now Texas State University). While at Dell, Mr. Hill continued his education at The University of Texas Graduate School of Business Executive Education program, The Aspen Institute and the Center for Creative Leadership. Mr. Hill was selected to serve on our board of directors due to his extensive senior management and venture capital experience across a variety of industries, including oil and gas, for over 25 years.

Ronald W. Zamber, M.D. Director - Chairman of the Board

Dr. Zamber has served as a member of our board of directors since January 24, 2009. Dr. Zamber is founder, Managing Director and Chairman of Visionary Private Equity Group since 2010, and a Managing Director of Navitus since 2011, Navitus Partners since 2011 and James Capital Energy since 2007. He brings more than 20 years of experience in corporate management and business development extending across the public, private and non-profit arenas. Dr. Zamber has helped build profitable companies in healthcare, private and public petroleum E&P, consumer products and Internet technology industries. Dr. Zamber is a Board Certified Ophthalmologist and founder of International Vision Quest, a non-profit organization that performs humanitarian medical and surgical missions, builds water treatment facilities and supports food delivery programs to impoverished communities around the world. He has served as an examiner with the American Board of Ophthalmologists and Secretariat for State Affairs with the American Academy of Ophthalmology. Dr. Zamber is the 2009 recipient of Notre Dame's prestigious Harvey Foster Humanitarian Award. He now serves on the advisory board of Feed My Starving Children, one of the highest rated and fastest growing charities in the country. Dr. Zamber received his Bachelor's degree with high honors from the University of Notre Dame and his medical degree with honors from the University of Washington. Dr. Zamber was

selected to serve on our board of directors due to his over 20 years of experience in corporate management and business development extending across the public, private and non-profit arenas.

Robert Grenley – Director

Mr. Grenley has served as a member of our board of directors since June 1, 2010. Mr. Grenley has over 25 years of experience in financial management, business development and entrepreneurial experience. This financial experience includes 12 years managing early stage organizations with equity capital. Mr. Grenley's broader financial management experience includes over 10 years of direct portfolio management and investment expertise including common and preferred stock, stock options, corporate and municipal bonds as well as syndicated investments and private placements. Recently, Mr. Grenley has been associated with the Visionary Private Equity Group since 2012, and is currently its Director of Capital Development, as well as the Chief Financial Officer of the Visionary Media Group, a wholly owned subsidiary. Mr. Grenley served as the Chief Financial Officer of POP Gourmet, a fast growing Seattle-based snack food company, since early 2013, where he was responsible for the creation, production, and execution of POP Gourmet's first equity financing (\$2.5 million in 2013), its second equity financing (\$8.5 million in 2015), and its first credit facility (\$2 million in 2015). As the company has matured, it has been able to attract a consumer product group specialist as Chief Financial Officer, and Mr. Grenley currently retains the Director, Corporate Finance title, focusing on credit facilities, investor relations, and other related matters. Mr. Grenley holds a BA in Economics from Duke University. Mr. Grenley was selected to serve on our board of directors due to his over 25 years of experience in financial management, business development and entrepreneurial experience. Ricardo A. Salas - Director

Mr. Salas has served as a member of our board of directors since August 21, 2017. He has served as the President of Armacor Holdings, LLC, an investment holding company for Liquidmetal Coatings, LLC, which develops, supplies and provides application service of leading metallic coatings which protect against wear and corrosion in oil & gas, power, pulp & paper and other industrial environments, since May of 2012. He has served as a Director of Liquidmetal Coatings, LLC since June 2007. Between 2008 and 2015, Mr. Salas served as Executive Vice President and a Director of Liquidmetal Technologies, Inc., a pioneer in developing and commercializing a family of amorphous metal alloys. In 2001, he founded and became CEO of iLIANT Corporation, a health care information technology and outsourcing service provider. Following iLIANT's merger with MED3000 Group, Inc., he continued to serve as a Director of MED3000 Group, Inc. and on its Special Committee leading up to its sale to McKesson Corporation in December of 2012. He serves as a Director of Advantum Health, a private equity backed healthcare IT enabled services company. Mr. Salas received an Economics degree from Harvard College in 1986. Mr. Salas was selected to serve on our board of directors due to his extensive management experience.

Kevin DeLeon - Director

Mr. DeLeon has served as a member of our board of directors since August 21, 2017. He has served as a General Partner and Director of Corporate Strategy for Visionary Private Equity Group, a private equity firm that invests in early stage, high growth companies, since 2015. Mr. DeLeon has spent more than twenty-five years in global finance, both on the buy and sell side, in New York, London, and Tokyo. For the past decade, his focus has been in natural resources, most recently as Senior Advisor to our company since February, 2015. Prior to joining our company, he served in the same capacity at Miller Energy, a NYSE-listed Alaska focused oil and gas exploration and production company, from June 2013 to February 2015. At Miller, Mr. DeLeon was responsible for overseeing corporate strategy, with particular focus on financing the company's drilling program and acquisitions, as well as investor relations and corporate governance. Prior to Miller, Mr. DeLeon spent approximately six years spearheading the U.S. operations for a boutique U.K. investment bank, with a strong focus in E&P and metals & mining. Early in his career, he worked for Yamaichi, one of the Big Four Japanese securities houses, where he received the Chairman's award for his consistent revenue contributions. Mr. DeLeon was also a founding partner of Bracken Partners, a London-based corporate finance advisory and fund management firm with particular focus on the U.K. private equity markets. He has served as both a senior executive and non-executive director of numerous public and private U.K. and U.S. companies. Mr. DeLeon is a 1990 graduate of Yale University, with a B.A in Economics. Mr. DeLeon was selected to serve on our board of directors due to his extensive global finance experience.

Julio C. Herrera – Director

Mr. Herrera has served as a member of our board of directors since August 21, 2017. Mr. Herrera started his oil and gas career in 1988 with ExxonMobil in Colombia, and then joined British Petroleum in 1992. His 29 years in the

energy industry include positions with responsibilities over Alaska, Texas, Wyoming, Angola, Azerbaijan, Scotland, Egypt, Colombia, Mexico, Brazil and the rest of Latin America. At British Petroleum, Mr. Herrera held various executive finance and Chief Financial Officer positions across local, national and international divisions. Upon his departure from British Petroleum in December 2014, he founded JC Proenergy LLC, which provided finance, control and private equity consulting services to energy and mining management teams, in January 2015. Upon his departure from JC Proenergy LLC in October 2015, he served as a Senior Infrastructure Director for the BlackRock-Pemex joint venture from November 2015 until April 2016. In May 2016, he became the Group Chief Financial Officer of Lewis Energy Group, an oil and gas company based in San Antonio, Texas. Mr. Herrera has a Masters' Degree in Certified Public Accounting from the Pontificia Universidad Javeriana in Bogotá, Colombia and has completed post-graduate studies in Marketing at the Kellogg School of Management and CFO Excellence at the Wharton School of Business. Mr. Herrera was selected to serve on our board of directors due to his extensive finance experience in the energy industry.

Eric Eilertsen – Director Nominee

Our board has recommended that stockholders elect Mr. Eilertsen to our board at the Meeting. During his 35-year business career, Mr. Eilertsen has served as chief executive officer, chief operating officer or corporate consultant for a wide range of companies across the U.S., Canada, China and Europe, including several Fortune 100 companies, Since November 2016, Mr. Eilertsen has served as the Chief Executive Officer for Experience Matters Consortium, Inc., an Arizona nonprofit organization with the mission to connect skilled volunteer talent with meaningful service opportunities across the many nonprofit and civic organizations in Arizona. From November 2013 to October 2016, Mr. Eilertsen served as the Co-Founder and President of SmartStory.com, a Silicon Valley startup he helped fund to create the next generation of customer, employee, and patient engagement/e-commerce platform. Before co-founding SmartStory.com, he founded ETI Consulting in March 1997, where he provided C-level advice on commercializing and productizing unique intellectual property developed by Kroger Co.'s technology group. Mr. Eilertsen is currently serving on the Arizona Department of Education Career & Technical Education Ouality Commission, the Arizona State University Community Council and the board of directors for the Global Pathways Institute. Mr. Eilertsen earned a Bachelor's degree in Business Administration from the University of Mississippi in Oxford and began his professional career in 1979 with The Procter and Gamble Company located in Cincinnati, Ohio. He has held various NASD and SEC licenses and has served as a director on multiple boards since 1985. Mr. Eilertsen was selected to serve on our board of directors due to his extensive management experience. Our directors currently have terms which will end at our next annual meeting of the stockholders or until their successors are elected and qualify, subject to their prior death, resignation or removal. Members of our board of directors are encouraged to attend meetings of the board of directors and the annual meetings of stockholders. The board of directors held 2 meetings and executed 1 unanimous written consent in lieu of a meeting in 2016. Family Relationships

There are no family relationships among any of our officers or directors.

Involvement in Certain Legal Proceedings

To the best of our knowledge, none of our directors or executive officers has, during the past ten years: been convicted in a criminal proceeding or been subject to a pending criminal proceeding (excluding traffic violations and other minor offences);

had any bankruptcy petition filed by or against the business or property of the person, or of any partnership, corporation or business association of which he was a general partner or executive officer, either at the time of the bankruptcy filing or within two years prior to that time;

been subject to any order, judgment, or decree, not subsequently reversed, suspended or vacated, of any court of competent jurisdiction or federal or state authority, permanently or temporarily enjoining, barring, suspending or otherwise limiting, his involvement in any type of business, securities, futures, commodities, investment, banking, savings and loan, or insurance activities, or to be associated with persons engaged in any such activity; been found by a court of competent jurisdiction in a civil action or by the SEC or the Commodity Futures Trading

Commission to have violated a federal or state securities or commodities law, and the judgment has not been reversed, suspended, or vacated;

been the subject of, or a party to, any federal or state judicial or administrative order, judgment, decree, or finding, not subsequently reversed, suspended or vacated (not including any settlement of a civil proceeding among private

litigants), relating to an alleged violation of any federal or state securities or commodities law or regulation, any law or regulation respecting financial institutions or insurance companies including, but not limited to, a temporary or permanent injunction, order of disgorgement or restitution, civil money penalty or temporary or permanent cease-and-desist order, or removal or prohibition order, or any law or regulation prohibiting mail or wire fraud or fraud in connection with any business entity; or

been the subject of, or a party to, any sanction or order, not subsequently reversed, suspended or vacated, of any self-regulatory organization (as defined in Section 3(a)(26) of the Exchange Act (15 U.S.C. 78c(a)(26))), any registered entity (as defined in Section 1(a)(29) of the Commodity Exchange Act (7 U.S.C. 1(a)(29))), or any equivalent exchange, association, entity or organization that has disciplinary authority over its members or persons associated with a member.

CORPORATE GOVERNANCE

Governance Structure

We chose to appoint a separate chairman of our board of directors who is not our Chief Executive Officer. Our board of directors has made this decision based on their belief that an independent chairman of the board can act as a balance to the Chief Executive Officer, who also serves as a non-independent director.

The Board's Role in Risk Oversight

Our board of directors administers its risk oversight function as a whole by making risk oversight a matter of collective consideration. While management is responsible for identifying risks, our board of directors has charged the audit committee of the board of directors with evaluating financial and accounting risk and the compensation committee of the board of directors with evaluating risks associated with employees and compensation. Investor-related risks are usually addressed by the board as a whole. We believe an independent chairman of the board adds an additional layer of insight to our board of directors' risk oversight process.

Independent Directors

In considering and making decisions as to the independence of each of the directors of our company, the board considered transactions and relationships between our company (and its subsidiaries) and each director (and each member of such director's immediate family and any entity with which the director or family member has an affiliation such that the director or family member may have a material indirect interest in a transaction or relationship with such entity). The board has determined that the following directors and director nominees are independent as defined in applicable SEC and NASDAQ rules and regulations, and that each constitutes an "Independent Director" as defined in NASDAQ Marketplace Rule 5605: Julio C. Herrera and Eric Eilertsen.

Audit Committee

On September 14, 2017, our board of directors established an audit committee to assist it in fulfilling its responsibilities for general oversight of our accounting and financial reporting processes, audits of our financial statements, and internal control and audit functions. The audit committee is responsible for, among other things: appointing, evaluating and determining the compensation of our independent auditors;

establishing policies and procedures for the review and pre-approval by the audit committee of all auditing services and permissible non-audit services (including the fees and terms thereof) to be performed by the independent auditor; reviewing with our independent auditors any audit problems or difficulties and management's response;

reviewing and approving all proposed related-party transactions, as defined in Item 404 of Regulation S-K under the Securities Act of 1933, as amended;

discussing our financial statements with management and our independent auditors;

reviewing and discussing reports from the independent auditor on critical accounting policies and practices used by our company and alternative accounting treatments;

reviewing major issues as to the adequacy of our internal controls and any special audit steps adopted in light of significant internal control deficiencies;

reviewing and discussing with management our major financial risk exposures and the steps management has taken to monitor and control such exposures;

meeting separately and periodically with management and our internal and independent auditors; reviewing matters related to the corporate compliance activities of our company;

•

reviewing and approving our code of ethics, as it may be amended and updated from time to time, and reviewing reported violations of the code of ethics;

annually reviewing and reassessing the adequacy of our audit committee charter; and

such other matters that are specifically delegated to our audit committee by our board from time to time.

The audit committee works closely with management as well as our independent auditors. The audit committee has the authority to obtain advice and assistance from, and receive appropriate funding from us for, outside legal, accounting or other advisors as the audit committee deems necessary to carry out its duties.

Our board of directors has adopted a written charter for the audit committee that meets the applicable standards of the SEC and NASDAQ. The proposed members of the audit committee are Ronald W. Zamber, Julio C. Herrera and Eric Eilertsen. Eric Eilertsen will serve as the chair of the audit committee.

Our board of directors determined that each member of the audit committee satisfies the independence and other composition requirements of the SEC and NASDAQ. Our board has determined that Julio C. Herrera qualifies as an "audit committee financial expert" under Item 407(d)(5) of Regulation S-K and has the requisite accounting or related financial expertise required by applicable NASDAQ rules.

Compensation Committee

On September 14, 2017, our board of directors established a compensation committee to discharge our board's responsibilities relating to compensation of our Chief Executive Officer and other executive officers and provide general oversight of compensation structure. Other specific duties and responsibilities of the compensation committee include:

reviewing and approving objectives relevant to executive officer compensation;

evaluating performance and recommending to the board of directors the compensation, including any incentive compensation, of our Chief Executive Officer and other executive officers in accordance with such objectives; reviewing and approving compensation packages for new executive officers and termination packages for executive officers;

recommending to the board of directors the compensation for our directors;

administering our equity compensation plans and other employee benefit plans;

reviewing periodic reports from management on matters relating to our personnel appointments and practices; evaluating periodically the compensation committee charter;

such other matters that are specifically delegated to our compensation committee by our board from time to time. Our board of directors has adopted a written charter for the compensation committee. The proposed members of the compensation committee are Ronald W. Zamber, Julio C. Herrera and Eric Eilertsen. Dr. Zamber will serve as the chair of the compensation committee. Our board of directors determined that each member of the compensation committee satisfies the independence requirements of NASDAQ.

The compensation committee reviews executive compensation from time to time and reports to the board of directors, which makes all final decisions with respect to executive compensation.

Director Nominations

We currently do not have a standing nominating committee or committee performing similar functions. Our entire board of directors undertakes the functions that would otherwise be undertaken by a nominating committee.

Our board utilizes a variety of methods for identifying and evaluating nominees for our directors. Our board regularly assesses the appropriate size of our board and whether any vacancies on the board are expected due to retirement or other circumstances. When considering potential director nominees, the board considers the candidate's character, judgment, diversity, age, skills, including financial literacy and experience in the context of the needs of our company and of our existing directors. The board also seeks director nominees who are from diverse backgrounds and who possess a range of experiences as well as a reputation for integrity. The board considers all of these factors to ensure that our board as a whole possesses a broad range of skills, knowledge and experience useful to the effective oversight and leadership of our company.

Our board does not have a specific policy with regard to the consideration of candidates recommended by stockholders, however any nominees proposed by our stockholders will be considered on the same basis as nominees proposed by the board. If you or another stockholder want to submit a candidate for consideration to the board, you may submit your proposal to our Corporate Secretary, Kenneth Hill in accordance with the stockholder communication procedures set forth below.

All director nominees included in this proxy statement were approved by our board.

Stockholder Communications with the Board of Directors

Our board of directors has established a process for stockholders to communicate with the board of directors or with individual directors. Stockholders who wish to communicate with our board of directors or with individual directors should direct written correspondence to Kenneth Hill, Corporate Secretary, at Kenny@vyey.com or to the following address (our principal executive offices): Board of Directors, c/o Corporate Secretary, 3355 Bee Caves Road, Suite 608, Austin, Texas 78746.

The Corporate Secretary will forward such communications to our board of directors or the specified individual director to whom the communication is directed unless such communication is unduly hostile, threatening, illegal or similarly inappropriate, in which case the Corporate Secretary has the authority to discard the communication or to take appropriate legal action regarding such communication.

Code of Ethics

We have adopted a code of ethics that applies to all of our directors, officers and employees, including our principal executive officer, principal financial officer and principal accounting officer. Such code of ethics addresses, among other things, honesty and ethical conduct, conflicts of interest, compliance with laws, regulations and policies, including disclosure requirements under the federal securities laws, and reporting of violations of the code.

We are required to disclose any amendment to, or waiver from, a provision of our code of ethics applicable to our principal executive officer, principal financial officer, principal accounting officer, controller, or persons performing similar functions. We intend to use our website as a method of disseminating this disclosure, as permitted by applicable SEC rules. Any such disclosure will be posted to our website within four business days following the date of any such amendment to, or waiver from, a provision of our code of ethics.

EXECUTIVE COMPENSATION

Summary Compensation Table - Fiscal Years Ended December 31, 2016 and 2015

The following table sets forth information concerning all cash and non-cash compensation awarded to, earned by or paid to the named persons for services rendered in all capacities during the noted periods. No other executive officers received total annual salary and bonus compensation in excess of \$100,000.

Name and Principal Position	Year	Salary (\$)	Stock Awards (\$) ⁽¹⁾	Option Awards (\$) ⁽¹⁾	Total (\$)
Kenneth Hill,	2016	229,167	-	-	229,167
Chief Executive Officer and Chief Financial Officer	2015	215,625	121,850	100,050	437,525
Fred J. Smith, Jr.,	2016	60,000	-	-	60,000
Former Chief Financial Officer	2015	172,500	-	_	172,500

These amounts shown represent the aggregate grant date fair value for stock awards, options and warrants granted to the named executive officers computed in accordance with FASB ASC Topic 718. Assumptions used in the

(1) calculation of these amounts are included in "Note 7 – Stockholders' Equity" to our audited financial statements for the fiscal year ended December 31, 2015 included in our annual report on Form 10-K for the year ended December 31, 2016.

On August 21, 2017, we entered into an amended and restated employment agreement with Mr. Kenneth Hill. Under the amended and restated employment agreement, we agreed to pay Mr. Hill a salary of \$250,000 per year, and he will be eligible for annual bonuses at the discretion of our board. In addition, we agreed to grant Mr. Hill an option to purchase 7,500,000 shares of our common stock, which option has an exercise price of \$0.04 per share and vests in 36 equal monthly installments. Mr. Hill will also be eligible to participate in the standard benefits plans offered to similarly situated employees by us from time to time, subject to plan terms and our generally applicable policies. The term of the amended and restated employment agreement is for three (3) years and automatically renews for additional one-year periods unless terminated. Either party may terminate the amended and restated employment agreement at any time upon at least 30 days written notice (other than a termination by us for Cause).

If Mr. Hill's employment is terminated by us (other than as a result of death, disability or Cause (as defined in the amended and restated employment agreement)) or is terminated by Mr. Hill for Good Reason (as defined in the amended and restated employment agreement), Mr. Hill is entitled to the following: (i) a lump sum payment in an amount equal to product of (a) the base salary and bonus paid during the immediately preceding twelve month period ending on the date of termination of employment, multiplied by (b) fifty percent (50%) (notwithstanding the foregoing, if termination is within 6 months before or 24 months following the occurrence of a Change of Control (as defined in the amended and restated employment agreement), such lump sum payment shall be equal to product of (a) the base salary and the maximum bonus under the amended and restated employment agreement, multiplied by (b) one hundred percent (100%), and all stock options shall immediately vest); (ii) all base salary earned, accrued or owing through the date his employment is terminated but not yet paid; and (iii) continued participation in all employee benefit plans, programs or arrangements available to our executives in which Mr. Hill was participating on the date of termination until the earliest of (but only to the extent the terms of such benefit plans, programs or arrangements permit such continued participation): (a) the second anniversary of the date of termination of employment, provided that if termination of employment is within 6 months before or 24 months following the occurrence of a Change of Control, then he shall be entitled to continue to participate in such employee benefit plans, programs or arrangements until the third anniversary of the date of termination of employment; (b) the date the amended and restated employment agreement would have expired (with timely written notice) but for the occurrence of the date of termination; or (c) the date, or dates, Mr. Hill receives coverage and benefits under the plans, programs and arrangements of a subsequent employer (such coverages and benefits to be determined on a coverage-by- coverage, or benefit-by-benefit, basis).

On May 27, 2014, we entered into an employment agreement with Mr. Fred J. Smith, Jr. The term of the employment agreement began on June 2, 2014 and ended upon his resignation effective April 15, 2016. Mr. Smith received a base annual salary of \$180,000 per year and he participated in our employee benefit plans made available to our executive officers generally. Mr. Smith's medical health insurance was paid by us for a period of 60 days following his resignation.

During fiscal years 2016 and 2015, we made the following grants of awards under our 2014 Long Term Incentive Plan to the named executive officers:

On March 31, 2015, Mr. Hill was granted an award of 20,000 shares of our common stock which vested on the date of grant.

On August 28, 2015, Mr. Hill was granted an award of 340,000 shares of our common stock which vested on the date of grant.

On August 28, 2015, Mr. Hill was granted an option to purchase 375,000 shares of our common stock at an exercise price of \$0.27, which was the fair market value as of the date of grant. The option vests in three equal installments on each anniversary of the date of grant and terminates on August 28, 2025.

On October 1, 2015, Mr. Hill was granted an award of 60,000 shares of our common stock which vested on the date of grant.

On December 31, 2015, Mr. Hill was granted an award of 60,000 shares of our common stock which vested on the date of grant.

Outstanding Equity Awards Value at Fiscal Year-End Table

The following table includes certain information with respect to the value of all unexercised options and unvested shares of restricted stock previously awarded to the executive officers named above at the fiscal year ended December 31, 2016.

Option Awards

Name	Number of Securities Underlying Unexercised Options (#) Exercisable		Equity Incentive Plan Awards: Number of Securities Underlying Unexercised Unearned Options (#)	Option	Option Expiration Date
Kenneth	60,000	-	-	\$1.00	8/27/2017
Hill	150,000	-	-	\$0.35	4/23/2024
ПШ	125,000	250,000	-	\$0.27	8/28/2025

Director Compensation

No member of our board of directors received any compensation for his services as a director during the fiscal year ended December 31, 2016.

TRANSACTIONS WITH RELATED PERSONS

The following includes a summary of transactions since the beginning of our 2015 fiscal year, or any currently proposed transaction, in which we were or are to be a participant and the amount involved exceeded or exceeds the lesser of \$120,000 or one percent of the average of our total assets at year-end for the last two completed fiscal years, and in which any related person had or will have a direct or indirect material interest (other than compensation described under "Executive Compensation"). We believe the terms obtained or consideration that we paid or received, as applicable, in connection with the transactions described below were comparable to terms available or the amounts that would be paid or received, as applicable, in arm's-length transactions

We have incurred legal fees with McCall Law Firm. David McCall, our former director until August 21, 2017, is partner in McCall Law Firm. The fees are attributable to litigation involving our oil and natural gas operations in Texas. As of June 30, 2017 and December 31, 2016, the Company owed The McCall Firm \$527,098, and \$503,377, respectively.

On August 21, 2017, we entered into a settlement agreement and mutual release with McCall Law Firm, pursuant to which all obligations of our company to McCall Law Firm to repay indebtedness for borrowed money, which totaled approximately \$380,323, was converted into 20,000 shares of our Series D Preferred Stock.

During the year ended December 31, 2016, the temporary capital advances totaling \$130,000 had been made by Navitus. James Capital Consulting, LLC is the Managing Partner of Navitus and Dr. Ronald Zamber, the chairman of our board of directors, is the Managing Member of James Capital Consulting, LLC.

On August 21, 2017, we entered into the Divestiture Agreement with Navitus, pursuant to which we agreed to divest and transfer our 50% ownership interest in Aurora to Navitus, which currently owns the remaining 50% interest, and issue 166,549,134 shares of our common stock (or 4,382,872 shares following the planned 1-for-38 reverse stock split) to Navitus in consideration for a release from Navitus of all of our obligations under the second amended partnership agreement, dated October 1, 2011, between us and Navitus. See "Proposal No. 2 – Divestiture Proposal" for more information regarding the Divestiture Agreement.

On August 21, 2017, we entered into a settlement agreement and mutual release with Messrs. Ronald Zamber and Greg Johnson (affiliate of Navitus), pursuant to which all obligations of our company to Messrs. Zamber and Johnson to repay indebtedness for borrowed money, which totaled approximately \$520,800, was converted into 65,591.4971298402 shares of our Series C Preferred Stock, 46,699.9368965913 shares of which were issued to Dr. Zamber and 18,891.5602332489 shares of which were issued to Mr. Johnson.

On August 21, 2017, we entered into a settlement agreement and mutual release with Dr. Zamber and Mrs. Kim Rubin Hill, the wife of Kenneth Hill, our Chief Executive Officer, pursuant to which all obligations of our company to Dr. Zamber and Mrs. Hill to repay indebtedness for borrowed money, which totaled approximately \$35,000, was converted into 4,408.03072109141 shares of our Series C Preferred Stock, 1,889.1560233248900 shares of which were issued to Dr. Zamber and 2,518.8746977665200 shares of which were issued to Mrs. Hill.

During the six months ended June 30, 2017, temporary capital advances totaling \$135,000 had been made by Visionary Private Equity Group I, LP ("VPEG"). Visionary PE GP I, LLC is the general partner of VPEG and Dr. Zamber is the Managing Director Visionary PE GP I, LLC.

On February 3, 2017, we completed a private placement, pursuant to which VPEG purchased a unit comprised of \$320,000 principal amount of a 12% unsecured six-month promissory note and a common stock purchase warrant to purchase 5,203,252 shares of our common stock at an exercise price of \$0.0923 per share.

On August 21, 2017, we entered into a settlement agreement and mutual release with VPEG, pursuant to which all obligations of our company to VPEG to repay indebtedness for borrowed money (other than the secured convertible original issue discount promissory note described below), which totaled approximately \$873,409.64, was converted into 110,000.472149068 shares of our Series C Preferred Stock. Some of the obligations to VPEG arose pursuant to the private placement note described above. Pursuant to the settlement agreement and mutual release, the 12% unsecured six-month promissory note was repaid in full and terminated, but VPEG retained the common stock purchase warrant.

On August 21, 2017, we entered into a loan agreement with VPEG, which was amended on October 11, 2017. Pursuant to the loan agreement, as amended, VPEG loaned \$565,000 to our company. Such loan is evidenced by a secured convertible original issue discount promissory note issued by our company to VPEG on August 21, 2017 and amended on October 11, 2017. The note reflects an original issue discount of \$56,500 such that the principal amount of the note is \$621,500, notwithstanding the fact that the loan is in the amount of \$565,000. The note does not bear any interest in addition to the original issue discount, provided that, upon the occurrence of an event of default, interest shall begin to accrue at a rate equal to the lesser of 8% per annum or the maximum interest rate allowed from time to time under applicable law. The note matures on November 30, 2017, is secured by a security interest in all of our assets, and is convertible at VPEG's option into shares of our common stock at a conversion price equal to \$0.04 per share, subject to adjustment.

SECTION 16(A) BENEFICIAL OWNERSHIP REPORTING COMPLIANCE

Section 16(a) of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), requires our directors and executive officers and beneficial holders of more than 10% of our common stock to file with the SEC initial reports of ownership and reports of changes in ownership of our equity securities. We believe, based solely on a review of the copies of such reports furnished to us and representations of these persons that all reports needed to be filed have been filed for the year ended December 31, 2016.

PROPOSAL NO. 1 - CHARTER PROPOSAL

Overview

On September13, 2017, our board of directors approved an amendment and restatement of our current Amended and Restated Articles of Incorporation (the "Current Charter") to, among other things, increase the number of authorized shares of our common stock and preferred stock and implement a 1-for-38 reverse split of our outstanding common stock. The full text of the proposed Amended and Restated Articles of Incorporation (the "Restated Charter") is attached hereto as Annex A.

Our Current Charter authorizes our company to issue 47,500,000 shares of common stock, \$0.001 par value per share, and 2,500,000 shares of preferred stock, \$0.001 par value per share. As of the Record Date, there were 31,220,326 shares of common stock, 800,000 shares of our non-voting Series B Convertible Preferred Stock (convertible only following stockholder approval and under certain circumstances into 2,453,256,486 shares of common stock), 180,000 shares of our Series C Preferred Stock (convertible only following stockholder approval and under certain circumstances into 35,730,241 shares of common stock) and 20,000 shares of our Series D Preferred Stock outstanding (convertible only following stockholder approval and under certain circumstances into 9,508,075 shares of common stock). We have also reserved 20,883,278 shares of common stock for issuance under outstanding warrants, 8,495,000 shares of common stock for issuance under outstanding options, 138,000 shares of common stock for issuance to holders of our Series A Preferred Stock who are in the process of converting their shares, and 570,000,000 shares of common stock for issuance under the Plan. The proposed Restated Charter will increase the number of authorized shares of common stock to 300,000,000 shares and increase the number of authorized shares of preferred stock to 10,000,000 shares. The additional shares of preferred stock so authorized will be undesignated shares.

In addition, the proposed Restated Charter would implement a 1-for-38 reverse split of our outstanding common stock.

The proposed Restated Charter will also include the following amendments to our Current Charter:

the addition of a provision regarding our election not to be governed by certain provisions of the Nevada Revised Statutes regulating control share acquisitions;

the removal of a provision regarding the number, election and removal of our directors, which is included in our Amended and Restated Bylaws; and

the removal of a provision regarding our officers, which is included in our Amended and Restated Bylaws.

The Restated Charter will become effective upon filing with the Nevada Secretary of State's Office and we anticipate that such filing will occur promptly following stockholder approval at the Meeting.

Reasons for the Amendments to our Current Charter

Increase in Authorized Stock

The primary purpose of the amendment to increase our authorize stock is to facilitate the conversion of our recently issued preferred stock pursuant to the transactions described below and the issuance of shares pursuant to the Divestiture Agreement described under "Proposal No. 2 – Divestiture Proposal" below.

On August 21, 2017, we entered into the Transaction Agreement with Armacor, pursuant to which Armacor (i) granted to us a worldwide, perpetual, royalty free, fully paid up and exclusive sublicense to all of Armacor's owned and licensed intellectual property for use in the Oilfield Services Business (as defined in the Transaction Agreement), and (ii) agreed to contribute \$5,000,000 to us, in exchange for which we issued 800,000 shares of our Series B Convertible Preferred Stock, Upon completion of certain funding conditions set forth in the Transaction Agreement, including stockholder approval of this Charter Proposal, and payment of the afore-mentioned cash contribution, each share of Series B Convertible Preferred Stock plus accrued, but unpaid, dividends thereon, shall be automatically converted into such number of fully paid and non-assessable shares of our common stock as is determined by dividing the Stated Value by the Conversion Price in effect on such conversion date. The "Stated Value" is initially \$122.6628243 per share, subject to appropriate adjustment in the event of any stock dividend, stock split, combination or other similar recapitalization with respect to the Series B Convertible Preferred Stock, and the "Conversion Price" is initially equal to \$0.04, subject to adjustment. Such shares are therefore convertible into 2,453,256,486 shares of common stock, or 64,559,381 shares of common stock following the 1-for-38 reverse stock split. On August 21, 2017, we also entered into settlement agreements and mutual releases with VPEG and certain other persons related to our company, pursuant to which we issued an aggregate of 180,000 shares of our Series C Preferred Stock to such persons. See "Transactions with Related Persons" for more information about these agreements. Upon completion of certain funding conditions set forth in the Transaction Agreement, including stockholder approval of this Charter Proposal, each share of Series C Preferred Stock plus accrued, but unpaid, dividends thereon, shall be automatically converted into such number of fully paid and non-assessable shares of our common stock as is determined by dividing the Stated Value by the Conversion Price in effect on such conversion date. The "Stated Value" is initially \$7.9400535555556 per share, subject to appropriate adjustment in the event of any stock dividend, stock split, combination or other similar recapitalization with respect to the Series C Preferred Stock, and the "Conversion Price" is initially equal to \$0.04, subject to adjustment. Such shares are therefore convertible into an aggregate of 35,730,241 shares of common stock, or 940,270 shares following the 1-for-38 reverse stock split. On August 21, 2017, we entered also into a settlement agreement and mutual release with McCall Law Firm, pursuant to which we issued 20,000 shares of our Series D Preferred Stock to McCall Law Firm. See "Transactions with Related Persons" for more information about this agreement. Shares of Series D Preferred Stock are only convertible (after stockholder approval of this Charter Proposal) if we fail to make certain redemption payments on the outstanding Series D Preferred Stock. In the event of such conversion, each share of Series D Preferred Stock plus accrued, but unpaid, dividends thereon, is convertible into such number of fully paid and non-assessable shares of our common stock as is determined by dividing the Stated Value by the Conversion Price in effect on such conversion date. The "Stated Value" is initially \$19.01615 per share, subject to appropriate adjustment in the event of any stock dividend, stock split, combination or other similar recapitalization with respect to the Series D Preferred Stock, and the "Conversion Price" is initially equal to \$0.04, subject to adjustment. Such shares are therefore convertible into an aggregate of 9,508,075 shares of common stock, or 250,213 shares following the 1-for-38 reverse stock split. Following the conversion of our recently issued preferred stock pursuant to the transactions described above and the issuance of shares pursuant to the Divestiture Agreement described under "Proposal No. 2 – Divestiture Proposal" below, and, assuming that the full cash contribution is made by Armacor as contemplated by the Transaction Agreement, Armacor will own 90% of our issued and outstanding common stock on a fully-diluted basis and our business will focus exclusively on technology-driven, friction reducing oilfield products and services under our exclusive sublicense to all of Armacor's intellectual property, rather than on the ownership of exploration and production assets. Our board felt that the transition of our company from an exploration and production company to a technology-driven oilfield services company utilizing the Armacor sublicense is in the best interests of our company for the following reasons, among others:

We believe that the expected \$5 million investment from Armacor and the Divestiture of our exploration and production operations with the associated divestiture of related liabilities will transition our company from one with a going concern risk to a more stable and better capitalized enterprise that will be in a position to utilize the Armacor sublicense to grow.

Unlike upstream exploration and production companies, as an oilfield services enterprise, we will be far less sensitive to volatility in commodity prices, which has negatively affected our company over the past few years.

We believe that we will be able to gain and retain a market position in the oilfield services industry and that

• such position will be protected by our perpetual, royalty free, exclusive sublicense and existing, as well as growing, patented product offerings under that sublicense.

Although the primary purpose of this amendment is to facilitate these transactions, this amendment will also provide us with increased flexibility in meeting future corporate needs and requirements by providing additional authorized shares of common stock and preferred stock, which will be available for issuance from time to time as determined by the board for any proper corporate purpose, including additional equity financings, without the expense and delay associated with a special stockholders' meeting, except where required by applicable rules, regulations and laws. Reverse Stock Split

The primary purpose of the reverse stock split is also to facilitate the conversion of our recently issued preferred stock pursuant to the transactions described above and the issuance of shares pursuant to the Divestiture Agreement described under "Proposal No. 2 – Divestiture Proposal" below. Absent a reverse stock split, the conversion and issuance of these shares would require us to issue 2,665,043,936 shares of our common stock.

Our board also believes that the reverse stock split will enhance the acceptability and marketability of our common stock to the financial community and the investing public and may mitigate any reluctance on the part of certain brokers and investors to trade in our common stock. Many institutional investors have policies prohibiting them from holding stocks in their own portfolios which trade at prices below certain levels. These policies reduce the number of potential investors in our common stock at its current market price. In addition, analysts at many leading brokerage firms are reluctant to recommend stocks to their clients, or monitor the activity of stocks, that trade at a price per share below certain levels. A variety of brokerage house policies and practices also tend to discourage individual brokers within those firms from dealing in stocks that trade at a price per share below certain levels. Some of those policies and practices pertain to the payment of brokers' commissions and to time-consuming procedures that function to make the handling of such stocks unattractive to brokers from an economic standpoint. Additionally, because brokers' commissions on such stocks generally represent a higher percentage of the stock price than commissions on higher-priced stocks, the current share price of our common stock can result in an individual stockholder paying transaction costs that represent a higher percentage of total share value than would be the case if our share price were higher. This factor may also limit the willingness of institutions to purchase our stock.

No Restrictions on Control Share Acquisitions

The proposed Restated Charter also amends our Current Charter to state that our company elects not to be governed by the terms and provisions of Sections 78.378 through 78.3793, inclusive, of the Nevada Revised Statutes (referred to herein as the "Control Share Statue"), restricting certain acquisitions of a controlling interest in a corporation. The Control Share Statute applies only to Nevada corporations with at least 200 stockholders of record, including at least 100 stockholders of record who are Nevada residents, and which conduct business directly or indirectly in Nevada. The Control Share Statute prohibits an acquiror, under certain circumstances, from voting its shares of a target corporation's stock after crossing certain ownership threshold percentages, unless the acquiror obtains approval of the target corporation's disinterested stockholders. The Control Share Statute specifies three thresholds: one-fifth or more but less than one-third, one-third but less than a majority, and a majority or more, of the outstanding voting power. Once an acquiror crosses one of the above thresholds, those shares in an offer or acquisition, and acquired within 90 days thereof, become "control shares" and such control shares are deprived of the right to vote until disinterested stockholders restore the right. The Control Share Statute also provides that if control shares are accorded full voting rights and the acquiring person has acquired a majority or more of all voting power, all other stockholders who do not vote in favor of authorizing voting rights to the control shares are entitled to demand payment for the fair value of their shares in accordance with statutory procedures established for dissenters' rights.

As of the date of this proxy statement, we do not have 100 or more stockholders of record who are Nevada residents, and are therefore not subject to the Control Share Statute, although there can be no assurance that in the future this

statute will not apply to us. Our board of directors has determined that, should we become subject to the Control Share Statute, it would place unnecessary burdens on our company in connection with the completion of third party financings, and has thus decided to amend our Current Charter to specifically forego the provisions of the Control Share Statue.

Removal of Duplicative Provisions

The proposed Restated Charter also amends our Current Charter to remove certain provisions regarding the size of our board, the method for the election and removal of directors, the method for filling vacancies in our board, and the officers of our company. These provisions are not required to be included in our articles of incorporation under Nevada law and are also included in our Amended and Restated Bylaws, so our board determined to remove these provisions from our Current Charter.

Potential Effects of the Proposed Reverse Stock Split

The immediate effect of a reverse stock split would be to reduce the number of shares of our common stock and to increase the trading price of our common stock. However, we cannot predict the specific effect of any reverse stock split upon the market price of our common stock. Based on the data we have reviewed leading up to this Charter Proposal, it appears that in some cases a reverse stock split improves stock performance while in other cases it does not, and in some cases a reverse stock split improves overall market capitalization while in other cases it does not. There is no assurance that the trading price of our common stock after the reverse stock split will rise in proportion to the reduction in the number of shares of our common stock outstanding as a result of the reverse stock split. Also, there is no assurance that a reverse stock split will lead to a sustained increase in the trading price of our common stock. The trading price of our common stock may change due to a variety of factors, such as our operating results and other factors related to our business and general market conditions.

As a summary and for illustrative purposes only, the following tables reflects the approximate number of shares of our common stock that would be outstanding as a result of the reverse stock split, without accounting for fractional shares, which will be rounded up to the nearest whole share:

Common Stock

Shares Outstanding Percentage Reduction Shares to Be Outstanding

31,220,326 97.37% 821,588

Fully-Diluted Common Stock

Shares Outstanding Percentage Reduction Shares to Be Outstanding

2,559,231,406 97.37% 67,348,195

Our stockholders should recognize that if a reverse stock split is effected, they will own a smaller number of shares than they currently own (approximately equal to the number of shares owned immediately prior to the reverse stock split divided by 38 and after giving effect to the rounding up of fractional shares to the nearest whole share). The reverse stock split would not affect any stockholder's percentage ownership interests in our company or such stockholder's proportionate voting power, except to the extent that interests in fractional shares would be rounded up to the nearest whole share.

Other Possible Effects of the Amendments to our Current Charter

Release No. 34-15230 of the staff of the SEC requires disclosure and discussion of the effects of any stockholder proposal that may be used as an anti-takeover device. Please see "Reasons for the Amendments to our Current Charter—No Restrictions on Control Share Acquisitions" above for certain anti-takeover effects of our election not to be governed by the Control Share Statute.

The proposed increase in the authorized number of shares of our common stock and preferred stock, and the reverse split, could have an anti-takeover effect, in that additional shares could be issued (within the limits imposed by applicable law) in one or more transactions that could make a change in control or takeover of us more difficult. For example, additional shares could be issued by us so as to dilute the stock ownership or voting rights of persons seeking to obtain control of us, even if the persons seeking to obtain control offers an above-market premium that is favored by a majority of the independent stockholders. Similarly, the issuance of additional shares to certain persons

allied with our management could have the effect of making it more difficult to remove our current management by diluting the stock ownership or voting rights of persons seeking to cause such removal. We have no plans or proposals to adopt other provisions or enter into other arrangements that may have material anti-takeover consequences. This proposal is not being presented with the intent that it be utilized as a type of anti-takeover device.

Our stockholders should recognize that, as a result of this proposal, they will own a fewer percentage of shares with respect to our total authorized shares than they presently own and will be diluted as a result of any issuance of shares by us in the future.

Other than in connection with the conversion of preferred stock described above and the issuance of shares pursuant to the Divestiture Agreement described under "Proposal No. 2 – Divestiture Proposal" below, there are currently no specific plans, arrangements, commitments or understandings for the issuance of the additional shares of stock which are proposed to be authorized (except with respect to potential issuances of shares upon exercise of currently outstanding options and warrants).

No Dissenters' Rights

Under Nevada law, holders of our common stock are not entitled to dissenter's rights of appraisal with respect to the approval of the proposed Restated Charter.

Vote Required

The approval of the Charter Proposal requires the affirmative vote of the holders of a majority of the issued and outstanding shares of our capital stock entitled to vote at the Meeting, which includes shares of our common stock, Series C Preferred Stock and Series D Preferred Stock, voting together as a single class on an as-converted to common stock basis. An abstention will be treated in the same manner as a vote against the Charter Proposal. Since the Charter Proposal is considered to be a "non-routine" matter, shares that constitute broker non-votes are not considered votes cast on the Charter Proposal.

Our board of directors unanimously recommends a vote "FOR" approval of the Charter Proposal.

PROPOSAL NO. 2 – DIVESTITURE PROPOSAL

Introduction

On August 21, 2017, we entered into the Divestiture Agreement with Navitus, and on September 14, 2017, we entered into Amendment No. 1 to the Divestiture Agreement. Pursuant to the Divestiture Agreement, as amended, we agreed to divest and transfer our 50% ownership interest in Aurora to Navitus, which currently owns the remaining 50% interest, in consideration for a release from Navitus of all of our obligations under the second amended partnership agreement, dated October 1, 2011, between us and Navitus (the "Partnership Agreement"), including, without limitation, obligations to return to Navitus investors their accumulated deferred capital, deferred interest and related allocations of equity. We also agreed to (i) issue 166,549,134 shares of our common stock (or 4,382,872 shares following the planned 1-for-38 reverse stock split) to Navitus and (ii) pay off or otherwise satisfy all indebtedness and other material liabilities of Aurora at or prior to closing of the Divestiture Agreement. We refer to this transaction as the "Divestiture." A copy of the Divestiture Agreement, as amended, is attached hereto as Annex B.

Summary of Terms of the Divestiture

This summary of the terms of the Divestiture highlights information included elsewhere in this proxy statement. This summary does not contain all of the information you should consider before voting on the Divestiture Proposal. You should read the entire proxy statement carefully, including the annexes attached hereto. For your convenience, we have included cross references to direct you to a more complete description of the topics described in this summary. Divestiture Agreement; Parties. On August 21, 2017, our company and Navitus entered into the Divestiture Agreement and on September 14, 2017, our company and Navitus entered into Amendment No. 1 to the Divestiture Agreement.

Divestiture. Pursuant to the Divestiture, we will divest and transfer our 50% ownership interest in Aurora to Navitus, which currently owns the remaining 50% interest, and issue 166,549,134 shares of our common stock (or 4,382,872 shares following the planned 1-for-38 reverse stock split) to Navitus in consideration for a release from Navitus of all of our obligations under the Partnership Agreement. See "The Divestiture" below.

Reasons for the Divestiture. Our board considered a number of factors before deciding to enter into the Divestiture Agreement, including, among other things, the strategic and financial benefits of divesting Aurora, the release of our company from obligations to return to Navitus investors their accumulated deferred capital, deferred interest and

related allocations of equity, the new business plan and strategy that will be undertaken by our company following the Divestiture, and the expected future business prospects of our company after we divest Aurora. Our board also considered our current financial condition. See "Reasons for the Divestiture" below.

Nature of Our Business Following the Divestiture. Following the Divestiture, our business will be primarily focused on the oilfield services industry and no longer focused on exploration and production. See "Activities of the Company Following the Divestiture" below.

Conditions to Closing. Closing of the Divestiture Agreement is subject to stockholder approval and other customary closing conditions, including, without limitation, the receipt of all authorizations, consents and approvals of all governmental authorities or agencies; the receipt of any required consents of any third parties; execution of an amendment and restatement of the Partnership Agreement; and execution of a mutual release by the parties. See "Terms of the Divestiture Agreement—Conditions to Closing" below.

Termination of the Divestiture Agreement. The Divestiture Agreement may be terminated (i) by mutual written consent, (ii) by either party if any governmental entity has issued an order or taken any other action prohibiting the transactions contemplated by Divestiture Agreement, (iii) by either party if the other party has breached its representations and warrants or any covenant or other agreement such that the closing conditions would not be satisfied; (iv) by Navitus if its conditions set forth in the Divestiture Agreement have been satisfied and our company fails to consummate the Divestiture on the date the closing should have occurred; or (v) by our company if its conditions set forth in the Divestiture Agreement have been satisfied and Navitus fails to consummate the Divestiture on the date the closing should have occurred. See "Terms of the Divestiture Agreement—Termination" below. Recommendation of our Board of Directors. Our board has determined by an unanimous vote that the Divestiture Proposal is desirable and has directed that it be submitted to our stockholders for their approval. Our board unanimously recommends that you vote "FOR" the Divestiture Proposal. You should read "Reasons for the Divestiture" for a discussion of factors that our board considered in deciding to recommend the approval of the Divestiture Proposal.

Required Vote. Approval of the Divestiture Proposal requires the affirmative vote of a majority of the shares present in person or represented by proxy at the Meeting. See "Vote Required" below.

Contact Information of the Parties Victory Energy Corporation 3355 Bee Caves Road, Suite 608 Austin, Texas 78746 (512) 347-7300

Navitus Energy Croup PO Box 10159 Fairbanks, Alaska 99710 (907) 451-7108

Business Conducted by the Parties

Victory Energy Corporation

Victory Energy Corporation (VYEY) is an Austin, Texas based publicly held oil and gas exploration and production company that is in the process of transitioning to an oilfield services business. Historically, we have been focused on the acquisition and development of unconventional resource play opportunities in the Permian Basin, the Eagle Ford shale of South Texas and other strategically important areas that offer predictable economic outcomes and long-lived reserve characteristics.

Our company was organized under the laws of the State of Nevada on January 7, 1982. We are focused on creating shareholder value by growing revenue through the acquisition and expansion of regionally based, high quality oil-field services providers who are in the major producing basins of the United States.

Currently, we utilize a team of third-party professionals on an interim as-needed basis to support both the operational and financial aspects of our company. This team includes geologists for prospect evaluation and assessment, reservoir engineering resources for the analysis of current and new properties and a finance executive for accounting, financial reporting and other back office support. Reserve reporting is performed by a third-party engineer located in Midland,

Texas. Each independent operator utilized by us also has their own array of experts tailored for the specific formations and well completion techniques of each property that we hold an interest in. We strategically utilize both internal capabilities and industry relationships to acquire non-operated, high-grade working interest positions in predictable, low-to-moderate risk oil and gas prospects, and are focused on oil and liquids rich gas.

As of June 30, 2017, we held a working interest in 30 completed wells located in Texas and New Mexico, predominantly in the Permian Basin of West Texas and the Eagle Ford area of south Texas. Our Relationship with Aurora

We are the managing partner of Aurora and we hold a fifty percent (50%) partnership interest in Aurora, which was established in January 2008. Aurora is a consolidated subsidiary with our company for financial statement purposes. We currently conduct all of our oil and natural gas operations through, and hold all of our oil and natural gas assets through, Aurora. Aurora is the record title-holder to substantially all of the oil and natural gas properties, wells and reserves referred to in this quarterly report. Through our partnership interest in Aurora, we are the beneficial owner of fifty percent (50%) of the oil and gas properties, wells and reserves held of record by Aurora.

We work together with Navitus, the other partner of Aurora, to increase proved reserves and the valuation of Aurora. Navitus Partners, LLC, a partner in Navitus, raises capital for contribution through Navitus to Aurora, the net proceeds of which are generally used to fund Aurora's operations, as well as for the potential acquisition and development of targeted oil and gas opportunities. The investors in this offering receive a ten percent (10%) preferred return through their indirect interest in Navitus for five years and one warrant to purchase one share of our common stock for every dollar invested and additional benefits. Under the terms of the offering, Navitus has the right to contribute up to \$15 million into Aurora and we are obligated to match the capital contribution amount of Navitus resulting from the offering. We are also required to match previous contributions made by Navitus. Under the agreement, we may also raise funds from other sources. Substantially all producing oil and natural gas assets are held in Aurora during the five year term of the Partnership Agreement, which ends in October 2017. As of June 30, 2017, Navitus has contributed an aggregate of \$10.6 million into Aurora.

Navitus Energy Group

Navitus is a private Texas general partnership made-up of over 150 individual sophisticated accredited investors that have provided funding to our company directly and through a private placement that began in October 2012. Background of the Divestiture

Our board and senior management have from time to time evaluated and considered a variety of strategic alternatives as part of a long-term strategy to increase stockholder value.

Since the oil and gas commodity price crash of 2014, our board and senior management have evaluated and considered a variety of strategic acquisitions and mergers as part of a long-term strategy to increase stockholder value. However, in almost every case we ran into obstacles ranging from deal valuation, the assets themselves, target entity management or other issues that made them unattractive. Since early 2016, we have, through our relationship with Visionary Private Equity Group ("Visionary") maintained an open dialogue with the founders of Liquidmetal Coatings ("LMC") and followed the marketplace acceptance of their revolutionary amorphous metal coating products that were designed specifically for the oil and gas industry. On November 16, 2016, Visionary entered into a non-disclosure agreement with consultants of LMC, and we commenced more formal discussions regarding the LMC company, their strategy, their products and the nature of their licensing structures.

Over the next few months, the parties, together with their respective advisors, continued to discuss the potential transaction and conducted due diligence. After the parties came to a mutual understanding as to the proposed material terms of the transaction, including a formula for determining the consideration (payable in equity of our company) to be paid for the contributed license and \$5 million cash contribution being made by LMC, the board began formal discussion to enter into a letter of intent with a LMC related company named Armacor Holdings, LLC. On June 9, 2017, the board adopted resolutions by written consent that, among other things, authorized management to enter into a letter of intent with Armacor Holdings, LLC. Our company executed the letter of intent on June 10, 2017. Armacor Holdings, LLC subsequently assigned its rights under the letter of intent to Armacor, a related company. On June 12, 2017, we entered into a non-disclosure agreement with Stratiqa, Inc. ("Stratiqa") an independent financial consulting firm, to perform the needed due diligence for the preparation of a valuation report for both the LMC transaction and the Divestiture.

On July 5, 2017, Stratiqa made a valuation presentation to our board and delivered a valuation report to our board of directors. Stratiqa gave a detailed presentation to the board regarding their analysis of the relative value of the license and cash being contributed by Armacor to our company and of the release of liabilities to be received by Navitus in exchange for our interest in Aurora. Stratiqa answered questions from members of the board regarding its valuation methodology, the nature of the license being granted to our company and the value of Aurora. After discussion and consideration, our board unanimously determined that the Armacor license and \$5 million cash contribution, as well as the Divestiture, offered our company the best path forward for its stockholders.

On July 14, 2017, the board adopted resolutions by written consent to authorize management to move forward with the review and final negotiations of all transaction documents for the Armacor transaction and the Divestiture. Our Board was updated informally by management on a regular basis regarding the status of the proposed transactions. During the period from July 14, 2017 through execution of the definitive agreements on August 21, 2017, we negotiated final terms of the various transaction agreements with the parties and worked with legal counsel to finalize related documents. On August 21, 2017, the board adopted resolutions by written consent to approve the final forms of all definitive agreements, including the Divestiture Agreement.

On August 21, 2017, the Divestiture Agreement was signed by all parties and was publicly disclosed on August 22, 2017.

Reasons for the Divestiture

At the time of approval, after (i) having decided to pursue a divestiture of Aurora, (ii) reviewing all material information reasonably available to it, including consideration of potential alternative courses of action, and (iii) consultation with our management and its financial and legal advisors, our board of directors determined that the Divestiture is advisable and fair to, and in the best interests of, our company and its stockholders. Our board of directors made this determination and approved the Divestiture based on its consideration of a number of factors, the material ones of which included (not necessarily in order of relative importance):

Its views on the potential value of Aurora given the liabilities associated with the ownership of Aurora and based on several different metrics, including from a net asset value perspective, and that the Divestiture represents an attractive valuation of Aurora and ability for us to eliminate liabilities and enhance our balance sheet;

Our recent discussions with advisors regarding strategic alternatives, potential acquisitions and merger and/or sale opportunities;

Its comparison of potential values, benefits, risks and uncertainties of the Divestiture to other potential courses of action, including continuing to retain and operate Aurora;

That we have obtained an exclusive sublicense from Armacor to the liquid metal technology and are changing our business plan to begin engaging in the oilfield services business such that the ownership of exploration and production assets is outside of our core business plan;

Our stockholders will continue to participate in our new oilfield services business following the completion of the Divestiture and our stockholders, therefore, will continue to retain the ability to participate in our potential growth and future prospects;

The belief that the Divestiture will benefit our company and enhance stockholder value with respect to the remaining assets of our company by, among other things: (i) placing focus on our existing assets and potential additional opportunities; and (ii) creating a stronger balance sheet through debt reduction; and

The terms and conditions of the Divestiture Agreement and the likelihood that the transactions contemplated by the Divestiture Agreement will be consummated in a timely manner.

The foregoing discussion of factors considered by our board of directors is not intended to be exhaustive, but is believed to include material factors considered in approving the Divestiture, although individual members of our board of directors may have given different weights to different factors. Based on the totality of the information presented, our board of directors collectively reached the unanimous decision to approve and declare advisable the Divestiture in light of the factors described above and other factors that the members of our board of directors felt were appropriate.

Activities of our Company Following the Divestiture

Following the Divestiture of its interests in Aurora to Navitus, we will remain a public company, which will be focused exclusively on technology-driven, friction reducing oilfield products and services. Specifically, delivering metal coating products and services that provide protection and friction reduction for nearly every metal component of

a drilling operation. Our goal is to help drillers become more efficient while creating better outcomes for every well they drill and complete.

As a well-capitalized technology-enabled oilfield services business, with ownership of a worldwide, perpetual, royalty-free, fully-paid and exclusive license and rights to all future LMC oil and gas product innovations, we believe that ongoing financing beyond the initial \$5 million investment that we expect to receive from Armacor will be more accessible. The protected intellectual property also creates a meaningfully differentiated services business, with little effective competition.

Our company will initially embark on a U.S. oilfield services company acquisition initiative, aimed at oilfield service companies already using one or more of the Armacor® brand of Liquidmetal® products and/or who are recognized as a high-quality services provider to strategic customers in the major Texas and Oklahoma basins. The acquisition of these oilfield services companies will provide immediate U.S. channel distribution for our products and increase revenue accordingly. These acquisitions, when complete, will create subsidiaries in each major basin. We intend to grow each of these company owned distribution entities by providing better access to capital, more disciplined sales and marketing development, customer needs collaboration and infrastructure build-out.

Terms of the Divestiture Agreement

On August 21, 2017, we entered into the Divestiture Agreement with Navitus, pursuant to which we agreed to divest and transfer our 50% ownership interest in Aurora to Navitus, which currently owns the remaining 50% interest, and issue 166,549,134 shares of our common stock (or 4,382,872 shares following the planned 1-for-38 reverse stock split) to Navitus in consideration for a release from Navitus of all of our obligations under the Partnership Agreement, including, without limitation, obligations to return to Navitus investors their accumulated deferred capital, deferred interest and related allocations of equity.

The Divestiture Agreement contains usual pre- and post-closing representations, warranties and covenants. In addition, Navitus agreed that our company may take any steps necessary to amend the exercise price of warrants issued to Navitus Partners, LLC to reflect an exercise price of \$0.04. We also agreed to provide Navitus with demand registration rights with respect to the shares to be issued to it under the Divestiture Agreement, whereby we agreed to, upon Navitus' request, file a registration statement on an appropriate form with the SEC covering the resale of such shares and use our commercially reasonable efforts to cause such registration statement to be declared effective within one hundred twenty (120) days following such filing.

Conditions to Closing

Closing of the Divestiture Agreement is subject to customary closing conditions and certain other specific conditions, including the following:

the issuance of 166,549,134 shares of our common stock (or 4,382,872 shares following the planned 1-for-38 reverse stock split) to Navitus;

the payment or satisfaction by our company of all indebtedness or other liabilities of Aurora, which total approximately \$1.2 million;

the receipt of any authorizations, consents and approvals of all governmental authorities or agencies;

the receipt of any required consents of any third parties;

•he execution of an amendment and restatement of the Partnership Agreement;

the execution of a mutual release by the parties; and

the execution of customary officer certificates by our company and Navitus regarding the representations, warrants and covenants contained in the Divestiture Agreement.

Closing is expected to occur shortly after approval at the Meeting.

Termination

The Divestiture Agreement may be terminated (i) by mutual written consent, (ii) by either party if any governmental entity has issued an order or taken any other action permanently enjoining, restraining or otherwise prohibiting the transactions contemplated by Divestiture Agreement, (iii) by either party if the other party has breached is representations and warrants or any covenant or other agreement to be performed by it in a manner such that the closing conditions would not be satisfied; (iv) by Navitus if (a) its conditions set forth in the Divestiture Agreement have been satisfied and our company fails to consummate the Divestiture on the date the closing should have occurred pursuant to the Divestiture Agreement and (b) Navitus has irrevocably confirmed in writing that all our conditions set forth in the Divestiture Agreement have been satisfied or that it is willing to waive all unsatisfied conditions and it

stands ready, willing and able to consummate the closing on such date; or (v) by our company if (a) its conditions set forth in the Divestiture Agreement have been satisfied and Navitus fails to consummate the Divestiture on the date the closing should have occurred pursuant to the Divestiture Agreement and (b) we have irrevocably confirmed in writing that all Navitus' conditions set forth in the Divestiture Agreement have been satisfied or that we are willing to waive all unsatisfied conditions and we stand ready, willing and able to consummate the closing on such date.

Tax Treatment

We are transferring our partnership interests in Aurora to Navitus and issuing common stock to Navitus. Our basis in these assets is being used to determine the gain. We expect the Divestiture to result in the recognition of a nearly \$400,000 gain by our company. However, it is not expected that the taxes resulting from the recognition of such gain will be material.

Regulatory Approvals

We are not aware of any federal or state regulatory requirements that must be complied with or approval that must be obtained in connection with the Divestiture.

Reports, Opinions and Appraisals

Our company engaged Stratiqa to provide a valuation presentation and report for the Divestiture to our board; however, due to our financial condition and the uncertainty of our ability to continue as a going concern, we did not engage financial and legal advisors to prepare a fairness opinion regarding the Divestiture, as such an opinion would be expensive and would strain our remaining financial resources.

As described above we engaged Stratiqa to make a presentation and deliver a valuation report to the board regarding, among other things, the value of Aurora and the release of the Company's Aurora related liabilities as part of the Divestiture. Stratiqa is a member of the Censere Group, a global advisory group focused on providing services in the areas of valuation and strategic advisory, forensics and research. We selected Stratiqa after interviewing Stratiqa and one other financial advisory firm. We instructed Stratiqa to determine the fair market value of 100% equity interest in Aurora, as we plan to divest the 50% interest in Aurora held by us as part of the Divestiture in exchange for Navitus forgiving certain deferred distributions which we are currently obligated to pay Navitus. In addition, we will assume certain payables currently on the Aurora balance sheet. We did not impose any limitation on Stratiqa regarding the scope of its investigation into the valuation of Aurora.

The valuation presentation and report given to our board by Stratiqa was not a fairness opinion and does not contain Stratiqa's recommendation or opinion as to whether the transactions with Armacor or the Divestiture are fair from a financial point of view or advisable, and should not be used as such. Our board did not obtain a fairness opinion because it felt that the cost of obtaining such an opinion outweighed the potential benefits thereof especially given the financial condition of our company and the need to utilize cash resources for other corporate purposes. One of the significant differences between a fairness opinion and a valuation presentation is that a fairness opinion is an objective third party opinion of the fairness of a proposed transaction whereas a valuation presentation and report like the one provided by Stratiqa does not contain the independent opinion of Stratiqa as to the fairness of the Divestiture. Other than our engagement of Stratiqa to prepare the valuation presentation and the payment to Stratiqa of customary market rate fees under such engagement, no material relationship existed during the past two years or is mutually understood to be contemplated between Stratiqa, its affiliates, and/or unaffiliated representatives and our company.

The valuation report concluded that the liabilities of Aurora exceed the assets of Aurora and that the fair market value of Aurora at June 30, 2017 is zero. Stratiqa employed the net asset value method to value Aurora. The net asset value method is a manner of determining a value of a business, business ownership interest, or security based on the value of the assets net of liabilities. Under this approach, the value of a business is equivalent to the fair value of its assets less the fair value of its liabilities. Stratiqa determined the fair market value of oil and gas assets of Aurora as at June 30, 2017 by utilizing the discounted cash flow method, cross-checked for reference only with an analysis of comparable transactions. The fair market value of oil and gas assets were adjusted with balance sheet assets and liabilities to arrive at the fair market value of Aurora.

The valuation report will be made available for inspection and copying at the principal executive offices of our company during its regular business hours by any interested equity security holder of our company or representative who has been so designated in writing.

Pro Forma Information

Please see Annex D for unaudited pro forma financial information showing the pro forma effect of the Divestiture. Our consolidated financial statements for the six months ended June 30, 2017 and 2016 and for the years ended December 31, 2016 and 2015 are attached as Annex E.

Requirements for Stockholder Approval

The divestiture and sale of our ownership interest in Aurora represents the sale of a substantial portion of our assets. Although not required under Nevada law, we therefore agreed, pursuant to the Divestiture Agreement, to obtain stockholder approval of the Divestiture.

Interests of Certain Persons in the Divestiture Proposal

As noted under "Transactions with Related Persons" above, Dr. Ronald Zamber, the chairman of our board of directors and beneficial owner of approximately 49.68% of the shares eligible to vote at the Meeting, is the managing member of James Capital Consulting, LLC, which is the managing partner of Navitus.

No Dissenters' Rights

Under Nevada law, holders of our common stock are not entitled to dissenter's rights of appraisal with respect to the approval of the Divestiture Proposal.

Vote Required

The approval of the Divestiture Proposal requires the affirmative vote of a majority of the shares present in person or represented by proxy at the Meeting. An abstention will have no effect on the Divestiture Proposal. Since the Divestiture Proposal is considered to be a "non-routine" matter, shares that constitute broker non-votes are not considered votes cast on the Divestiture Proposal.

Our board of directors unanimously recommends a vote "FOR" approval of the Divestiture Proposal.

PROPOSAL NO. 3 – 2017 EQUITY INCENTIVE PLAN

Overview

Our board of directors is asking stockholders to approve the Plan. Long-term incentives have been a critical component of our compensation programs and are intended to reward our employees for long-term sustained performance that is aligned with stockholder interests. Our board approved the Plan for grants of restricted stock, stock options and other forms of incentive compensation to our officers, employees, directors and consultants. The Plan was adopted on September 14, 2017, subject to stockholder approval.

Approval of the Plan is needed to replenish the pool of shares available for the grant of stock-based compensation. No shares remain available as of the Record Date for grant under our 2014 Long Term Incentive Plan. Therefore, if stockholder approval is not obtained, we will not be able to grant incentive equity awards other than nonqualified awards granted outside of any formal plan.

The Plan incorporates key corporate governance practices, including the following:

limits the number of shares available to 570,000,000 (or 15,000,000 following the planned 1-for-38 reverse stock split), which represents approximately 18% of our issued and outstanding common shares on a fully-diluted basis (assuming conversion of all outstanding preferred stock and all other securities that are exercisable or exchangeable for, or convertible into, our common stock) as of the Record Date;

all shares granted in connection with awards other than options and stock appreciation rights count as two (2) shares against the share limit;

the price of any option may not be altered or repriced without stockholder approval;

discounted stock options and stock appreciation rights are prohibited;

reload options are not permitted;

performance goals may be imposed on grants;

no ability of participants to receive dividend payments with respect to restricted stock until the shares are vested; diberal share counting is not permitted;

payment of the exercise price or applicable taxes made by delivery of shares, or withholding of shares, in satisfaction of a participant's obligation, will not result in additional shares becoming available for subsequent awards under the Plan; and

qualifies performance incentive awards as "performance-based compensation" under Section 162(m) of the Internal Revenue Code of 1986, as amended (the "Code").

The Plan is being submitted to the stockholders for approval in order to qualify certain awards made to certain officers as deductible for federal income tax purposes under section 162(m) of the Code. Stockholder approval is also necessary under the federal income tax rules applicable to incentive stock options.

Our board believes equity compensation is an important component of our compensation programs. Our ability to attract, retain and motivate top quality employees is material to our success, and we believe we can better achieve these objectives with grants made under the Plan. In addition, our board believes that the interests of both our company and our stockholders are advanced by affording our employees, officers and directors the opportunity to acquire or increase their proprietary interests in our company.

Significant Features of the Plan

The following is a summary of certain significant features of the Plan. The information which follows is subject to, and qualified in its entirety by reference to, the Plan document, which is attached to this proxy statement as Annex C. We urge you to read the Plan in its entirety.

Awards that may be granted include: (a) Incentive Stock Options, (b) Non-qualified Stock Options, (c) Stock Appreciation Rights, (d) Restricted Awards, (e) Performance Share Awards, and (f) Performance Compensation Awards. These awards offer our officers, employees, consultants and directors the possibility of future value, depending on the long-term price appreciation of our common stock and the award holder's continuing service with our company.

Stock options give the option holder the right to acquire from our company a designated number of shares of common stock at a purchase price that is fixed upon the grant of the option. The exercise price will be not less than the market price of the common stock on the date of grant. Stock options granted may be either tax-qualified stock options (so-called "incentive stock options") or non-qualified stock options.

Stock appreciation rights ("SARs"), which may be granted alone or in tandem with options, have an economic value similar to that of options. When a SAR for a particular number of shares is exercised, the holder receives a payment equal to the difference between the market price of the shares on the date of exercise and the exercise price of the shares under the SAR. Again, the exercise price for SARs normally is the market price of the shares on the date the SAR is granted. Under the Plan, holders of SARs may receive this payment – the appreciation value – either in cash or shares of common stock valued at the fair market value on the date of exercise. The form of payment will be determined by our company.

Restricted shares are shares of common stock awarded to participants at no cost. Restricted shares can take the form of awards of restricted stock, which represent issued and outstanding shares of our common stock subject to vesting criteria, or restricted stock units, which represent the right to receive shares of our common stock subject to satisfaction of the vesting criteria. Restricted shares are forfeitable and non-transferable until the shares vest. The vesting date or dates and other conditions for vesting are established when the shares are awarded.

The Plan also provides for performance compensation awards, representing the right to receive a payment, which may be in the form of cash, shares of common stock, or a combination, based on the attainment of pre-established goals. All of the permissible types of awards under the Plan are described in more detail as follows:

Purposes of Plan: The purposes of the Plan are to: attract and retain officers, employees and directors for our company and its subsidiaries; motivate them by means of appropriate incentives to achieve long-range goals; provide incentive compensation opportunities; and further align their interests with those of our stockholders through compensation that is based on our common stock.

Administration of the Plan: Administration of the Plan is entrusted to the compensation committee of the board of directors (the "Committee"). Among other things, the Committee has the authority to select persons who will receive awards, determine the types of awards and the number of shares to be covered by awards, and to establish the terms, conditions, performance criteria, restrictions and other provisions of awards. The Committee has authority to establish, amend and rescind rules and regulations relating to the Plan.

Eligible Recipients: Persons eligible to receive awards under the Plan will be those officers, employees, consultants, and directors of our company and its subsidiaries who are selected by the Committee administering the Plan. Shares Available Under the Plan: The maximum number of shares of our common stock that may be delivered to participants under the Plan is 570,000,000, subject to adjustment for certain corporate changes affecting the shares, such as stock splits. Following the planned 1-for-38 reverse stock split, such number shall be adjusted to 15,000,000 shares. Shares subject to an award under the Plan for which the award is canceled, forfeited or expires again become

available for grants under the Plan. Shares subject to an award that is settled in cash will not again be made available for grants under the Plan. The maximum number of shares that may be covered by awards to any single individual in any year is 9,500,000 (or 250,000 following the planned 1-for-38 reverse stock split) and the maximum cash payment that can be made to any individual for any single or combined performance goals for any performance period is \$250,000.

Stock Options:

General. Subject to the provisions of the Plan, the Committee has the authority to determine all grants of stock options. That determination will include: (i) the number of shares subject to any option; (ii) the exercise price per share; (iii) the expiration date of the option; (iv) the manner, time and date of permitted exercise; (v) other restrictions, if any, on the option or the shares underlying the option; and (vi) any other terms and conditions as the Committee may determine.

Option Price. The exercise price for stock options will be determined at the time of grant. Normally, the exercise price will not be less than the fair market value on the date of grant. As a matter of tax law, the exercise price for any incentive stock option awarded may not be less than the fair market value of the shares on the date of grant. However, incentive stock option grants to any person owning 10% or more of our voting stock must have an exercise price of not less than 110% of the fair market value on the grant date.

Exercise of Options. An option may be exercised only in accordance with the terms and conditions for the option agreement as established by the Committee at the time of the grant. The option must be exercised by notice to our company, accompanied by payment of the exercise price. Payments may be made in cash or, at the option of the Committee, by actual or constructive delivery of shares of common stock to the holder of the option based upon the fair market value of the shares on the date of exercise.

Expiration or Termination. Options, if not previously exercised, will expire on the expiration date established by the Committee at the time of grant. In the case of incentive stock options, such term cannot exceed ten years provided that in the case of holders of 10% or more of our voting stock, such term cannot exceed five years. Options will terminate before their expiration date if the holder's service with our company or a subsidiary terminates before the expiration date. The option may remain exercisable for specified periods after certain terminations of employment, including terminations as a result of death, disability or retirement, with the precise period during which the option may be exercised to be established by the Committee and reflected in the grant evidencing the award.

Incentive and Non-Qualified Options. As described elsewhere in this summary, an incentive stock option is an option that is intended to qualify under certain provisions of Code for more favorable tax treatment than applies to non-qualified stock options. Any option that does not qualify as an incentive stock option will be a non-qualified stock option. Under the Code, certain restrictions apply to incentive stock options. For example, the exercise price for incentive stock options may not be less than the fair market value of the shares on the grant date and the term of the option may not exceed ten years. In addition, an incentive stock option may not be transferred, other than by will or the laws of descent and distribution, and is exercisable during the holder's lifetime only by the holder. In addition, no incentive stock options may be granted to a holder that is first exercisable in a single year if that option, together with all incentive stock options previously granted to the holder that also first become exercisable in that year, relate to shares having an aggregate market value in excess of \$100,000, measured at the grant date.

Stock Appreciation Rights: Awards of SARs may be granted alone or in tandem with stock options. SARs provide the holder with the right, upon exercise, to receive a payment, in cash or shares of stock, having a value equal to the excess of the fair market value on the exercise date of the shares covered by the award over the exercise price of those shares. Essentially, a holder of a SAR benefits when the market price of the common stock increases, to the same extent that the holder of an option does, but, unlike an option holder, the SAR holder need not pay an exercise price upon exercise of the award.

Stock Awards: Stock awards can also be granted under the Plan. A stock award is a grant of shares of common stock or of a right to receive shares in the future. These awards will be subject to such conditions, restrictions and contingencies as the Committee shall determine at the date of grant. Those may include requirements for continuous service and/or the achievement of specified performance goals.

Cash Awards: A cash award is an award that may be in the form of cash or shares of common stock or a combination, based on the attainment of pre-established performance goals and other conditions, restrictions and contingencies identified by the Committee.

Section 162(m) of the Code: Section 162(m) of the Code limits publicly-held companies to an annual deduction for U.S. federal income tax purposes of \$1.0 million for compensation paid to each of their chief executive officer and their three highest compensated executive officers (other than the chief financial officer) determined at the end of each year, referred to as covered employees. However, performance-based compensation is excluded from this limitation. The Plan is designed to permit the Committee to grant awards that qualify as performance-based for purposes of satisfying the conditions of Section 162(m), but it is not required under the Plan that awards qualify for this exception. Performance Criteria: Under the Plan, one or more of the following performance criteria will be used by the Committee in establishing performance goals:

net earnings or net income (before or after taxes);

basic or diluted earnings per share (before or after taxes);

net revenue or net revenue growth;

gross revenue;

gross profit or gross profit growth;

net operating profit (before or after taxes);

return on assets, capital, invested capital, equity, or sales;

eash flow (including, but not limited to, operating cash flow, free cash flow, and cash flow return on capital);

earnings before or after taxes, interest, depreciation and/or amortization;

gross or operating margins;

improvements in capital structure;

budget and expense management;

productivity ratios;

economic value added or other value added measurements;

share price (including, but not limited to, growth measures and total shareholder return);

expense targets;

margins;

operating efficiency;

working capital targets;

enterprise value;

safety record;

completion of acquisitions or business expansion;

achieving research and development goals and milestones;

achieving product commercialization goals; and

other criteria as may be set by the Committee from time to time.

Any one or more of the performance criteria may be used on an absolute or relative basis to measure the performance of our company, as the Committee may deem appropriate, or as compared to the performance of a group of comparable companies, or published or special index that the Committee deems appropriate.

In determining the actual size of an individual performance compensation award, the Committee may reduce or eliminate the amount of the award through the use of negative discretion if, in its sole judgment, such reduction or elimination is appropriate. The Committee shall not have the discretion to (i) grant or provide payment in respect of performance compensation awards if the performance goals have not been attained or (ii) increase a performance compensation award above the maximum amount payable under the Plan.

Other Material Provisions: Awards will be evidenced by a written agreement, in such form as may be approved by the Committee. In the event of various changes to the capitalization of our company, such as stock splits, stock dividends and similar re-capitalizations, an appropriate adjustment will be made by the Committee to the number of shares covered by outstanding awards or to the exercise price of such awards. The Committee is also permitted to include in the written agreement provisions that provide for certain changes in the award in the event of a change of control of our company, including acceleration of vesting. Except as otherwise determined by the Committee at the date of grant, awards will not be transferable, other than by will or the laws of descent and distribution. Prior to any award distribution, our company is permitted to deduct or withhold amounts sufficient to satisfy any employee withholding tax requirements. Our board also has the authority, at any time, to discontinue the granting of awards. The board also has the authority to alter or amend the Plan or any outstanding award or may terminate the Plan as to further grants,

provided that no amendment will, without the approval of our stockholders, to the extent that such approval is required by law or the rules of an applicable exchange, increase the number of shares available under the Plan, change the persons eligible for awards under the Plan, extend the time within which awards may be made, or amend the provisions of the Plan related to amendments. No amendment that would adversely affect any outstanding award made under the Plan can be made without the consent of the holder of such award.

Federal Income Tax Consequences of Awards: The following is based on current laws, regulations and interpretations, all of which are subject to change. It does not purport to be complete and does not describe the state, local or foreign tax considerations or the consequences for any particular individual.

Stock Options. In general, the grant of a stock option will not be a taxable event to the recipient and will not result in a tax deduction to our company. The tax consequences resulting from an exercise of a stock option and the subsequent disposition of the shares acquired upon the exercise depends, in part, on whether the option is an incentive stock option or a non-qualified stock option. Upon the exercise of a non-qualified stock option, the holder will recognize ordinary compensation income in an amount equal to the excess of the fair market value of the shares received upon exercise over the exercise price (the "spread"). Our company will be able to claim a tax deduction for this spread, provided it satisfies compensation reporting requirements under the Code and is not otherwise precluded from taking a deduction because of Section 162(m) deduction limitations described below. Any gain or loss upon the subsequent sale or exchange of the shares by the holder will be capital gain or loss, long term or short term, depending upon the holding period for the shares. Upon the exercise of an incentive stock option, a holder will generally not recognize taxable income at the time of exercise and no tax deduction will be available to our company, provided the option is exercised when the holder is an employee of our company or, in certain circumstances, within a limited time thereafter. The difference between the exercise price and the fair market value of the shares on the date of exercise is treated by the holder as an item of adjustment for purposes of the alternative minimum tax. If the shares acquired upon an exercise of an incentive stock option are subsequently sold by the holder and such sale takes place after the statutory "holding period" (which is the later of two years from the date of grant or one year after the date of exercise), the gain or loss realized will be the difference between the sales price and the exercise price and will be treated as a long term capital gain or loss. If the sale takes place prior to expiration of the holding period, the holder of the shares will recognize ordinary income at the time of sale equal to the spread and our company will be entitled to a tax deduction in equal amount. The remaining gain to the holder, if any, will be capital gain, either long term or short

Stock Appreciation Rights. No taxable income will be realized by a recipient in connection with the grant of a SAR. Generally, when the holder of a SAR exercises the SAR, the amount of cash or the fair market value of the shares received upon exercise will be ordinary compensation income to the holder and our company will be entitled to a corresponding tax deduction, subject to Section 162(m).

Restricted Shares. An award of restricted shares, like the grant of an option, is not taxable to the recipient. The holder of restricted shares generally will recognize ordinary compensation income at the time the restrictions on the shares lapse, which is the vesting date thereof, based on the fair market value of our shares on that date. Subject to the Section 162(m) limitations, this amount is deductible for federal income tax purposes by our company. Dividends paid with respect to restricted shares prior to vesting will be taxable as ordinary compensation income to the holder (not as "qualifying dividends") and will be deductible by our company. A holder of restricted shares may elect under Section 83(b) of the Code, in lieu of the treatment described above, to take immediate recognition of income at the time the shares are received. In that event, the holder will recognize ordinary compensation income equal to the fair market value of the shares at the date of grant, which amount will be deductible by our company, and dividends subsequently paid to the holder with respect to the shares will be taxable to the holder as "qualifying dividends" and will not be deductible by our company.

Other Awards. Cash awards are generally taxable as ordinary compensation income in the year of receipt and will be deductible as such by our company. Restricted stock units, deferred cash awards and other types of deferred awards are subject to Section 409A of the Code regarding non-qualified deferred compensation plans. Our company intends to use reasonable efforts to design any such awards in a manner that avoids Section 409A or that complies with Section 409A.

Potential Limitation on Company Deductions. Our company will generally be entitled to a tax deduction in connection with awards in an amount equal to the ordinary income recognized by a recipient and at the time the recipient realizes

such income, subject to Section 162(m) limitations of the Code, as discussed elsewhere in this proxy statement. Recognition of Compensation Expense. In accordance with Statement of Financial Accounting Standards No.123R, "Share-Based Payment," our company is required to recognize compensation expense in its income statement for the grant-date fair value of stock options and other equity-based compensation issued to its employees and directors, the amount of which can only be determined at the time of grant.

New Plan Benefits

The future awards, if any, that will be made to eligible persons under the Plan are subject to the discretion of the Committee and, therefore, we cannot currently determine the benefits or number of shares subject to awards that may be granted in the future to our employees, consultants and non-employee directors under the Plan. Therefore, a New Plan Benefits Table is not provided.

Vote Required

The approval of the Plan requires the affirmative vote of a majority of the shares present in person or represented by proxy at the Meeting. An abstention will have no effect on this proposal. Since this proposal is considered to be a "non-routine" matter, shares that constitute broker non-votes are not considered votes cast on this proposal. Our board of directors unanimously recommends a vote "FOR" approval of the Plan.

PROPOSAL NO. 4 – DIRECTOR PROPOSAL

Director Nominees

Our board has recommended the election of the seven (7) director nominees listed below.

Kenneth Hill

Ronald W. Zamber

Robert Grenley

Ricardo A. Salas

Kevin DeLeon

Julio C. Herrera

Eric Eilertsen

For biographical information regarding these nominees, see "Directors and Executive Officers" above. If elected at the Meeting, these nominees will hold office until the next annual meeting or until their successors are qualified, subject to their prior death, resignation or removal.

The slate of nominees to the Board is favored by our board. The present board believes that the slate reflects a broad range of experience with regard to financial, investment and regulatory matters and to the various interests of our company. Finally, the present board believes that the slate of directors contains individuals who will be able to assist in the further development of our company.

Vote Required

The election of each director nominee requires the affirmative vote of a plurality of votes of the shares cast at the election. That means the seven (7) nominees will be elected if they receive more affirmative votes than any other nominees. In the absence of instructions to the contrary, shares of common stock represented by properly executed proxies will be voted for the seven (7) nominees listed herein below, all of whom are recommended by our board and who have consented to be named and to serve if elected.

In the event that any nominee recommended by our board is unable or declines to serve as a director at the time of the Meeting, the proxies will be voted for any nominee who is designated by the present board to fill the vacancy. It is not expected that any nominee will be unable or will decline to serve as a director. Our board knows of no reason why any of the nominees will be unavailable or decline to serve as a director.

Our Board of Directors unanimously recommends a vote "FOR" each director nominee.

PROPOSAL NO. 5 – ADJOURNMENT PROPOSAL

Our board has determined that the adjournment of the Meeting for any purpose, including to solicit additional proxies if there are insufficient votes at the time of the Meeting to approve the proposals described herein, is advisable and in

the best interests of our company and its stockholders and has approved the adjournment of the Meeting for any purpose, including to solicit additional proxies if there are insufficient votes at the time of the Meeting to approve the proposals described herein.

The Meeting may be adjourned by the affirmative vote of a majority of the shares present in person or represented by proxy at the Meeting.

Our board of directors unanimously recommends a vote "FOR" approval of the Adjournment Proposal.

TRANSACTION OF OTHER BUSINESS

At the date of this proxy statement, the only business which the board intends to present at the Meeting is as set forth above. If any other matter or matters are properly brought before the Meeting, or an adjournment or postponement thereof, it is the intention of the persons named in the accompanying form of proxy to vote the proxy on such matters in accordance with their best judgment.

HOUSEHOLDING OF PROXY STATEMENT

The rules promulgated by the SEC permit companies, brokers, banks or other intermediaries to deliver a single copy of our proxy materials to households at which two or more stockholders reside ("Householding"). Stockholders sharing an address who have been previously notified by their broker, bank or other intermediary and have consented to Householding, either affirmatively or implicitly by not objecting to Householding, received only one copy of our proxy materials. A stockholder who wishes to participate in Householding in the future must contact his or her broker, bank or other intermediary directly to make such request. Alternatively, a stockholder who wishes to revoke his or her consent to Householding and receive separate proxy materials for each stockholder sharing the same address must contact his or her broker, bank or other intermediary to revoke such consent. Stockholders may also obtain a separate proxy statement or may receive a printed or an e-mail copy of this proxy statement without charge by sending a written request to Victory Energy Corporation, Attn: Corporate Secretary, 3355 Bee Caves Road, Suite 608, Austin, Texas 78746. We will promptly deliver a copy of this proxy statement upon request. Householding does not apply to stockholders with shares registered directly in their name.

WHERE YOU CAN FIND MORE INFORMATION

We file annual, quarterly and current reports, proxy statements and other information with the SEC under the Exchange Act. You may read and copy this information at, or obtain copies of this information by mail from, the SEC's Public Reference Room, 450 Fifth Street, N.W., Washington, D.C. 20549, at prescribed rates. Please call the SEC at (800) SEC-0330 for further information about the public reference room. Our filings with the SEC are also available to the public from commercial document retrieval services and at the web site maintained by the SEC at http://www.sec.gov.

The SEC allows us to "incorporate by reference" into this proxy statement documents that we file with the SEC. This means that we can disclose important information to you by referring you to those documents. The information incorporated by reference is considered to be a part of this proxy statement, and later information that we file with the SEC will update and supersede that information. We incorporate by reference the documents listed below and any documents filed by us pursuant to Sections 13(a), 13(c), 14 or 15(d) of the Exchange Act after the date of this proxy statement and before the date of the Meeting:

Annual Report on Form 10-K for the fiscal year ended December 31,

Quarterly Reports on Form 10-Q for the quarters ended March 31, 2017 and June 30, 2017; and

Current Reports on Form 8-K filed on February 7, 2017, April 13, 2017 and August 24,

Notwithstanding the foregoing, information furnished under Items 2.02, 7.01 and 8.01 of any Current Report on Form 8-K, including the related exhibits, is not incorporated by reference in this proxy statement. In addition, statements contained in this proxy statement, or in any document incorporated in this proxy statement by reference,

regarding the contents of any contract or other document, are only summaries of the material terms and as such we encourage you to carefully read in its entirety that contract or other document filed as an exhibit with the SEC. Any person, including any beneficial owner, to whom this proxy statement is delivered may request copies of proxy statements and any of the documents incorporated by reference in this document or other information concerning us, without charge, by written request directed to Victory Energy Corporation, Attn: Corporate Secretary, 3355 Bee Caves Road, Suite 608, Austin, Texas 78746, or by calling us at (512) 347-7300. Documents incorporated by reference are available without charge, excluding any exhibits to those documents unless the exhibit is specifically incorporated by reference into those documents.

THIS PROXY STATEMENT DOES NOT CONSTITUTE THE SOLICITATION OF A PROXY IN ANY JURISDICTION TO OR FROM ANY PERSON TO WHOM OR FROM WHOM IT IS UNLAWFUL TO MAKE SUCH PROXY SOLICITATION IN THAT JURISDICTION. YOU SHOULD RELY ONLY ON THE INFORMATION CONTAINED OR INCORPORATED BY REFERENCE IN THIS PROXY STATEMENT TO VOTE YOUR SHARES AT THE MEETING. WE HAVE NOT AUTHORIZED ANYONE TO PROVIDE YOU WITH INFORMATION THAT IS DIFFERENT FROM WHAT IS CONTAINED IN THIS PROXY STATEMENT. THIS PROXY STATEMENT IS DATED [], 2017. YOU SHOULD NOT ASSUME THAT THE INFORMATION CONTAINED IN THIS PROXY STATEMENT IS ACCURATE AS OF ANY DATE OTHER THAN THAT DATE, AND THE MAILING OF THIS PROXY STATEMENT TO STOCKHOLDERS DOES NOT CREATE ANY IMPLICATION TO THE CONTRARY.

ANNEX A
AMENDED AND RESTATED
ARTICLES OF INCORPORATION
OF
VICTORY ENERGY CORPORATION

Victory Energy Corporation (hereinafter referred to as the "Corporation"), a corporation organized and existing under and by virtue of the laws of the State of Nevada, does hereby certify as follows:

- 1. The current name of the Corporation is Victory Energy Corporation.
- 2. The original Articles of Incorporation of the Corporation were filed in the Office of the Secretary of State on January 7, 1982.
- 3. These Amended and Restated Articles of Incorporation have been duly approved by the Board of Directors of the Corporation pursuant to a unanimous written consent in lieu of a meeting, dated September 14, 2017, and by the holders of a majority of the Corporation's issued and outstanding capital stock entitled to vote thereon at a meeting duly called and held on ______, 2017, in accordance with the provisions of Sections 78.390 and 78.403 of the Nevada Revised Statutes.
- 4. The provisions of the Articles of Incorporation of the Corporation as heretofore amended and/or supplemented are hereby restated, integrated and further amended to read in its entirety as follows:

ARTICLE I NAME

The name of the corporation is Victory Energy Corporation (hereinafter, the "Corporation").

ARTICLE II PURPOSE

The purpose of the Corporation is to engage in any lawful act or activity for which corporations may be organized under the laws of the State of Nevada.

ARTICLE III

AUTHORIZATION TO ISSUE CAPITAL STOCK

The aggregate number of shares which the Corporation shall have the authority to issue is 300,000,000 shares of Common Stock having a par value of \$0.001 per share and 10,000,000 shares of Preferred Stock having a par value of \$0.001 per share. All Common Stock of the Corporation shall be of the same class and shall have the same rights and preferences. The Corporation shall have authority to issue the shares of Preferred Stock in one or more series with such rights, preferences and designations as determined by the Board of Directors of the Corporation. Authority is hereby expressly granted to the Board of Directors from time to time to issue Preferred Stock in one or more series, and in connection with the creation of any such series, by resolution or resolutions providing for the issue of the shares thereof, to determine and fix such voting powers, full or limited, or no voting powers, and such designations, preferences and relative participating, optional or other special rights, and qualifications, limitations or restrictions thereof, including, without limitation thereof, dividend rights, special voting rights, conversion rights, redemption privileges and liquidation preferences, as shall be stated and expressed in such resolutions, all to the full extent now or hereafter permitted by the Nevada Revised Statutes. Full-paid stock of the Corporation shall not be liable to any further call or assessment.

Effective as of _______, 2017, each thirty-eight (38) shares of the issued and outstanding Common Stock of the Corporation shall be reverse split into one (1) share of common stock of the Corporation. No fractional shares shall be issued in connection with the reverse split. Any fractional share that would otherwise be issued as a result of the reverse split will be rounded up to the nearest whole share.

ARTICLE IV GOVERNING BOARD

The members of the governing board of the Corporation shall be known as the board of directors. The number of directors comprising the board of directors shall be set forth in the bylaws of the Corporation.

ARTICLE V

INDEMNIFICATION OF OFFICERS AND DIRECTORS

The Corporation is authorized to indemnify any person who was or is a party or is threatened to be made a party to any threatened, pending or completed action, suit or proceeding, whether civil, criminal, administrative or investigative, by reason of the fact that such person is or was a director or officer of the Corporation, or who is or was serving at the request of the Corporation as a director or officer of another corporation, partnership, joint venture, trust or other enterprise, against expenses, including attorneys' fees, judgments, fines and amounts paid in settlement actually and reasonably incurred by such person in connection with the action, suit or proceeding, to the full extent permitted by the Nevada Revised Statutes as such statutes may be amended from time to time.

ARTICLE VI LIABILITY OF DIRECTORS AND OFFICERS

No director or officer shall be personally liable to the Corporation or any of its stockholders for damages for any breach of fiduciary duty as a director or officer; provided, however, that the foregoing provision shall not eliminate or limit the liability of a director or officer (i) for acts or omissions which involve intentional misconduct, fraud or a knowing violation of law, or (ii) for the payment of dividends in violation of Section 78.300 of the Nevada Revised Statutes. Any repeal or modification of this Article VI by the stockholders of the Corporation shall be prospective only, and shall not adversely affect any limitation of the personal liability of a director of officer of the Corporation for acts or omissions prior to such repeal or modification.

ARTICLE VII ACQUISITION OF CONTROLLING INTEREST

The Corporation elects not to be governed by the terms and provisions of Sections 78.378 through 78.3793, inclusive, of the Nevada Revised Statutes, as the same may be amended, superseded, or replaced by an successor section, statute, or provision. No amendment to these Amended and Restated Articles of Incorporation, directly or indirectly, by merger or consolidation or otherwise, having the effect of amending or repealing any provision of this Article VII shall apply to or have any effect on any transaction involving acquisition of control by any person occurring prior to such amendment or repeal.

ARTICLE VIII COMBINATIONS WITH INTERESTED STOCKHOLDERS

The Corporation elects not to be governed by the terms and provisions of Sections 78.411 through 78.444, inclusive, of the Nevada Revised Statutes, as the same may be amended, superseded, or replaced by any successor section, statute, or provision. No amendment to these Amended and Restated Articles of Incorporation, directly or indirectly, by merger or consolidation or otherwise, having the effect of amending or repealing any provision of this Article VIII shall apply to or have any effect on any transaction with an interested stockholder occurring prior to such amendment or repeal.

IN WITNESS	WHEREOF,	the undersigned has	executed these	Amended and Restated	d Articles of Incorpo	oration this
day of		_ 2017.				

VICTORY ENERGY CORPORATION

By:

Name: Kenneth Hull

Title: Chief Executive Officer

ANNEX B

DIVESTITURE AGREEMENT

dated as of August 21, 2017

between

VICTORY ENERGY CORPORATION

AND

NAVITUS ENERGY GROUP

8.10. Severability. 1 8.11. Expenses. 1

TABLE OF CONTENTS Page Article I DEFINITIONS 1 1.1.Certain Definitions. 1 Article II DIVESTITURE 1 2.1.Divestiture 1 <u>2.2.</u>Closing. 1 2.3. Transactions to be Effected at the Closing. 1 Article III REPRESENTATIONS AND WARRANTIES OF VICTORY 1 3.1. Organization; Authority and Enforceability. 1 3.2. Noncontravention. 1 3.3.The Interests. 1 3.4.Brokers' Fees. 1 Article IV REPRESENTATIONS AND WARRANTIES OF NAVITUS 1 <u>4.1.</u>Organization; Authority and Enforceability. 1 4.2. Noncontravention. 1 4.3.Brokers' Fees. 1 Article V COVENANTS 1 5.1.Consents. 1 5.2. Taking of Necessary Action; Further Action. 1 5.3. Stockholders Meeting; Preparation of Proxy Statement. 1 5.4. Repricing of Warrants. 1 Article VI CONDITIONS TO OBLIGATIONS TO CLOSE 1 6.1. Conditions to the Obligation of Parties. 1 6.2. Conditions to Obligation of Navitus. 1 6.3. Conditions to Obligation of Victory. 1 Article VII TERMINATION; AMENDMENT; WAIVER 1 7.1. Termination of Agreement. 1 7.2.Effect of Termination. 1 7.3.Amendments. 1 <u>7.4.</u>Waiver. 1 Article VIII MISCELLANEOUS 1 8.1. Press Releases and Public Announcement. 1 8.2. No Third-Party Beneficiaries. 1 8.3. Entire Agreement. 1 8.4. Succession and Assignment. 1 8.5.Construction. 1 8.6. Notices. 1 8.7. Governing Law. 1 8.8. Consent to Jurisdiction and Service of Process. 1 <u>8.9.</u>Headings. 1

8.12.Incorporation of Exhibits and Schedules. 1

8.13.Limited Recourse. 1

8.14. Specific Performance. 1

8.15. Counterparts. 1

8.16. No Survival of Representations and Warranties. 1

DIVESTITURE AGREEMENT

DIVESTITURE AGREEMENT, dated August 21, 2017 (the "Agreement"), between VICTORY ENERGY CORPORATION, a Nevada corporation ("Victory") and NAVITUS ENERGY GROUP, a Texas general partnership ("Navitus").

RECITALS

- A. Victory is the managing partner and beneficial owner of fifty percent (50%) of the partnership interests (the "Interests") in Aurora Energy Partners, a Texas general partnership (the "Partnership"). Navitus is the owner of the remaining fifty percent (50%) of the partnership interests in the Partnership.
- B. Victory desires to divest and transfer to Navitus all of the Interests (the "Divestiture") in consideration for a release from Navitus of all of Victory's obligations under the Partnership Agreement (as defined below), including, without limitation, obligations to return to Navitus investors their accumulated deferred capital, deferred interest and related allocations of equity.

AGREEMENT

NOW, THEREFORE, in consideration of the mutual promises herein contained, the parties hereto, intending to be legally bound, hereby agree as follows:

Article I

DEFINITIONS

- 1.1. Certain Definitions.
- (a) When used in this Agreement, the following terms will have the meanings assigned to them in this Section 1.1(a) and other defined terms will have the meanings given to them elsewhere in this Agreement:
- "Affiliate" means, with respect to a Person, any other Person that, directly or indirectly, through one or more intermediaries, Controls, is Controlled by or is under common Control with, such Person. For purposes of this definition, "Control" (including the terms "Controlled by" and "under common Control with") means possession of the power to direct or cause the direction of the management or policies of a Person, whether through the ownership of stock, partnership interests, membership interests or other equity interests, as trustee or executor, by contract or otherwise.
- "Amended Partnership Agreement" means the third amended partnership agreement of the Partnership reflecting, among other things, Navitus as the managing partner and owner of the Interests, to be executed and delivered at the Closing, in the form reasonably satisfactory to Navitus and Victory.
- "Business Day" means a day other than a Saturday, Sunday or other day on which banks located in Austin, Texas are authorized or required by Law to close.
- "Common Stock" means the common stock, par value \$0.001 per share, of Victory.
- "Exchange Act" means the Securities Exchange Act of 1934, as amended.
- "Governmental Entity" means any entity or body exercising executive, legislative, judicial, regulatory or administrative functions of or pertaining to United States federal, state or local government or foreign, international, multinational or other government, including any department, commission, board, agency, bureau, official or other regulatory, administrative or judicial authority thereof.
- "Law" means any statute, law, ordinance, rule, or regulation of any Governmental Entity.
- "Liability" means all indebtedness, obligations and other liabilities and contingencies of a Person, whether absolute, accrued, contingent, fixed or otherwise, or whether due or to become due.
- "Lien" means, with respect to any property or asset, any mortgage, lien, pledge, charge, security interest, hypothecation or other encumbrance in respect of such property or asset.
- "Partnership Agreement" means the second amended partnership agreement of the Partnership, dated October 1, 2011, between Victory and Navitus, currently in effect, which shall be amended by the Third Amended Partnership Agreement pursuant to the terms of this Agreement.

- "Permit" means any authorization, approval, consent, certificate, license, permit or franchise of or from any Governmental Entity of competent jurisdiction or pursuant to any Law.
- "Permitted Liens" means (i) Liens for taxes that are not yet due and payable or that may hereafter be paid without material penalty or that are being contested in good faith, (ii) statutory Liens of landlords and workers', carriers' and mechanics' or other like Liens incurred in the ordinary course of business or that are being contested in good faith, (iii) Liens and encroachments which do not materially interfere with the present or proposed use of the properties or assets they affect, (iv) Liens that will be released prior to or as of the Closing and (v) Liens arising under this Agreement. "Person" means an individual, a corporation, a partnership, a limited liability company, a trust, an unincorporated association, a Governmental Entity or any agency, instrumentality or political subdivision of a Governmental Entity, or any other entity or body.
- "Release" means the mutual release in the form of Exhibit A to this Agreement.
- "Securities Act" means the Securities Act of 1933, as amended.
- "Shareholder Approval" means a special meeting of the shareholders of Victory that is to take place as soon as practicable following the date hereof (and if possible, on or before November 30, 2017), to obtain the approval of the Victory shareholders of, among other things, the Divestiture contemplated by this Agreement. "\$" means United States dollars.
- (b) For purposes of this Agreement, except as otherwise expressly provided herein or unless the context otherwise requires: (i) the meaning assigned to each term defined herein will be equally applicable to both the singular and the plural forms of such term and vice versa, and words denoting any gender will include all genders as the context requires; (ii) where a word or phrase is defined herein, each of its other grammatical forms will have a corresponding meaning; (iii) the terms "hereof", "herein", "hereunder", "hereby" and "herewith" and words of similar import will, unless otherwise stated, be construed to refer to this Agreement as a whole and not to any particular provision of this Agreement; (iv) when a reference is made in this Agreement to an Article, Section, paragraph, Exhibit or Schedule without reference to a document, such reference is to an Article, Section, paragraph, Exhibit or Schedule to this Agreement; (v) a reference to a subsection without further reference to a Section is a reference to such subsection as contained in the same Section in which the reference appears, and this rule will also apply to paragraphs and other subdivisions; (vi) the word "include", "includes" or "including" when used in this Agreement will be deemed to include the words "without limitation", unless otherwise specified; (vii) a reference to any party to this Agreement or any other agreement or document will include such party's predecessors, successors and permitted assigns; (viii) a reference to any Law means such Law as amended, modified, codified, replaced or reenacted as of the date hereof, and all rules and regulations promulgated thereunder as of the date hereof; and (ix) all accounting terms used and not defined herein have the respective meanings given to them under U.S. generally accepted accounting principles.

ARTICLE II

DIVESTITURE

- 2.1. Divestiture. Upon the terms and subject to the conditions set forth in this Agreement, Victory will sell, transfer and deliver to Navitus free and clear of all Liens, indebtedness and other Liabilities, and Navitus will acquire from Victory, the Interest in exchange for the Release. At or prior to the Closing, Victory shall pay off or otherwise satisfy all indebtedness and other Liabilities of the Partnership specifically listed on Schedule 2.1 hereto, such that the Partnership shall own all of its assets free and clear of all Liens other than Permitted Liens.
- 2.2. Closing. The consummation of the transaction contemplated hereby (the "Closing") will take place by the reciprocal delivery of closing documents by electronic mail, regular mail, fax or any other means mutually agreed upon by the parties on a date that is no later than two Business Days immediately following the day on which the last of the conditions to closing contained in Article IV (other than any conditions that by their nature are to be satisfied at the Closing) is satisfied or waived in accordance with this Agreement or at such other location or on such other date as Navitus and Victory may mutually determine (the date on which the Closing actually occurs is referred to as the "Closing Date").
- 2.3. Transactions to be Effected at the Closing. At the Closing, Navitus and Victory will execute and deliver to each other the Amended Partnership Agreement, which, among other things, will reflect Navitus as the sole owner of the Partnership and will reflect the release of Victory from all of Victory's obligations under the Partnership Agreement, including, without limitation, obligations to return to Navitus investors their accumulated deferred capital, deferred interest and related allocations of equity and Navitus will execute and deliver to Victory the Release, and each Party

will execute and deliver to the other Party all other documents, instruments or certificates required to be delivered by such Party in connection therewith.

ARTICLE III

REPRESENTATIONS AND WARRANTIES OF VICTORY

Victory represents and warrants to Navitus as follows:

- 3.1. Organization; Authority and Enforceability. Victory is a Nevada corporation that is validly existing and in good standing under the Laws of the State of Nevada. Victory has the requisite power and authority to execute and deliver this Agreement, to perform its obligations hereunder and, subject to the Shareholder Approval, to consummate the Acquisition and the other transactions contemplated hereby. The execution, delivery and performance by Victory of this Agreement and the consummation by Victory of the Acquisition and the other transactions contemplated hereby have been duly authorized by all necessary action on the part of Victory and no other action is necessary on the part of Victory to authorize this Agreement or to consummate the Acquisition or the other transactions contemplated hereby (other than the Shareholder Approval). This Agreement has been duly executed and delivered by Victory and, assuming the due authorization, execution and delivery by each other party hereto, constitutes a legal, valid and binding obligation of Victory, enforceable against Victory in accordance with its terms, except as limited by (a) bankruptcy, insolvency, reorganization, moratorium, fraudulent conveyance or other similar laws relating to creditors' rights generally and (b) general principles of equity, whether such enforceability is considered in a proceeding in equity or at Law.
- 3.2. Noncontravention.
- (a) Neither the execution and the delivery of this Agreement nor the consummation of the Divestiture or the other transactions contemplated by this Agreement, will, with or without the giving of notice or the lapse of time or both, (i) to the actual knowledge of Victory, violate any law applicable to Victory or (ii) violate any contract or agreement to which Victory is a party, except in the case of clauses (i) and (ii) to the extent that any such violation would not reasonably be expected to have, individually or in the aggregate, a material adverse effect on the Partnership.
- (b) The execution and delivery of this Agreement by Victory does not, and the performance of this Agreement by Victory will not, require any consent, approval, authorization or Permit of, or filing with or notification to, any Governmental Entity, except for (i) the filings which will be made at or prior to the Closing or following the Closing, but within required time limits or (ii) where the failure to take such action would not reasonably be expected to have, individually or in the aggregate, have a material adverse effect on the Partnership.
- 3.3. The Interests.
- (a) Victory holds of record and owns beneficially all of the Interests, free and clear of all Liens (other than Permitted Liens), except for Liens that will be removed at the Closing.
- (b) Victory is not party to any contract or agreement obligating Victory to vote or dispose of any Interests.
- 3.4. Balance Sheet. Attached as Schedule 3.4 is a true and correct copy of the balance sheet of the Partnership.
- 3.5. Brokers' Fees. Victory does not have any Liability to pay any fees or commissions to any broker, finder or agent with respect to this Agreement, the Acquisition or the transactions contemplated by this Agreement.

ARTICLE IV

REPRESENTATIONS AND WARRANTIES OF NAVITUS

Navitus represents and warrants to Victory as follows:

4.1. Organization; Authority and Enforceability. Navitus is a general partnership that is validly existing and in good standing under the Laws of the State of Texas. Navitus has the requisite power and authority to execute and deliver this Agreement, to perform its obligations hereunder and to consummate the Acquisition and the other transactions contemplated hereby. The execution, delivery and performance by Navitus of this Agreement and the consummation by Navitus of the Acquisition and the other transactions contemplated hereby have been duly authorized by all necessary action on the part of Navitus and no other action is necessary on the part of Navitus to authorize this Agreement or to consummate the Acquisition or the other transactions contemplated hereby. This Agreement has been duly executed and delivered by Navitus and, assuming the due authorization, execution and delivery by each other party hereto, constitutes a legal, valid and binding obligation of Navitus, enforceable against Navitus in accordance with its terms, except as limited by (a) bankruptcy, insolvency, reorganization, moratorium, fraudulent conveyance or other similar Laws relating to creditors' rights generally and (b) general principles of equity, whether such enforceability is considered in a proceeding in equity or at Law.

4.2. Noncontravention.

- (a) Neither the execution and the delivery of this Agreement nor the consummation of the Acquisition or the other transactions contemplated by this Agreement, will, with or without the giving of notice or the lapse of time or both, (i) to the actual knowledge of Navitus violate any Law applicable to Navitus or (ii) violate any contract or agreement to which Navitus is a party, except in the case of clauses (i) and (ii) to the extent that any such violation would not reasonably be expected to have, individually or in the aggregate, a material adverse effect upon Victory.
- (b) The execution and delivery of this Agreement by Navitus does not, and the performance of this Agreement by Navitus will not, require any consent, approval, authorization or Permit of, or filing with or notification to, any Governmental Entity, except for (i) the filings which will be made at or prior to the Closing or following the Closing, but within required time limits or (ii) where the failure to take such action would not reasonably be expected to have, individually or in the aggregate, a material adverse effect upon Victory.
- 4.3. Brokers' Fees. Navitus does not have any Liability to pay any fees or commissions to any broker, finder or agent with respect to this Agreement, the Acquisition or the transactions contemplated by this Agreement.

ARTICLE V

COVENANTS

- 5.1. Consents. Each of Victory and Navitus, respectively, will use their commercially reasonable efforts to obtain any required third-party consents to the Divestiture and the other transactions contemplated by this Agreement in writing.
- 5.2. Taking of Necessary Action; Further Action. Subject to the terms and conditions of this Agreement, each of Victory and Navitus will take all such reasonable and lawful action as may be necessary or appropriate in order to effectuate the Divestiture in accordance with this Agreement as promptly as practicable.
- 5.3. Stockholders Meeting; Preparation of Proxy Statement.
- (a) Subject to the terms set forth in this Agreement, Victory shall take all action necessary to duly call, give notice of, convene and hold a stockholders meeting relating to the approval of Divestiture and such other items as Victory may seek stockholder approval of (the "Stockholders Meeting") as soon as reasonably practicable after the date of this Agreement, and, in connection therewith, Victory shall mail a proxy statement (the "Proxy Statement") to the holders of Common Stock in advance of such meeting. Victory shall use reasonable best efforts to (i) solicit from the holders of Common Stock proxies in favor of the approval of the Divestiture and (ii) take all other actions necessary or advisable to secure the vote or consent of the holders of Common Stock required by applicable Law to obtain such approval. Notwithstanding anything contained herein to the contrary, Victory shall not be required to hold the Stockholders Meeting if this Agreement is terminated before the meeting is held.
- (b) In connection with the Stockholders Meeting, as soon as reasonably practicable following the date of this Agreement Victory shall prepare and file the Proxy Statement with the U.S. Securities and Exchange Commission (the "Commission"). Without limiting the generality of the foregoing, Victory will furnish Navitus the information relating to it required by the Exchange Act and the rules and regulations promulgated thereunder to be set forth in the Proxy Statement. Victory shall use its reasonable best efforts to resolve, and each party agrees to consult and cooperate with the other party in resolving, all Commission comments with respect to the Proxy Statement as promptly as practicable after receipt thereof and to cause the Proxy Statement in definitive form to be cleared by the Commission and mailed to the Victory's stockholders as promptly as reasonably practicable following filing with the Commission.
- 5.4. Repricing of Warrants. As soon as reasonably practicable following the Closing Victory shall take such steps as may be necessary to amend the exercise price of the warrants of Victory that are listed on Schedule 5.4 to reflect an exercise price \$0.04. Navitus acknowledges and agrees that such repricing of the Warrants may require Victory to make certain filings with Commission, including the filing of a Schedule TO and that such filings may be reviewed by the Commission, which may result in delays in effectuating such repricing.

ARTICLE VI

CONDITIONS TO OBLIGATIONS TO CLOSE

- 6.1. Conditions to the Obligation of Parties. The respective obligations of the parties to consummate transactions contemplated hereby are subject to the satisfaction or waiver of the following conditions:
- (a) No temporary, preliminary or permanent restraining order preventing the consummation of the transactions contemplated hereby will be in effect.

- (b) The parties shall have received the authorizations, consents and approvals of, or otherwise notified, the Governmental Entities as necessary to consummate the transactions contemplated hereby.
- (c) The parties shall have received the authorizations, consents and approvals of, third parties necessary to accomplish the transactions contemplated hereby.
- (d) Navitus and Victory shall have concluded and implemented recommendations made by their appointed tax professionals.
- (e) Shareholder Approval shall have been obtained.
- 6.2. Conditions to Obligation of Navitus. The obligation of Navitus to consummate the transactions contemplated hereunder is subject to the satisfaction or waiver by Navitus of the following conditions:
- (a) The representations and warranties of Victory set forth in this Agreement will be true and correct in all respects as of the date of this Agreement and as of the Closing Date (except to the extent such representations and warranties speak as of another date, in which case such representations and warranties will be true and correct as of such other date), except where the failure of such representations and warranties to be so true and correct does not adversely affect the ability of the Victory to consummate the Divestiture and the other transactions contemplated by this Agreement.
- (b) Victory will have performed all of the covenants required to be performed by it under this Agreement at or prior to the Closing, except where the failure to perform does not have, and would not reasonably be expected to have, individually or in the aggregate, a material adverse effect on the Partnership or materially adversely affect the ability of Victory to consummate the Acquisition or perform its other obligations hereunder.
- (c) Navitus shall have received a certificate signed by Victory to the effect of Sections 6.2(a) and 6.2(b) above.
- (d) Navitus shall have received the Release duly executed by Victory.
- (e) Navitus shall have received the Amended Partnership Agreement duly executed by Victory.
- 6.3. Conditions to Obligation of Victory. The obligation of Victory to consummate the transactions contemplated hereby is subject to the satisfaction or waiver by Victory of the following conditions:
- (a) The representations and warranties of Navitus set forth in this Agreement will be true and correct in all respects as of the date of this Agreement and as of the Closing Date (except to the extent such representations and warranties speak as of another date, in which case such representations and warranties will be true and correct as of such other date).
- (b) Navitus will have performed in all material respects all of the covenants required to be performed by it under this Agreement at or prior to the Closing, except such failures to perform as do not materially adversely affect the ability of Navitus to consummate the Acquisition and the other transactions contemplated by this Agreement.
- (c) Victory shall have received a certificate signed by Navitus to the effect of Sections 6.3(a) and 6.3(b) above.
- (d) Victory shall have received the Release duly executed by Navitus.
- (e) Victory shall have received the Amended Partnership Agreement duly executed by Navitus.

ARTICLE VII

TERMINATION; AMENDMENT; WAIVER

- 7.1. Termination of Agreement. This Agreement may be terminated as follows:
- (a) by mutual written consent of Victory and Navitus at any time prior to the Closing;
- (b) by either Victory or Navitus if any Governmental Entity will have issued an order or taken any other action permanently enjoining, restraining or otherwise prohibiting the transactions contemplated by this Agreement;
- (c) by Navitus if Victory has breached its representations and warranties or any covenant or other agreement to be performed by it in a manner such that the Closing conditions set forth in Section 6.2(a) or 6.2(b) would not be satisfied;
- (d) by Victory if Navitus has breached its representations and warranties or any covenant or other agreement to be performed by it in a manner such that the Closing conditions set forth in Section 6.3(a) or 6.3(b) would not be satisfied; or
- (e) by Navitus if (i) all of the conditions set forth in Section 6.1 and Section 6.3 have been satisfied (other than those conditions that by their nature are to be satisfied at the Closing or on the Closing Date provided that such conditions are reasonably capable of being satisfied at the Closing or on the Closing Date), and Victory fails to consummate the Divestiture on the date the Closing should have occurred pursuant to Section 2.2 and (ii) Navitus has irrevocably confirmed in writing that all conditions set forth in Section 6.1 and Section 6.2 have been satisfied or that it is willing

to waive all unsatisfied conditions in Section 6.2 and it stands ready, willing and able to consummate the Closing on such date.

- (f) by Victory if (i) all of the conditions set forth in Section 6.1 and Section 6.2 have been satisfied (other than those conditions that by their nature are to be satisfied at the Closing or on the Closing Date provided that such conditions are reasonably capable of being satisfied at the Closing or on the Closing Date), and Navitus fails to consummate the Divestiture on the date the Closing should have occurred pursuant to Section 2.2, and (ii) Victory has irrevocably confirmed in writing that all conditions set forth in Section 6.1 and Section 6.3 have been satisfied or that they are willing to waive all unsatisfied conditions in Section 6.3 and stand ready, willing and able to consummate the Closing on such date.
- 7.2. Effect of Termination. In the event of termination of this Agreement by Victory or by Navitus as provided in Section 7.1, this Agreement will forthwith become void and have no effect, without any Liability (other than with respect to any suit for breach of this Agreement) on the part of Navitus or Victory (or any member, partner, stockholder agent, consultant or representative of any such party); provided, that the provisions of Sections 8.1, 8.6, 8.7, 8.8, 8.11, 8.13, and this Section 7.2 will survive any termination hereof pursuant to Section 7.1.
- 7.3. Amendments. This Agreement may be amended by the parties hereto, by action taken or authorized by, in the case of Victory, by Victory's Board of Directors and, in the case of Navitus, by Navitus and its partners, as applicable. This Agreement may not be amended except by an instrument in writing signed on behalf of Victory and Navitus.
- 7.4. Waiver. At any time prior to the Closing, either Party may extend the time for the performance of any of the covenants, obligations or other acts of the other Party or (b) waive any inaccuracy of any representations or warranties or compliance with any of the agreements, covenants or conditions of the other Party or any conditions to its own obligations. Any agreement on the part of a Party to any such extension or waiver will be valid only if such waiver is set forth in an instrument in writing signed on its behalf by its duly authorized officer. The failure of any party to this Agreement to assert any of its rights under this Agreement or otherwise will not constitute a waiver of such rights. The waiver of any such right with respect to particular facts and other circumstances will not be deemed a waiver with respect to any other facts and circumstances, and each such right will be deemed an ongoing right that may be asserted at any time and from time to time.

ARTICLE VIII MISCELLANEOUS

- 8.1. Press Releases and Public Announcement. Neither Navitus on the one hand, nor Victory on the other, will issue any press release or make any public announcement relating to this Agreement, the Divestiture or the other transactions contemplated by this Agreement without the prior written approval of the other party; provided, however, that Victory may make regulatory filings referring to this Agreement or attaching a copy hereof as may be required by applicable Law.
- 8.2. No Third-Party Beneficiaries. This Agreement will not confer any rights or remedies upon any Person other than the parties hereto and their respective successors and permitted assigns.
- 8.3. Succession and Assignment. This Agreement will be binding upon and inure to the benefit of the parties named herein and their respective successors and permitted assigns. No party hereto may assign either this Agreement or any of its rights, interests or obligations hereunder without the prior written approval, in the case of assignment by Navitus, by Victory, and, in the case of assignment by Victory, by Navitus.
- 8.4. Construction. The parties have participated jointly in the negotiation and drafting of this Agreement, and, in the event an ambiguity or question of intent or interpretation arises, this Agreement will be construed as if drafted jointly by the parties, and no presumption or burden of proof will arise favoring or disfavoring any party by virtue of the authorship of any of the provisions of this Agreement.
- 8.5. Notices. All notices, requests and other communications hereunder must be in writing and will be deemed to have been duly given only if delivered personally against written receipt or by facsimile transmission or mailed (by registered or certified mail, postage prepaid, return receipt requested) or delivered by reputable overnight courier, fee prepaid, to the parties hereto at the addresses of the parties as specified on the signature pages hereto. Any party may change the address to which notices, requests, demands, claims and other communications hereunder are to be delivered by giving the other parties notice in the manner set forth herein.
- 8.6. Governing Law. This Agreement will be governed by, and construed in accordance with, the Laws of the State of Texas, without giving effect to any choice of Law or conflict of Law provision or rule that would cause the application of the Laws of any jurisdiction other than the State of Texas.

- 8.7. Consent to Jurisdiction and Service of Process. EACH OF THE PARTIES HERETO CONSENTS TO THE JURISDICTION OF ANY STATE OR FEDERAL COURT LOCATED WITHIN THE STATE OF TEXAS AND IRREVOCABLY AGREES THAT ALL ACTIONS OR PROCEEDINGS RELATING TO THIS AGREEMENT, THE ACQUISITION OR THE OTHER TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT MAY BE LITIGATED IN SUCH COURTS. EACH OF THE PARTIES HERETO ACCEPTS FOR ITSELF AND IN CONNECTION WITH ITS RESPECTIVE PROPERTIES, GENERALLY AND UNCONDITIONALLY, THE EXCLUSIVE JURISDICTION OF THE AFORESAID COURTS AND WAIVES ANY DEFENSE OF FORUM NON CONVENIENS, AND IRREVOCABLY AGREES TO BE BOUND BY ANY FINAL AND NONAPPEALABLE JUDGMENT RENDERED THEREBY IN CONNECTION WITH THIS AGREEMENT, THE ACOUISITION OR THE OTHER TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT. EACH OF THE PARTIES HERETO FURTHER IRREVOCABLY CONSENTS TO THE SERVICE OF PROCESS OUT OF ANY OF THE AFOREMENTIONED COURTS IN ANY SUCH ACTION OR PROCEEDING BY THE MAILING OF COPIES THEREOF BY REGISTERED OR CERTIFIED MAIL, POSTAGE PREPAID, TO SUCH PARTY AT THE ADDRESS SPECIFIED IN THIS AGREEMENT, SUCH SERVICE TO BECOME EFFECTIVE 15 CALENDAR DAYS AFTER SUCH MAILING. NOTHING HEREIN WILL IN ANY WAY BE DEEMED TO LIMIT THE ABILITY OF ANY PARTY HERETO TO SERVE ANY SUCH LEGAL PROCESS, SUMMONS, NOTICES AND DOCUMENTS IN ANY OTHER MANNER PERMITTED BY APPLICABLE LAW OR TO OBTAIN JURISDICTION OVER OR TO BRING ACTIONS, SUITS OR PROCEEDINGS AGAINST ANY OTHER PARTY HERETO IN SUCH OTHER JURISDICTIONS, AND IN SUCH MANNER, AS MAY BE PERMITTED BY ANY APPLICABLE LAW.
- 8.8. Headings. The descriptive headings contained in this Agreement are included for convenience of reference only and will not affect in any way the meaning or interpretation of this Agreement.
- 8.9. Severability. If any provision of this Agreement is held to be illegal, invalid or unenforceable under any present or future Law (a) such provision will be fully severable, (b) this Agreement will be construed and enforced as if such illegal, invalid or unenforceable provision had never comprised a part hereof, (c) the remaining provisions of this Agreement will remain in full force and effect and will not be affected by the illegal, invalid or unenforceable provision or by its severance herefrom and (d) in lieu of such illegal, invalid or unenforceable provision, there will be added automatically as a part of this Agreement a legal, valid and enforceable provision as similar in terms of such illegal, invalid or unenforceable provision as may be possible.
- 8.10. Expenses. Except as otherwise provided in this Agreement, including Section 5.5 of this Agreement, whether or not the Acquisition is consummated, all Expenses incurred in connection with this Agreement and the transactions contemplated hereby will be paid by the party incurring such Expenses. As used in this Agreement, "Expenses" means the out-of-pocket fees and expenses of the financial advisor, counsel and accountants incurred in connection with this Agreement and the transactions contemplated hereby.
- 8.11. Incorporation of Exhibits and Schedules. The Exhibits and Schedules identified in this Agreement are incorporated herein by reference and made a part hereof.
- 8.12. Limited Recourse. Notwithstanding anything in this Agreement to the contrary, the obligations and Liabilities of the parties hereunder will be without recourse to any stockholder, partner or member of such party or any of such stockholder's, partner's or member's Affiliates (other than such party), or any of their respective representatives or agents (in each case, in their capacity as such).
- 8.13. Specific Performance. The parties hereto agree that irreparable damage would occur in the event that any provision of this Agreement was not performed in accordance with the terms hereof and that the parties will be entitled to specific performance of the terms hereof in addition to any other remedy at Law or equity.
- 8.14. Counterparts. This Agreement may be executed in two (2) or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument. Counterparts may be delivered via facsimile, electronic mail (including pdf or any electronic signature complying with the U.S. federal ESIGN Act of 2000, e.g., www.docusign.com) or other transmission method and any counterpart so delivered shall be deemed to have been duly and validly delivered and be valid and effective for all purposes.
- 8.15. No Survival of Representations and Warranties. The representations and warranties made herein and in any certificate delivered in connection herewith shall expire as of the Closing.

 [REMAINDER OF PAGE INTENTIONALLY LEFT BLANK]

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed as of the date first set forth above.

VICTORY ENERGY CORPORATION

By:/s/ Kenneth Hill
Name: Kenneth Hill

Title: Chief Executive Officer

Address: 3355 Bee Caves Road, Suite 608 Austin, TX 78746

Phone Fax: Email:

Attention: Kenneth Hill

With a copy (which shall not constitute notice) to:

BEVILACQUA PLLC

1050 Connecticut Street, NW, Suite 500

Washington, DC 20036

Attn: Louis A. Bevilacqua, Esq.

Email:

NAVITUS ENERGY GROUP

BY: JAMES CAPITAL CONSULTING, LLC,

its Managing Partner

By:/s/ Ronald Zamber Name: Ronald Zamber Title: Managing Member

Address: Fax No.:

Attention: EXHIBIT A

Form of Release

(See Attached)

MUTUAL RELEASE

MUTUAL RELEASE, dated ______, 2017 (this "Release"), by and between Navitus Energy Group, a Texas general partnership ("Navitus") and Victory Energy Corporation, a Nevada corporation ("Victory"). Each of Navitus and Victory are referred to herein as a "Party" and, collectively, as the "Parties."

RECITALS

- A. Navitus and Victory are parties to a Divestiture Agreement, dated August 21, 2017 (the "Divestiture Agreement").
- B. It is a condition to the closing of the transactions contemplated by the Divestiture Agreement that the Parties enter into this Release providing for the mutual release contained herein.

AGREEMENT

NOW, THEREFORE, the Parties hereto, in exchange for the mutual promises herein contained, intending to be legally bound, hereby agree as follows:

- 1. MUTUAL RELEASE. Each of the Parties, on behalf of themselves, and all persons or entities claiming by, through or under them, and their respective successors and assigns, hereby fully, completely and finally waive, release, remise, acquit, and forever discharge and covenant not to sue the other Party, as well as the other Party's respective officers, directors, shareholders, partners, trustees, parent companies, sister companies, affiliates, subsidiaries, employers, attorneys, accountants, predecessors, successors, insurers, representatives, and agents with respect to any and all claims, demands, suits, manner of obligation, debt, liability, tort, covenant, contract, or causes of action of any kind whatsoever, at law or in equity, including without limitation, all claims and causes of action arising out of or in any way relating to any dealings between the Parties prior to the date hereof relating to or arising under the Partnership Agreement (as defined in the Divestiture Agreement), except that Navitus is not releasing or otherwise discharging Victory of any obligations that Victory may have resulting from the ownership by Navitus, its partners, or any of its affiliates, of any capital stock, options to purchase capital stock or warrants to purchase capital stock held by Navitus, its partners, or any such affiliates. The Parties warrant and represent that they have not assigned or otherwise transferred any claim or cause of action released by this Release. The Parties acknowledge and agree that these releases are GENERAL RELEASES. The Parties expressly waive and assume the risk of any and all claims for damages which exist as of this date, but which they do not know or suspect to exist, whether through ignorance, oversight, error, negligence, or otherwise, and which, if known, would materially affect its decision to enter into this Release The Parties expressly acknowledge that this waiver of claims includes any claims for any alleged fraud, deception, concealment, misrepresentation or any other misconduct of any kind in procuring this Release. The Parties specifically do not, however, waive or release any claim that may arise for breach of this Release.
- 2. NO ADMISSION OF LIABILITY. Neither the payment of any sums nor the execution of this Release shall be construed as an admission of liability or fault by any Party. Any and all liability is expressly denied by both Parties.
- 3. AUTHORITY. The Parties represent and warrant that they possess full authority to enter into this Release and to lawfully and effectively release the opposing Party as set forth herein, free of any rights of settlement, approval, subrogation, or other condition or impediment. This undertaking includes specifically, without limitation, the representation and warranty that no third party has now acquired or will acquire rights to present or pursue any claims arising from or based upon the claims that have been released herein.
- 4. ENTIRE AGREEMENT. The Parties represent and agree that no promise, inducement, or agreement other than as expressed herein has been made to them and that this Release is fully integrated, supersedes all prior agreements and understandings, and any other agreement between the Parties, and contains the entire agreement between the Parties.
- 5. VOLUNTARY AND INFORMED ASSENT. The Parties represent and agree that they each have read and fully understand this Release, that they are fully competent to enter into and sign this Release, and that they are executing this Release voluntarily, free of any duress or coercion.
- 6. COSTS, EXPENSES AND ATTORNEYS' FEES. Each of the Parties will bear its own costs, expenses, and attorneys' fees incurred in connection with the transactions and dealings between the Parties prior to the date hereof.
- 7. GOVERNING LAW AND JURISDICTION. The laws of the State of Texas shall apply to and control any interpretation, construction, performance or enforcement of this Release. The Parties agree that the exclusive jurisdiction for any legal proceeding arising out of or relating to this Release shall be the State or Federal courts located in Travis County, Texas and the Parties hereby waive any challenge to personal jurisdiction or venue in that court.
- 8. ATTORNEYS' FEES AND COSTS FOR BREACH. The prevailing party in any action to enforce or interpret this Release is entitled to recover from the other party its reasonable attorneys' fees.
- 9. CONSTRUCTION. This Release shall be construed as if the parties jointly prepared it, and any uncertainty or ambiguity shall not be interpreted against any one party.
- 10. MODIFICATION. No oral agreement, statement, promise, undertaking, understanding, arrangement, act or omission of any Party, occurring subsequent to the date hereof may be deemed an amendment or modification of this Release unless reduced to writing and signed by the Parties hereto or their respective successors or assigns.
- 11. SEVERABILITY. The Parties agree that if, for any reason, a provision of this Release is held unenforceable by any court of competent jurisdiction, this Release shall be automatically conformed to the law, and otherwise this Release shall continue in full force and effect.

- 12. NUMBER. Whenever applicable within this Release, the singular shall include the plural and the plural shall include the singular.
- 13. HEADINGS. The headings of paragraphs herein are included solely for convenience of reference and shall not control the meaning or interpretation of any of the provisions of this Release.
- 14. COUNTERPARTS. This Release may be executed in several counterparts and all counterparts so executed shall constitute one agreement binding on all parties hereto, notwithstanding that all the parties are not signatories to the original or the same counterpart. Facsimile signatures shall be accepted the same as an original signature. A photocopy of this Release may be used in any action brought to enforce or construe this Release.
- 15. NO WAIVER. No failure to exercise and no delay in exercising any right, power or remedy under this Release shall impair any right, power or remedy which any Party may have, nor shall any such delay be construed to be a waiver of any such rights, powers or remedies or an acquiescence in any breach or default under this Release, nor shall any waiver of any breach or default of any Party be deemed a waiver of any default or breach subsequently arising.

IN WITNESS WHEREOF, the parties hereto have caused this Release to be duly executed as of the date first set forth above.
VICTORY ENERGY CORPORATION
By: Name: Kenneth Hill Title: Chief Executive Officer
Address: 3355 Bee Caves Road, Suite 608 Austin, TX 78746 Phone: Fax: Email: Attention: Kenneth Hill
With a copy (which shall not constitute notice) to: BEVILACQUA PLLC 1050 Connecticut Street, NW, Suite 500 Washington, DC 20036 Attn: Louis A. Bevilacqua, Esq. email:
NAVITUS ENERGY GROUP
By: Name: Title:

Address: Fax No.:

Attention:

AMENDMENT NO. 1 TO THE DIVESTITURE AGREEMENT AMENDMENT NO. 1 TO THE DIVESTITURE AGREEMENT, dated September 14, 2017 (the "Amendment"), between VICTORY ENERGY CORPORATION, a Nevada corporation ("Victory") and NAVITUS ENERGY GROUP,

a Texas general partnership ("Navitus").

RECITALS

- C. Victory and Navitus previously entered into that certain Divestiture Agreement, dated August 21, 2017 (the "Divestiture Agreement").
- D. Victory and Navitus desire to amend the Divestiture Agreement to provide for additional consideration from Victory to Navitus in consideration for the Release and to provide Navitus with registration rights as described herein.
- E. Pursuant to Section 7.3 of the Divestiture Agreement, the Divestiture Agreement may be amended by the parties thereto, by action taken or authorized by, in the case of Victory, by Victory's Board of Directors and, in the case of Navitus, by Navitus and its partners, as applicable, and by an instrument in writing signed on behalf of Victory and Navitus.
- F. This Amendment has been authorized by Victory's Board of Directors and by Navitus and its partners. AGREEMENT
- NOW, THEREFORE, for good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the parties hereto, intending to be legally bound, hereby agree as follows:
- 1.Definitions. All capitalized terms used herein without definition shall have the meanings ascribed to them in the Divestiture Agreement.
- 2. Amendments.
- (a) Section 2.1 of the Divestiture Agreement shall be amended and restated in its entirety as follows:
- "Divestiture and Issuance. Upon the terms and subject to the conditions set forth in this Agreement, Victory will (i) sell, transfer and deliver to Navitus free and clear of all Liens, indebtedness and other Liabilities, and Navitus will acquire from Victory, the Interest, and (ii) upon obtaining Shareholder Approval, issue to Navitus 166,549,134 shares of Victory's Common Stock (or 4,382,872 shares of Common Stock following the Victory's planned 1-for-38 reverse stock split) (the "Navitus Shares"), in each case, in exchange for the Release. At or prior to the Closing, Victory shall pay off or otherwise satisfy all indebtedness and other Liabilities of the Partnership specifically listed on Schedule 2.1 hereto, such that the Partnership shall own all of its assets free and clear of all Liens other than Permitted Liens."

 (b) The addition of Section 2.4 of the Divestiture Agreement shall be added in its entirety as follows:
- "Registration Rights. Victory hereby grants the following registration rights to Navitus:
- (a)Registration Statement. At any time following the issuance of the Navitus Shares, Navitus may request that Victory file with the Securities and Exchange Commission (the "SEC") within thirty (30) days of such request a registration statement on an appropriate form (the "Registration Statement") covering the resale of the Navitus Shares and shall use its commercially reasonable efforts to cause the Registration Statement to be declared effective within one hundred twenty (120) days following such filing. Notwithstanding anything to the contrary herein, at any time, Victory may delay the filing of the Registration for a reasonable period of time (the "Grace Period") if it would result in the disclosure of material, non-public information concerning Victory the disclosure of which at the time is not, in the good faith opinion of Victory's Board of Directors, in Victory's best interest and otherwise required; provided, that Victory shall promptly (i) notify Navitus in writing of the existence of material, non-public information giving rise to a Grace Period (provided that in each notice Victory will not disclose the content of such material, non-public information to Navitus unless requested by Navitus and subject to a confidentiality agreement) and the date on which the Grace Period will begin, and (ii) use commercially reasonable efforts to resolve any issue that makes disclosure of
- (b) Registration Procedures. In connection with the Registration Statement, Victory will:

the material, non-public information not in Victory's best interests.

- (i) Prepare and file with the SEC such amendments and supplements to the Registration Statement and the prospectus used in connection therewith as may be necessary to keep such Registration Statement effective with respect to Navitus until all the Navitus Shares may be resold without restriction under the Securities Act; and
- (ii) Immediately notify Navitus when the prospectus included in the Registration Statement is required to be delivered under the Securities Act, of the happening of any event of which Victory has knowledge as a result of which the prospectus contained in such Registration Statement, as then in effect, includes an untrue statement of a material fact or omits to state a material fact required to be stated therein or necessary to make the statements therein not misleading in light of the circumstances then existing. If Victory notifies Navitus to suspend the use of any prospectus until the requisite changes to such prospectus have been made, then Navitus shall suspend use of such prospectus. In such event, Victory will use its commercially reasonable efforts to update such prospectus as promptly as is

practicable.

- (c) Provision of Documents etc. In connection with the Registration Statement, Navitus will furnish to Victory in writing such information and representation letters with respect to itself and the proposed distribution by it as reasonably shall be necessary in order to assure compliance with federal and applicable state securities laws. Navitus covenants and agrees that it will comply with the prospectus delivery requirements of the Securities Act, if applicable, in connection with sales of Navitus Shares pursuant to the Registration Statement.
- (d) Victory shall be entitled to include in any Registration Statement referred to in this Section 2.4, shares of Common Stock to be sold by Victory for its own account (to the extent that the inclusion of such shares by Victory shall not adversely affect the offering).
- (e) Expenses. All expenses incurred by Victory in complying with this Section 2.4, including, without limitation, all registration and filing fees, printing expenses, fees and disbursements of counsel and independent public accountants for Victory, fees of transfer agents and registrars shall be borne by Victory. All underwriting discounts and selling commissions applicable to the sale of the Navitus Shares, including any fees and disbursements of any counsel to Navitus, shall be borne by Navitus.
- 3.Effect of Amendment. Except as amended as set forth above, the Divestiture Agreement shall continue in full force and effect.
- 4. Counterparts. This Amendment may be executed in two (2) or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument. Counterparts may be delivered via facsimile, electronic mail (including pdf or any electronic signature complying with the U.S. federal ESIGN Act of 2000, e.g., www.docusign.com) or other transmission method and any counterpart so delivered shall be deemed to have been duly and validly delivered and be valid and effective for all purposes.
- 5.Governing Law. This Agreement will be governed by, and construed in accordance with, the Laws of the State of Texas, without giving effect to any choice of Law or conflict of Law provision or rule that would cause the application of the Laws of any jurisdiction other than the State of Texas.

[REMAINDER OF PAGE INTENTIONALLY LEFT BLANK]

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be duly executed as of the date first set forth above.

VICTORY ENERGY CORPORATION

By: /s/ Kenneth Hill Name: Kenneth Hill

Title: Chief Executive Officer

NAVITUS ENERGY GROUP BY: JAMES CAPITAL CONSULTING, LLC, its Managing Partner

By: /s/ Ronald Zamber Name: Ronald Zamber

Title: Managing Member

ANNEX C VICTORY ENERGY CORPORATION 2017 EQUITY INCENTIVE PLAN 1. Purpose; Eligibility.

- 1.1. General Purpose. The name of this plan is the Victory Energy Corporation 2017 Equity Incentive Plan (the "Plan"). The purposes of the Plan are to (a) enable Victory Energy Corporation, a Nevada corporation (the "Company"), and any Affiliate to attract and retain the types of Employees, Consultants and Directors who will contribute to the Company's long-term success; (b) provide incentives that align the interests of Employees, Consultants and Directors with those of the stockholders of the Company; and (c) promote the success of the Company's business.
- 1.2. Eligible Award Recipients. The persons eligible to receive Awards are the Employees, Consultants and Directors of the Company and its Affiliates and such other individuals designated by the Committee who are reasonably expected to become Employees, Consultants and Directors after the receipt of Awards.
- 1.3. Available Awards. Awards that may be granted under the Plan include: (a) Incentive Stock Options, (b) Non-qualified Stock Options, (c) Stock Appreciation Rights, (d) Restricted Awards, (e) Performance Share Awards, and (f) Performance Compensation Awards.
- 2. Definitions.
- "Affiliate" means a corporation or other entity that, directly or through one or more intermediaries, controls, is controlled by or is under common control with, the Company.
- "Applicable Laws" means the requirements related to or implicated by the administration of the Plan under applicable state corporate law, United States federal and state securities laws, the Code, any stock exchange or quotation system on which the shares of Common Stock are listed or quoted, and the applicable laws of any foreign country or jurisdiction where Awards are granted under the Plan.
- "Award" means any right granted under the Plan, including an Incentive Stock Option, a Non-qualified Stock Option, a Stock Appreciation Right, a Restricted Award, a Performance Share Award or a Performance Compensation Award. "Award Agreement" means a written agreement, contract, certificate or other instrument or document evidencing the terms and conditions of an individual Award granted under the Plan which may, in the discretion of the Company, be transmitted electronically to any Participant. Each Award Agreement shall be subject to the terms and conditions of the Plan.
- "Beneficial Owner" has the meaning assigned to such term in Rule 13d-3 and Rule 13d-5 under the Exchange Act, except that in calculating the beneficial ownership of any particular "person" (as that term is used in Section 13(d)(3) of the Exchange Act), such "person" shall be deemed to have beneficial ownership of all securities that such "person" has the right to acquire by conversion or exercise of other securities, whether such right is currently exercisable or is exercisable only after the passage of time. The terms "Beneficially Owns" and "Beneficially Owned" have a corresponding meaning.
- "Board" means the Board of Directors of the Company, as constituted at any time.
- "Cause" means:

With respect to any Employee or Consultant: (a) if the Employee or Consultant is a party to an employment or service agreement with the Company or its Affiliates and such agreement provides for a definition of Cause, the definition contained therein; or (b) if no such agreement exists, or if such agreement does not define Cause: (i) the commission of, or plea of guilty or no contest to, a felony or a crime involving moral turpitude or the commission of any other act involving willful malfeasance or material fiduciary breach with respect to the Company or an Affiliate; (ii) conduct that results in or is reasonably likely to result in harm to the reputation or business of the Company or any of its Affiliates; (iii) gross negligence or willful misconduct with respect to the Company or an Affiliate; or (iv) material violation of state or federal securities laws.

With respect to any Director, a determination by a majority of the disinterested Board members that the Director has engaged in any of the following: (a) malfeasance in office; (b) gross misconduct or neglect; (c) false or fraudulent misrepresentation inducing the director's appointment; (d) willful conversion of corporate funds; or (e) repeated failure to participate in Board meetings on a regular basis despite having received proper notice of the meetings in advance. The Committee, in its absolute discretion, shall determine the effect of all matters and questions relating to whether a Participant has been discharged for Cause.

"Change in Control" means (a) the direct or indirect sale, transfer, conveyance or other disposition (other than by way of merger or consolidation), in one or a series of related transactions, of all or substantially all of the properties or assets of the Company and its subsidiaries, taken as a whole, to any Person that is not a subsidiary of the Company; (b) the Incumbent Directors cease for any reason to constitute at least a majority of the Board; (c) the date which is 10 business days prior to the consummation of a complete liquidation or dissolution of the Company; (d) the acquisition

by any Person of Beneficial Ownership of more than 50% (on a fully diluted basis) of either (i) the then outstanding shares of Common Stock of the Company, taking into account as outstanding for this purpose such Common Stock issuable upon the exercise of options or warrants, the conversion of convertible stock or debt, and the exercise of any similar right to acquire such Common Stock (the "Outstanding Company Common Stock") or (ii) the combined voting power of the then outstanding voting securities of the Company entitled to vote generally in the election of directors (the "Outstanding Company Voting Securities"); provided, however, that for purposes of this Plan, the following acquisitions shall not constitute a Change in Control: (A) any acquisition by the Company or any Affiliate, (B) any acquisition by any employee benefit plan sponsored or maintained by the Company or any subsidiary, (C) any acquisition which complies with clauses, (i), (ii) and (iii) of subsection (e) of this definition or (D) in respect of an Award held by a particular Participant, any acquisition by the Participant or any group of persons including the Participant (or any entity controlled by the Participant or any group of persons including the Participant); or (e) the consummation of a reorganization, merger, consolidation, statutory share exchange or similar form of corporate transaction involving the Company that requires the approval of the Company's stockholders, whether for such transaction or the issuance of securities in the transaction (a "Business Combination"), unless immediately following such Business Combination: (i) more than 50% of the total voting power of (A) the entity resulting from such Business Combination (the "Surviving Company"), or (B) if applicable, the ultimate parent entity that directly or indirectly has beneficial ownership of sufficient voting securities eligible to elect a majority of the members of the board of directors (or the analogous governing body) of the Surviving Company (the "Parent Company"), is represented by the Outstanding Company Voting Securities that were outstanding immediately prior to such Business Combination (or, if applicable, is represented by shares into which the Outstanding Company Voting Securities were converted pursuant to such Business Combination), and such voting power among the holders thereof is in substantially the same proportion as the voting power of the Outstanding Company Voting Securities among the holders thereof immediately prior to the Business Combination; (ii) no Person (other than any employee benefit plan sponsored or maintained by the Surviving Company or the Parent Company) is or becomes the Beneficial Owner, directly or indirectly, of 50% or more of the total voting power of the outstanding voting securities eligible to elect members of the board of directors of the Parent Company (or the analogous governing body) (or, if there is no Parent Company, the Surviving Company); and (iii) at least a majority of the members of the board of directors (or the analogous governing body) of the Parent Company (or, if there is no Parent Company, the Surviving Company) following the consummation of the Business Combination were Board members at the time of the Board's approval of the execution of the initial agreement providing for such Business Combination.

"Code" means the Internal Revenue Code of 1986, as it may be amended from time to time. Any reference to a section of the Code shall be deemed to include a reference to any regulations promulgated thereunder.

"Committee" means a committee of one or more members of the Board appointed by the Board to administer the Plan in accordance with Section 3.3 and Section 3.4.

"Common Stock" means the common stock, \$0.001 par value per share, of the Company, or such other securities of the Company as may be designated by the Committee from time to time in substitution thereof.

"Company" means Victory Energy Corporation, a Nevada corporation, and any successor thereto.

"Consultant" means any individual who is engaged by the Company or any Affiliate to render consulting or advisory services.

"Consultant or Director, is not interrupted or terminated. The Participant's Continuous Service shall not be deemed to have terminated merely because of a change in the capacity in which the Participant renders service to the Company or an Affiliate as an Employee, Consultant or Director or a change in the entity for which the Participant renders such service, provided that there is no interruption or termination of the Participant's Continuous Service; provided further that if any Award is subject to Section 409A of the Code, this sentence shall only be given effect to the extent consistent with Section 409A of the Code. For example, a change in status from an Employee of the Company to a Director of an Affiliate will not constitute an interruption of Continuous Service. The Committee or its delegate, in its sole discretion, may determine whether Continuous Service shall be considered interrupted in the case of any leave of absence approved by that party, including sick leave, military leave or any other personal or family leave of absence. "Covered Employee" has the same meaning as set forth in Section 162(m)(3) of the Code, as interpreted by IRS Notice 2007-49.

"Director" means a member of the Board.

"Disability" means that the Participant is unable to engage in any substantial gainful activity by reason of any medically determinable physical or mental impairment; provided, however, for purposes of determining the term of an Incentive Stock Option pursuant to Section 6.10 hereof, the term Disability shall have the meaning ascribed to it under Section 22(e)(3) of the Code. The determination of whether an individual has a Disability shall be determined under procedures established by the Committee. Except in situations where the Committee is determining Disability for purposes of the term of an Incentive Stock Option pursuant to Section 6.10 hereof within the meaning of Section 22(e)(3) of the Code, the Committee may rely on any determination that a Participant is disabled for purposes of benefits under any long-term disability plan maintained by the Company or any Affiliate in which a Participant participates.

"Effective Date" shall mean the date as of which this Plan is adopted by the Board.

"Employee" means any person, including an Officer or Director, employed by the Company or an Affiliate; provided, that, for purposes of determining eligibility to receive Incentive Stock Options, an Employee shall mean an employee of the Company or a parent or subsidiary corporation within the meaning of Section 424 of the Code. Mere service as a Director or payment of a director's fee by the Company or an Affiliate shall not be sufficient to constitute "employment" by the Company or an Affiliate.

"Exchange Act" means the Securities Exchange Act of 1934, as amended.

"Fair Market Value" means, as of any date, the value of the Common Stock as determined below. If the Common Stock is listed on any established stock exchange or a national market system, including without limitation, the New York Stock Exchange or the Nasdaq Stock Market, the Fair Market Value shall be the closing price of a share of Common Stock (or if no sales were reported the closing price on the date immediately preceding such date) as quoted on such exchange or system on the day of determination, as reported in the Wall Street Journal or similar publication. In the absence of an established market for the Common Stock, the Fair Market Value shall be determined in good faith by the Committee and such determination shall be conclusive and binding on all persons.

"Grant Date" means the date on which the Committee adopts a resolution, or takes other appropriate action, expressly granting an Award to a Participant that specifies the key terms and conditions of the Award or, if a later date is set forth in such resolution, then such date as is set forth in such resolution.

"Incentive Stock Option" means an Option intended to qualify as an incentive stock option within the meaning of Section 422 of the Code.

"Incumbent Directors" means individuals who, on the Effective Date, constitute the Board, provided that any individual becoming a Director subsequent to the Effective Date whose election or nomination for election to the Board was approved by a vote of at least two-thirds of the Incumbent Directors then on the Board (either by a specific vote or by approval of the proxy statement of the Company in which such person is named as a nominee for Director without objection to such nomination) shall be an Incumbent Director. No individual initially elected or nominated as a director of the Company as a result of an actual or threatened election contest with respect to Directors or as a result of any other actual or threatened solicitation of proxies by or on behalf of any person other than the Board shall be an Incumbent Director.

"Negative Discretion" means the discretion authorized by the Plan to be applied by the Committee to eliminate or reduce the size of a Performance Compensation Award in accordance with Section 7.4(d)(iv) of the Plan; provided, that, the exercise of such discretion would not cause the Performance Compensation Award to fail to qualify as "performance-based compensation" under Section 162(m) of the Code.

"Non-Employee Director" means a Director who is a "non-employee director" within the meaning of Rule 16b-3.

"Non-qualified Stock Option" means an Option that by its terms does not qualify or is not intended to qualify as an Incentive Stock Option.

"Officer" means a person who is an officer of the Company within the meaning of Section 16 of the Exchange Act and the rules and regulations promulgated thereunder.

"Option" means an Incentive Stock Option or a Non-qualified Stock Option granted pursuant to the Plan.

"Optionholder" means a person to whom an Option is granted pursuant to the Plan or, if applicable, such other person who holds an outstanding Option.

"Option Exercise Price" means the price at which a share of Common Stock may be purchased upon the exercise of an Option.

"Outside Director" means a Director who is an "outside director" within the meaning of Section 162(m) of the Code and Treasury Regulations Section 1.162-27(e)(3) or any successor to such statute and regulation.

"Participant" means an eligible person to whom an Award is granted pursuant to the Plan or, if applicable, such other person who holds an outstanding Award.

"Performance Compensation Award" means any Award designated by the Committee as a Performance Compensation Award pursuant to Section 7.4 of the Plan.

"Performance Criteria" means the criterion or criteria that the Committee shall select for purposes of establishing the Performance Goal(s) for a Performance Period with respect to any Performance Compensation Award under the Plan. The Performance Criteria that will be used to establish the Performance Goal(s) shall be based on the attainment of specific levels of performance of the Company (or Affiliate, division, business unit or operational unit of the Company) and may include the following: (a) net earnings or net income (before or after taxes); (b) basic or diluted earnings per share (before or after taxes); (c) net revenue or net revenue growth; (d) gross revenue; (e) gross profit or gross profit growth; (f) net operating profit (before or after taxes); (g) return on assets, capital, invested capital, equity, or sales; (h) cash flow (including, but not limited to, operating cash flow, free cash flow, and cash flow return on capital); (i) earnings before or after taxes, interest, depreciation and/or amortization; (j) gross or operating margins; (k) improvements in capital structure; (l) budget and expense management; (m) productivity ratios; (n) economic value added or other value added measurements; (o) share price (including, but not limited to, growth measures and total stockholder return); (p) expense targets; (q) margins; (r) operating efficiency; (s) working capital targets; (t) enterprise value; (u) safety record; (v) completion of acquisitions or business expansion; (w) achieving research and development goals and milestones; (x) achieving product commercialization goals; and (y) other criteria as may be set by the Committee from time to time.

Any one or more of the Performance Criteria may be used on an absolute or relative basis to measure the performance of the Company and/or an Affiliate as a whole or any division, business unit or operational unit of the Company and/or an Affiliate or any combination thereof, as the Committee may deem appropriate, or as compared to the performance of a group of comparable companies, or published or special index that the Committee, in its sole discretion, deems appropriate, or the Committee may select Performance Criterion (o) above as compared to various stock market indices. The Committee also has the authority to provide for accelerated vesting of any Award based on the achievement of Performance Goals pursuant to the Performance Criteria specified in this paragraph. To the extent required under Section 162(m) of the Code, the Committee shall, within the first 90 days of a Performance Period (or, if longer or shorter, within the maximum period allowed under Section 162(m) of the Code), define in an objective fashion the manner of calculating the Performance Criteria it selects to use for such Performance Period. In the event that applicable tax and/or securities laws change to permit the Committee discretion to alter the governing Performance Criteria without obtaining stockholder approval.

"Performance Formula" means, for a Performance Period, the one or more objective formulas applied against the relevant Performance Goal to determine, with regard to the Performance Compensation Award of a particular Participant, whether all, some portion but less than all, or none of the Performance Compensation Award has been earned for the Performance Period.

"Performance Goals" means, for a Performance Period, the one or more goals established by the Committee for the Performance Period based upon the Performance Criteria. The Committee is authorized at any time during the first 90 days of a Performance Period (or, if longer or shorter, within the maximum period allowed under Section 162(m) of the Code), or at any time thereafter (but only to the extent the exercise of such authority after such period would not cause the Performance Compensation Awards granted to any Participant for the Performance Period to fail to qualify as "performance-based compensation" under Section 162(m) of the Code), in its sole and absolute discretion, to adjust or modify the calculation of a Performance Goal for such Performance Period to the extent permitted under Section 162(m) of the Code in order to prevent the dilution or enlargement of the rights of Participants based on the following events: (a) asset write-downs; (b) litigation or claim judgments or settlements; (c) the effect of changes in tax laws, accounting principles, or other laws or regulatory rules affecting reported results; (d) any reorganization and restructuring programs; (e) extraordinary nonrecurring items as described in Accounting Principles Board Opinion No. 30 (or any successor or pronouncement thereto) and/or in management's discussion and analysis of financial condition and results of operations appearing in the Company's annual report to stockholders for the applicable year;

(f) acquisitions or divestitures; (g) any other specific unusual or nonrecurring events, or objectively determinable category thereof; (h) foreign exchange gains and losses; and (i) a change in the Company's fiscal year. "Performance Period" means the one or more periods of time not less than one fiscal quarter in duration, as the Committee may select, over which the attainment of one or more Performance Goals will be measured for the purpose of determining a Participant's right to and the payment of a Performance Compensation Award. "Performance Share" means the grant of a right to receive a number of actual shares of Common Stock or share units based upon the performance of the Company during a Performance Period, as determined by the Committee. "Permitted Transferee" means: (a) a member of the Optionholder's immediate family (child, stepchild, grandchild, parent, stepparent, grandparent, spouse, former spouse, sibling, niece, nephew, mother-in-law, father-in-law, son-in-law, daughter-in-law, brother-in-law, or sister-in-law, including adoptive relationships), any person sharing the Optionholder's household (other than a tenant or employee), a trust in which these persons have more than 50% of the beneficial interest, a foundation in which these persons (or the Optionholder) control the management of assets, and any other entity in which these persons (or the Optionholder) own more than 50% of the voting interests; (b) third parties designated by the Committee in connection with a program established and approved by the Committee pursuant to which Participants may receive a cash payment or other consideration in consideration for the transfer of a Non-qualified Stock Option; and (c) such other transferees as may be permitted by the Committee in its sole discretion.

"Plan" means this Victory Energy Corporation 2017 Equity Incentive Plan, as amended and/or amended and restated from time to time.

"Restricted Award" means any Award granted pursuant to Section 7.2(a).

"Rule 16b-3" means Rule 16b-3 promulgated under the Exchange Act or any successor to Rule 16b-3, as in effect from time to time.

"Securities Act" means the Securities Act of 1933, as amended.

"Stock Appreciation Right" means the right pursuant to an Award granted under Section 7.1 to receive, upon exercise, an amount payable in cash or shares equal to the number of shares subject to the Stock Appreciation Right that is being exercised multiplied by the excess of (a) the Fair Market Value of a share of Common Stock on the date the Award is exercised, over (b) the exercise price specified in the Stock Appreciation Right Award Agreement. "Ten Percent Stockholder" means a person who owns (or is deemed to own pursuant to Section 424(d) of the Code) stock possessing more than 10% of the total combined voting power of all classes of stock of the Company or of any of its Affiliates.

- 3. Administration.
- 3.1. Authority of Committee. The Plan shall be administered by the Committee or, in the Board's sole discretion, by the Board. Subject to the terms of the Plan, the Committee's charter and Applicable Laws, and in addition to other express powers and authorization conferred by the Plan, the Committee shall have the authority:
- (a) to construe and interpret the Plan and apply its provisions;
- (b) to promulgate, amend, and rescind rules and regulations relating to the administration of the Plan;
- (c) to authorize any person to execute, on behalf of the Company, any instrument required to carry out the purposes of the Plan;
- (d) to delegate its authority to one or more Officers of the Company with respect to Awards that do not involve Covered Employees or "insiders" within the meaning of Section 16 of the Exchange Act;
- (e) to determine when Awards are to be granted under the Plan and the applicable Grant Date;
- (f) from time to time to select, subject to the limitations set forth in this Plan, those Participants to whom Awards shall be granted;
- (g) to determine the number of shares of Common Stock to be made subject to each Award;
- (h) to determine whether each Option is to be an Incentive Stock Option or a Non-qualified Stock Option;
- (i) to prescribe the terms and conditions of each Award, including, without limitation, the exercise price and medium of payment and vesting provisions, and to specify the provisions of the Award Agreement relating to such grant;
- (j) to determine the target number of Performance Shares to be granted pursuant to a Performance Share Award, the performance measures that will be used to establish the performance goals, the performance period(s) and the number of Performance Shares earned by a Participant;

- (k) to designate an Award (including a cash bonus) as a Performance Compensation Award and to select the Performance Criteria that will be used to establish the Performance Goals;
- (l) to amend any outstanding Awards, including for the purpose of modifying the time or manner of vesting, or the term of any outstanding Award; provided, however, that if any such amendment impairs a Participant's rights or increases a Participant's obligations under his or her Award or creates or increases a Participant's federal income tax liability with respect to an Award, such amendment shall also be subject to the Participant's consent;
- (m) to determine the duration and purpose of leaves of absences which may be granted to a Participant without constituting termination of their employment for purposes of the Plan, which periods shall be no shorter than the periods generally applicable to Employees under the Company's employment policies;
- (n) to make decisions with respect to outstanding Awards that may become necessary upon a change in corporate control or an event that triggers anti-dilution adjustments;
- (o) to interpret, administer, reconcile any inconsistency in, correct any defect in and/or supply any omission in the Plan and any instrument or agreement relating to, or Award granted under, the Plan; and
- (p) to exercise discretion to make any and all other determinations which it determines to be necessary or advisable for the administration of the Plan.

The Committee also may modify the purchase price or the exercise price of any outstanding Award, provided that if the modification effects a repricing, stockholder approval shall be required before the repricing is effective.

- 3.2. Committee Decisions Final. All decisions made by the Committee pursuant to the provisions of the Plan shall be final and binding on the Company and the Participants, unless such decisions are determined by a court having jurisdiction to be arbitrary and capricious.
- 3.3. Delegation. The Committee, or if no Committee has been appointed, the Board, may delegate administration of the Plan to a committee or committees of one or more members of the Board, and the term "Committee" shall apply to any person or persons to whom such authority has been delegated. The Committee shall have the power to delegate to a subcommittee any of the administrative powers the Committee is authorized to exercise (and references in this Plan to the Board or the Committee shall thereafter be to the committee or subcommittee), subject, however, to such resolutions, not inconsistent with the provisions of the Plan, as may be adopted from time to time by the Board. The Board may abolish the Committee at any time and re-vest in the Board the administration of the Plan. The members of the Committee shall be appointed by and serve at the pleasure of the Board. From time to time, the Board may increase or decrease the size of the Committee, add additional members to, remove members (with or without cause) from, appoint new members in substitution therefor, and fill vacancies, however caused, in the Committee. The Committee shall act pursuant to a vote of the majority of its members or, in the case of a Committee comprised of only two members, the unanimous consent of its members, whether present or not, or by the written consent of the majority of its members and minutes shall be kept of all of its meetings and copies thereof shall be provided to the Board. Subject to the limitations prescribed by the Plan and the Board, the Committee may establish and follow such rules and regulations for the conduct of its business as it may determine to be advisable.
- 3.4. Committee Composition. Except as otherwise determined by the Board, the Committee shall consist solely of two or more Non-Employee Directors who are also Outside Directors. The Board shall have discretion to determine whether or not it intends to comply with the exemption requirements of Rule 16b-3 and/or Section 162(m) of the Code. However, if the Board intends to satisfy such exemption requirements, with respect to Awards to any Covered Employee and with respect to any insider subject to Section 16 of the Exchange Act, the Committee shall be a compensation committee of the Board that at all times consists solely of two or more Non-Employee Directors who are also Outside Directors. Within the scope of such authority, the Board or the Committee may (a) delegate to a committee of one or more members of the Board who are not Outside Directors the authority to grant Awards to eligible persons who are either (i) not then Covered Employees and are not expected to be Covered Employees at the time of recognition of income resulting from such Award or (ii) not persons with respect to whom the Company wishes to comply with Section 162(m) of the Code or (b) delegate to a committee of one or more members of the Board who are not Non-Employee Directors the authority to grant Awards to eligible persons who are not then subject to Section 16 of the Exchange Act. Nothing herein shall create an inference that an Award is not validly granted under the Plan in the event Awards are granted under the Plan by a compensation committee of the Board that does not at all times consist solely of two or more Non-Employee Directors who are also Outside Directors.

- 3.5. Indemnification. In addition to such other rights of indemnification as they may have as Directors or members of the Committee, and to the extent allowed by Applicable Laws, the Committee shall be indemnified by the Company against the reasonable expenses, including attorney's fees, actually incurred in connection with any action, suit or proceeding or in connection with any appeal therein, to which the Committee may be party by reason of any action taken or failure to act under or in connection with the Plan or any Award granted under the Plan, and against all amounts paid by the Committee in settlement thereof (provided, however, that the settlement has been approved by the Company, which approval shall not be unreasonably withheld) or paid by the Committee in satisfaction of a judgment in any such action, suit or proceeding, except in relation to matters as to which it shall be adjudged in such action, suit or proceeding that such Committee did not act in good faith and in a manner which such person reasonably believed to be in the best interests of the Company, or in the case of a criminal proceeding, had no reason to believe that the conduct complained of was unlawful; provided, however, that within 60 days after institution of any such action, suit or proceeding, such Committee shall, in writing, offer the Company the opportunity at its own expense to handle and defend such action, suit or proceeding.
- 4. Shares Subject to the Plan.
- 4.1. Subject to adjustment in accordance with Section 11, a total of 570,000,000 shares of Common Stock shall be available for the grant of Awards under the Plan. Any shares of Common Stock granted in connection with Options and Stock Appreciation Rights shall be counted against this limit as one (1) share for every one (1) Option or Stock Appreciation Right awarded. Any shares of Common Stock granted in connection with Awards other than Options and Stock Appreciation Rights shall be counted against this limit as two (2) shares of Common Stock for every one (1) share of Common Stock granted in connection with such Award. During the terms of the Awards, the Company shall keep available at all times the number of shares of Common Stock required to satisfy such Awards.
- 4.2. Shares of Common Stock available for distribution under the Plan may consist, in whole or in part, of authorized and unissued shares, treasury shares or shares reacquired by the Company in any manner.
- 4.3. Subject to adjustment in accordance with Section 11, no Participant shall be granted, during any one (1) year period, Options to purchase Common Stock and Stock Appreciation Rights with respect to more than 9,500,000 shares of Common Stock in the aggregate or any other Awards with respect to more than 9,500,000 shares of Common Stock in the aggregate. If an Award is to be settled in cash, the number of shares of Common Stock on which the Award is based shall count toward the individual share limit set forth in this Section 4.
- 4.4. Any shares of Common Stock subject to an Award that is canceled, forfeited or expires prior to exercise or realization, either in full or in part, shall again become available for issuance under the Plan. Any shares of Common Stock that again become available for future grants pursuant to this Section 4.4 shall be added back as one (1) share if such shares were subject to Options or Stock Appreciation Rights and as two (2) shares if such shares were subject to other Awards. Notwithstanding anything to the contrary contained herein: shares subject to an Award under the Plan shall not again be made available for issuance or delivery under the Plan if such shares are (a) shares tendered in payment of an Option, (b) shares delivered or withheld by the Company to satisfy any tax withholding obligation, or (c) shares covered by a stock-settled Stock Appreciation Right or other Awards that were not issued upon the settlement of the Award.
- 5. Eligibility.
- 5.1. Eligibility for Specific Awards. Incentive Stock Options may be granted only to Employees. Awards other than Incentive Stock Options may be granted to Employees, Consultants and Directors and those individuals whom the Committee determines are reasonably expected to become Employees, Consultants and Directors following the Grant Date.
- 5.2. Ten Percent Stockholders. A Ten Percent Stockholder shall not be granted an Incentive Stock Option unless the Option Exercise Price is at least 110% of the Fair Market Value of the Common Stock at the Grant Date and the Option is not exercisable after the expiration of five years from the Grant Date.
- 6. Option Provisions. Each Option granted under the Plan shall be evidenced by an Award Agreement. Each Option so granted shall be subject to the conditions set forth in this Section 6, and to such other conditions not inconsistent with the Plan as may be reflected in the applicable Award Agreement. All Options shall be separately designated Incentive Stock Options or Non-qualified Stock Options at the time of grant, and, if certificates are issued, a separate certificate or certificates will be issued for shares of Common Stock purchased on exercise of each type of Option. Notwithstanding the foregoing, the Company shall have no liability to any Participant or any other person if an Option

designated as an Incentive Stock Option fails to qualify as such at any time or if an Option is determined to constitute "nonqualified deferred compensation" within the meaning of Section 409A of the Code and the terms of such Option do not satisfy the requirements of Section 409A of the Code. The provisions of separate Options need not be identical, but each Option shall include (through incorporation of provisions hereof by reference in the Option or otherwise) the substance of each of the following provisions:

- 6.1. Term. Subject to the provisions of Section 5.2 regarding Ten Percent Stockholders, no Incentive Stock Option shall be exercisable after the expiration of 10 years from the Grant Date. The term of a Non-qualified Stock Option granted under the Plan shall be determined by the Committee; provided, however, no Non-qualified Stock Option shall be exercisable after the expiration of 10 years from the Grant Date.
- 6.2. Exercise Price of An Incentive Stock Option. Subject to the provisions of Section 5.2 regarding Ten Percent Stockholders, the Option Exercise Price of each Incentive Stock Option shall be not less than 100% of the Fair Market Value of the Common Stock subject to the Option on the Grant Date. Notwithstanding the foregoing, an Incentive Stock Option may be granted with an Option Exercise Price lower than that set forth in the preceding sentence if such Option is granted pursuant to an assumption or substitution for another option in a manner satisfying the provisions of Section 424(a) of the Code.
- 6.3. Exercise Price of a Non-qualified Stock Option. The Option Exercise Price of each Non-qualified Stock Option shall be not less than 100% of the Fair Market Value of the Common Stock subject to the Option on the Grant Date. Notwithstanding the foregoing, a Non-qualified Stock Option may be granted with an Option Exercise Price lower than that set forth in the preceding sentence if such Option is granted pursuant to an assumption or substitution for another option in a manner satisfying the provisions of Section 409A of the Code.
- 6.4. Consideration. The Option Exercise Price of Common Stock acquired pursuant to an Option shall be paid, to the extent permitted by applicable statutes and regulations, either (a) in cash or by certified or bank check at the time the Option is exercised or (b) in the discretion of the Committee, upon such terms as the Committee shall approve, the Option Exercise Price may be paid: (i) by delivery to the Company of other Common Stock, duly endorsed for transfer to the Company, with a Fair Market Value on the date of delivery equal to the Option Exercise Price (or portion thereof) due for the number of shares being acquired, or by means of attestation whereby the Participant identifies for delivery specific shares of Common Stock that have an aggregate Fair Market Value on the date of attestation equal to the Option Exercise Price (or portion thereof) and receives a number of shares of Common Stock equal to the difference between the number of shares thereby purchased and the number of identified attestation shares of Common Stock (a "Stock for Stock Exchange"); (ii) a "cashless" exercise program established with a broker; (iii) by reduction in the number of shares of Common Stock otherwise deliverable upon exercise of such Option with a Fair Market Value equal to the aggregate Option Exercise Price at the time of exercise; (iv) any combination of the foregoing methods; or (v) in any other form of legal consideration that may be acceptable to the Committee. Unless otherwise specifically provided in the Option, the exercise price of Common Stock acquired pursuant to an Option that is paid by delivery (or attestation) to the Company of other Common Stock acquired, directly or indirectly from the Company, shall be paid only by shares of the Common Stock of the Company that have been held for more than six months (or such longer or shorter period of time required to avoid a charge to earnings for financial accounting purposes). Notwithstanding the foregoing, during any period for which the Common Stock is publicly traded (i.e., the Common Stock is listed on any established stock exchange or a national market system) an exercise by a Director or Officer that involves or may involve a direct or indirect extension of credit or arrangement of an extension of credit by the Company, directly or indirectly, in violation of Section 402(a) of the Sarbanes-Oxley Act of 2002 shall be prohibited with respect to any Award under this Plan.
- 6.5. Transferability of An Incentive Stock Option. An Incentive Stock Option shall not be transferable except by will or by the laws of descent and distribution and shall be exercisable during the lifetime of the Optionholder only by the Optionholder. Notwithstanding the foregoing, the Optionholder may, by delivering written notice to the Company, in a form satisfactory to the Company, designate a third party who, in the event of the death of the Optionholder, shall thereafter be entitled to exercise the Option.
- 6.6. Transferability of a Non-qualified Stock Option. A Non-qualified Stock Option may, in the sole discretion of the Committee, be transferable to a Permitted Transferee, upon written approval by the Committee to the extent provided in the Award Agreement. If the Non-qualified Stock Option does not provide for transferability, then the Non-qualified Stock Option shall not be transferable except by will or by the laws of descent and distribution and shall

be exercisable during the lifetime of the Optionholder only by the Optionholder. Notwithstanding the foregoing, the Optionholder may, by delivering written notice to the Company, in a form satisfactory to the Company, designate a third party who, in the event of the death of the Optionholder, shall thereafter be entitled to exercise the Option.

- 6.7. Vesting of Options. Each Option may, but need not, vest and therefore become exercisable in periodic installments that may, but need not, be equal. The Option may be subject to such other terms and conditions on the time or times when it may be exercised (which may be based on performance or other criteria) as the Committee may deem appropriate. The vesting provisions of individual Options may vary. No Option may be exercised for a fraction of a share of Common Stock. The Committee may, but shall not be required to, provide for an acceleration of vesting and exercisability in the terms of any Award Agreement upon the occurrence of a specified event.
- 6.8. Termination of Continuous Service. Unless otherwise provided in an Award Agreement or in an employment agreement the terms of which have been approved by the Committee, in the event an Optionholder's Continuous Service terminates (other than upon the Optionholder's death or Disability), the Optionholder may exercise his or her Option (to the extent that the Optionholder was entitled to exercise such Option as of the date of termination) but only within such period of time ending on the earlier of (a) the date three months following the termination of the Optionholder's Continuous Service or (b) the expiration of the term of the Option as set forth in the Award Agreement; provided that, if the termination of Continuous Service is by the Company for Cause, all outstanding Options (whether or not vested) shall immediately terminate and cease to be exercisable. If, after termination, the Optionholder does not exercise his or her Option within the time specified in the Award Agreement, the Option shall terminate.
- 6.9. Extension of Termination Date. An Optionholder's Award Agreement may also provide that if the exercise of the Option following the termination of the Optionholder's Continuous Service for any reason would be prohibited at any time because the issuance of shares of Common Stock would violate the registration requirements under the Securities Act or any other state or federal securities law or the rules of any securities exchange or interdealer quotation system, then the Option shall terminate on the earlier of (a) the expiration of the term of the Option in accordance with Section 6.1 or (b) the expiration of a period after termination of the Participant's Continuous Service that is three months after the end of the period during which the exercise of the Option would be in violation of such registration or other securities law requirements.
- 6.10. Disability of Optionholder. Unless otherwise provided in an Award Agreement, in the event that an Optionholder's Continuous Service terminates as a result of the Optionholder's Disability, the Optionholder may exercise his or her Option (to the extent that the Optionholder was entitled to exercise such Option as of the date of termination), but only within such period of time ending on the earlier of (a) the date 12 months following such termination or (b) the expiration of the term of the Option as set forth in the Award Agreement. If, after termination, the Optionholder does not exercise his or her Option within the time specified herein or in the Award Agreement, the Option shall terminate.
- 6.11. Death of Optionholder. Unless otherwise provided in an Award Agreement, in the event an Optionholder's Continuous Service terminates as a result of the Optionholder's death, then the Option may be exercised (to the extent the Optionholder was entitled to exercise such Option as of the date of death) by the Optionholder's estate, by a person who acquired the right to exercise the Option by bequest or inheritance or by a person designated to exercise the Option upon the Optionholder's death, but only within the period ending on the earlier of (a) the date 12 months following the date of death or (b) the expiration of the term of such Option as set forth in the Award Agreement. If, after the Optionholder's death, the Option is not exercised within the time specified herein or in the Award Agreement, the Option shall terminate.
- 6.12. Incentive Stock Option \$100,000 Limitation. To the extent that the aggregate Fair Market Value (determined at the time of grant) of Common Stock with respect to which Incentive Stock Options are exercisable for the first time by any Optionholder during any calendar year (under all plans of the Company and its Affiliates) exceeds \$100,000, the Options or portions thereof which exceed such limit (according to the order in which they were granted) shall be treated as Non-qualified Stock Options.
- 7. Provisions of Awards Other Than Options.
- 7.1. Stock Appreciation Rights.
- (a) General. Each Stock Appreciation Right granted under the Plan shall be evidenced by an Award Agreement. Each Stock Appreciation Right so granted shall be subject to the conditions set forth in this Section 7.1, and to such other conditions not inconsistent with the Plan as may be reflected in the applicable Award Agreement. Stock

Appreciation Rights may be granted alone ("Free Standing Rights") or in tandem with an Option granted under the Plan ("Related Rights").

- (b) Grant Requirements. Any Related Right that relates to a Non-qualified Stock Option may be granted at the same time the Option is granted or at any time thereafter but before the exercise or expiration of the Option. Any Related Right that relates to an Incentive Stock Option must be granted at the same time the Incentive Stock Option is granted.
- (c) Term of Stock Appreciation Rights. The term of a Stock Appreciation Right granted under the Plan shall be determined by the Committee; provided, however, no Stock Appreciation Right shall be exercisable later than the tenth anniversary of the Grant Date.
- (d) Vesting of Stock Appreciation Rights. Each Stock Appreciation Right may, but need not, vest and therefore become exercisable in periodic installments that may, but need not, be equal. The Stock Appreciation Right may be subject to such other terms and conditions on the time or times when it may be exercised as the Committee may deem appropriate. The vesting provisions of individual Stock Appreciation Rights may vary. No Stock Appreciation Right may be exercised for a fraction of a share of Common Stock. The Committee may, but shall not be required to, provide for an acceleration of vesting and exercisability in the terms of any Stock Appreciation Right upon the occurrence of a specified event.
- (e) Exercise and Payment. Upon exercise of a Stock Appreciation Right, the holder shall be entitled to receive from the Company an amount equal to the number of shares of Common Stock subject to the Stock Appreciation Right that is being exercised multiplied by the excess of (i) the Fair Market Value of a share of Common Stock on the date the Award is exercised, over (ii) the exercise price specified in the Stock Appreciation Right or related Option. Payment with respect to the exercise of a Stock Appreciation Right shall be made on the date of exercise. Payment shall be made in the form of shares of Common Stock (with or without restrictions as to substantial risk of forfeiture and transferability, as determined by the Committee in its sole discretion), cash or a combination thereof, as determined by the Committee.
- (f) Exercise Price. The exercise price of a Free Standing Stock Appreciation Right shall be determined by the Committee, but shall not be less than 100% of the Fair Market Value of one share of Common Stock on the Grant Date of such Stock Appreciation Right. A Related Right granted simultaneously with or subsequent to the grant of an Option and in conjunction therewith or in the alternative thereto shall have the same exercise price as the related Option, shall be transferable only upon the same terms and conditions as the related Option, and shall be exercisable only to the same extent as the related Option; provided, however, that a Stock Appreciation Right, by its terms, shall be exercisable only when the Fair Market Value per share of Common Stock subject to the Stock Appreciation Right and related Option exceeds the exercise price per share thereof and no Stock Appreciation Rights may be granted in tandem with an Option unless the Committee determines that the requirements of Section 7.1(b) are satisfied.
- (g) Reduction in the Underlying Option Shares. Upon any exercise of a Related Right, the number of shares of Common Stock for which any related Option shall be exercisable shall be reduced by the number of shares for which the Stock Appreciation Right has been exercised. The number of shares of Common Stock for which a Related Right shall be exercisable shall be reduced upon any exercise of any related Option by the number of shares of Common Stock for which such Option has been exercised.
- 7.2. Restricted Awards.
- (a) General. A Restricted Award is an Award of actual shares of Common Stock ("Restricted Stock") or hypothetical Common Stock units ("Restricted Stock Units") having a value equal to the Fair Market Value of an identical number of shares of Common Stock, which may, but need not, provide that such Restricted Award may not be sold, assigned, transferred or otherwise disposed of, pledged or hypothecated as collateral for a loan or as security for the performance of any obligation or for any other purpose for such period (the "Restricted Period") as the Committee shall determine. Each Restricted Award granted under the Plan shall be evidenced by an Award Agreement. Each Restricted Award so granted shall be subject to the conditions set forth in this Section 7.2, and to such other conditions not inconsistent with the Plan as may be reflected in the applicable Award Agreement.
- (b) Restricted Stock and Restricted Stock Units.
- (i) Each Participant granted Restricted Stock shall execute and deliver to the Company an Award Agreement with respect to the Restricted Stock setting forth the restrictions and other terms and conditions applicable to such Restricted Stock. If the Committee determines that the Restricted Stock shall be held by the Company or in escrow rather than delivered to the Participant pending the release of the applicable restrictions, the Committee may require

the Participant to additionally execute and deliver to the Company (A) an escrow agreement satisfactory to the Committee, if applicable and (B) the appropriate blank stock power with respect to the Restricted Stock covered by such agreement. If a Participant fails to execute an agreement evidencing an Award of Restricted Stock and, if applicable, an escrow agreement and stock power, the Award shall be null and void. Subject to the restrictions set forth in the Award, the Participant generally shall have the rights and privileges of a stockholder as to such Restricted Stock, including the right to vote such Restricted Stock and the right to receive dividends; provided that, any cash dividends and stock dividends with respect to the Restricted Stock shall be withheld by the Company for the Participant's account, and interest may be credited on the amount of the cash dividends withheld at a rate and subject to such terms as determined by the Committee. The cash dividends or stock dividends so withheld by the Committee and attributable to any particular share of Restricted Stock (and earnings thereon, if applicable) shall be distributed to the Participant in cash or, at the discretion of the Committee, in shares of Common Stock having a Fair Market Value equal to the amount of such dividends, if applicable, upon the release of restrictions on such share and, if such share is forfeited, the Participant shall have no right to such dividends.

- (ii) The terms and conditions of a grant of Restricted Stock Units shall be reflected in an Award Agreement. No shares of Common Stock shall be issued at the time a Restricted Stock Unit is granted, and the Company will not be required to set aside a fund for the payment of any such Award. A Participant shall have no voting rights with respect to any Restricted Stock Units granted hereunder. The Committee may also grant Restricted Stock Units with a deferral feature, whereby settlement is deferred beyond the vesting date until the occurrence of a future payment date or event set forth in an Award Agreement ("Deferred Stock Units"). At the discretion of the Committee, each Restricted Stock Unit or Deferred Stock Unit (representing one share of Common Stock) may be credited with cash and stock dividends paid by the Company in respect of one share of Common Stock ("Dividend Equivalents"). Dividend Equivalents shall be withheld by the Company and credited to the Participant's account, and interest may be credited on the amount of cash Dividend Equivalents credited to the Participant's account at a rate and subject to such terms as determined by the Committee. Dividend Equivalents credited to a Participant's account and attributable to any particular Restricted Stock Unit or Deferred Stock Unit (and earnings thereon, if applicable) shall be distributed in cash or, at the discretion of the Committee, in shares of Common Stock having a Fair Market Value equal to the amount of such Dividend Equivalents and earnings, if applicable, to the Participant upon settlement of such Restricted Stock Unit or Deferred Stock Unit and, if such Restricted Stock Unit or Deferred Stock Unit is forfeited, the Participant shall have no right to such Dividend Equivalents.
- (c) Restrictions.
- (i) Restricted Stock awarded to a Participant shall be subject to the following restrictions until the expiration of the Restricted Period, and to such other terms and conditions as may be set forth in the applicable Award Agreement: (A) if an escrow arrangement is used, the Participant shall not be entitled to delivery of the stock certificate; (B) the shares shall be subject to the restrictions on transferability set forth in the Award Agreement; (C) the shares shall be subject to forfeiture to the extent provided in the applicable Award Agreement; and (D) to the extent such shares are forfeited, the stock certificates shall be returned to the Company, and all rights of the Participant to such shares and as a stockholder with respect to such shares shall terminate without further obligation on the part of the Company.
- (ii) Restricted Stock Units and Deferred Stock Units awarded to any Participant shall be subject to (A) forfeiture until the expiration of the Restricted Period, and satisfaction of any applicable Performance Goals during such period, to the extent provided in the applicable Award Agreement, and to the extent such Restricted Stock Units or Deferred Stock Units are forfeited, all rights of the Participant to such Restricted Stock Units or Deferred Stock Units shall terminate without further obligation on the part of the Company and (B) such other terms and conditions as may be set forth in the applicable Award Agreement.
- (iii) The Committee shall have the authority to remove any or all of the restrictions on the Restricted Stock, Restricted Stock Units and Deferred Stock Units whenever it may determine that, by reason of changes in Applicable Laws or other changes in circumstances arising after the date the Restricted Stock or Restricted Stock Units or Deferred Stock Units are granted, such action is appropriate.
- (d) Restricted Period. With respect to Restricted Awards, the Restricted Period shall commence on the Grant Date and end at the time or times set forth on a schedule established by the Committee in the applicable Award Agreement. No Restricted Award may be granted or settled for a fraction of a share of Common Stock. The Committee may, but shall not be required to, provide for an acceleration of vesting in the terms of any Award Agreement upon the

occurrence of a specified event.

- (e) Delivery of Restricted Stock and Settlement of Restricted Stock Units. Upon the expiration of the Restricted Period with respect to any shares of Restricted Stock, the restrictions set forth in Section 7.2(c) and the applicable Award Agreement shall be of no further force or effect with respect to such shares, except as set forth in the applicable Award Agreement. If an escrow arrangement is used, upon such expiration, the Company shall deliver to the Participant, or his or her beneficiary, without charge, the stock certificate evidencing the shares of Restricted Stock which have not then been forfeited and with respect to which the Restricted Period has expired (to the nearest full share) and any cash dividends or stock dividends credited to the Participant's account with respect to such Restricted Stock and the interest thereon, if any. Upon the expiration of the Restricted Period with respect to any outstanding Restricted Stock Units, or at the expiration of the deferral period with respect to any outstanding Deferred Stock Units, the Company shall deliver to the Participant, or his or her beneficiary, without charge, one share of Common Stock for each such outstanding vested Restricted Stock Unit or Deferred Stock Unit ("Vested Unit") and cash equal to any Dividend Equivalents credited with respect to each such Vested Unit in accordance with Section 7.2(b)(ii) hereof and the interest thereon or, at the discretion of the Committee, in shares of Common Stock having a Fair Market Value equal to such Dividend Equivalents and the interest thereon, if any; provided, however, that, if explicitly provided in the applicable Award Agreement, the Committee may, in its sole discretion, elect to pay cash or part cash and part Common Stock in lieu of delivering only shares of Common Stock for Vested Units. If a cash payment is made in lieu of delivering shares of Common Stock, the amount of such payment shall be equal to the Fair Market Value of the Common Stock as of the date on which the Restricted Period lapsed in the case of Restricted Stock Units, or the delivery date in the case of Deferred Stock Units, with respect to each Vested Unit.
- (f) Stock Restrictions. Each certificate representing Restricted Stock awarded under the Plan shall bear a legend in such form as the Company deems appropriate.
- 7.3. Performance Share Awards.
- (a) Grant of Performance Share Awards. Each Performance Share Award granted under the Plan shall be evidenced by an Award Agreement. Each Performance Share Award so granted shall be subject to the conditions set forth in this Section 7.3, and to such other conditions not inconsistent with the Plan as may be reflected in the applicable Award Agreement. The Committee shall have the discretion to determine: (i) the number of shares of Common Stock or stock-denominated units subject to a Performance Share Award granted to any Participant; (ii) the performance period applicable to any Award; (iii) the conditions that must be satisfied for a Participant to earn an Award; and (iv) the other terms, conditions and restrictions of the Award.
- (b) Earning Performance Share Awards. The number of Performance Shares earned by a Participant will depend on the extent to which the performance goals established by the Committee are attained within the applicable Performance Period, as determined by the Committee. No payout shall be made with respect to any Performance Share Award except upon written certification by the Committee that the minimum threshold performance goal(s) have been achieved.
- 7.4. Performance Compensation Awards.
- (a) General. The Committee shall have the authority, at the time of grant of any Award described in this Plan (other than Options and Stock Appreciation Rights granted with an exercise price equal to or greater than the Fair Market Value per share of Common Stock on the Grant Date), to designate such Award as a Performance Compensation Award in order to qualify such Award as "performance-based compensation" under Section 162(m) of the Code. In addition, the Committee shall have the authority to make an Award of a cash bonus to any Participant and designate such Award as a Performance Compensation Award in order to qualify such Award as "performance-based compensation" under Section 162(m) of the Code.
- (b) Eligibility. The Committee will, in its sole discretion, designate within the first 90 days of a Performance Period (or, if longer or shorter, within the maximum period allowed under Section 162(m) of the Code) which Participants will be eligible to receive Performance Compensation Awards in respect of such Performance Period. However, designation of a Participant eligible to receive an Award hereunder for a Performance Period shall not in any manner entitle the Participant to receive payment in respect of any Performance Compensation Award for such Performance Period. The determination as to whether or not such Participant becomes entitled to payment in respect of any Performance Compensation Award shall be decided solely in accordance with the provisions of this Section 7.4. Moreover, designation of a Participant eligible to receive an Award hereunder for a particular Performance Period

shall not require designation of such Participant eligible to receive an Award hereunder in any subsequent Performance Period and designation of one person as a Participant eligible to receive an Award hereunder shall not require designation of any other person as a Participant eligible to receive an Award hereunder in such period or in any other period.

- (c) Discretion of Committee with Respect to Performance Compensation Awards. With regard to a particular Performance Period, the Committee shall have full discretion to select the length of such Performance Period (provided any such Performance Period shall be not less than one fiscal quarter in duration), the type(s) of Performance Compensation Awards to be issued, the Performance Criteria that will be used to establish the Performance Goal(s), the kind(s) and/or level(s) of the Performance Goal(s) that is (are) to apply to the Company and the Performance Formula. Within the first 90 days of a Performance Period (or, if longer or shorter, within the maximum period allowed under Section 162(m) of the Code), the Committee shall, with regard to the Performance Compensation Awards to be issued for such Performance Period, exercise its discretion with respect to each of the matters enumerated in the immediately preceding sentence of this Section 7.4(c) and record the same in writing.
- (d) Payment of Performance Compensation Awards.
- (i) Condition to Receipt of Payment. Unless otherwise provided in the applicable Award Agreement, a Participant must be employed by the Company on the last day of a Performance Period to be eligible for payment in respect of a Performance Compensation Award for such Performance Period.
- (ii) Limitation. A Participant shall be eligible to receive payment in respect of a Performance Compensation Award only to the extent that: (A) the Performance Goals for such period are achieved; and (B) the Performance Formula as applied against such Performance Goals determines that all or some portion of such Participant's Performance Compensation Award has been earned for the Performance Period.
- (iii) Certification. Following the completion of a Performance Period, the Committee shall review and certify in writing whether, and to what extent, the Performance Goals for the Performance Period have been achieved and, if so, calculate and certify in writing the amount of the Performance Compensation Awards earned for the period based upon the Performance Formula. The Committee shall then determine the actual size of each Participant's Performance Compensation Award for the Performance Period and, in so doing, may apply Negative Discretion in accordance with Section 7.4(d)(iv) hereof, if and when it deems appropriate.
- (iv) Use of Discretion. In determining the actual size of an individual Performance Compensation Award for a Performance Period, the Committee may reduce or eliminate the amount of the Performance Compensation Award earned under the Performance Formula in the Performance Period through the use of Negative Discretion if, in its sole judgment, such reduction or elimination is appropriate. The Committee shall not have the discretion to (A) grant or provide payment in respect of Performance Compensation Awards for a Performance Period if the Performance Goals for such Performance Period have not been attained or (B) increase a Performance Compensation Award above the maximum amount payable under Section 7.4(d)(vi) of the Plan.
- (v) Timing of Award Payments. Performance Compensation Awards granted for a Performance Period shall be paid to Participants as soon as administratively practicable following completion of the certifications required by this Section 7.4 but in no event later than 2 1/2 months following the end of the fiscal year during which the Performance Period is completed.
- (vi) Maximum Award Payable. Notwithstanding any provision contained in this Plan to the contrary, the maximum Performance Compensation Award payable to any one Participant under the Plan for a Performance Period (excluding any Options and Stock Appreciation Rights) is 9,500,000 shares of Common Stock or, in the event such Performance Compensation Award is paid in cash, the equivalent cash value thereof on the first or last day of the Performance Period to which such Award relates, as determined by the Committee. The maximum amount that can be paid in any calendar year to any Participant pursuant to a cash bonus Award described in the last sentence of Section 7.4(a) shall be \$250,000. Furthermore, any Performance Compensation Award that has been deferred shall not (between the date as of which the Award is deferred and the payment date) increase (A) with respect to a Performance Compensation Award that is payable in cash, by a measuring factor for each fiscal year greater than a reasonable rate of interest set by the Committee or (B) with respect to a Performance Compensation Award that is payable in shares of Common Stock, by an amount greater than the appreciation of a share of Common Stock from the date such Award is deferred to the payment date.

- 8. Securities Law Compliance. Each Award Agreement shall provide that no shares of Common Stock shall be purchased or sold thereunder unless and until (a) any then applicable requirements of state or federal laws and regulatory agencies have been fully complied with to the satisfaction of the Company and its counsel and (b) if required to do so by the Company, the Participant has executed and delivered to the Company a letter of investment intent in such form and containing such provisions as the Committee may require. The Company shall use reasonable efforts to seek to obtain from each regulatory commission or agency having jurisdiction over the Plan such authority as may be required to grant Awards and to issue and sell shares of Common Stock upon exercise of the Awards; provided, however, that this undertaking shall not require the Company to register under the Securities Act the Plan, any Award or any Common Stock issued or issuable pursuant to any such Award. If, after reasonable efforts, the Company is unable to obtain from any such regulatory commission or agency the authority which counsel for the Company deems necessary for the lawful issuance and sale of Common Stock under the Plan, the Company shall be relieved from any liability for failure to issue and sell Common Stock upon exercise of such Awards unless and until such authority is obtained.
- 9. Use of Proceeds from Stock. Proceeds from the sale of Common Stock pursuant to Awards, or upon exercise thereof, shall constitute general funds of the Company.
- 10. Miscellaneous.
- 10.1. Acceleration of Exercisability and Vesting. The Committee shall have the power to accelerate the time at which an Award may first be exercised or the time during which an Award or any part thereof will vest in accordance with the Plan, notwithstanding the provisions in the Award stating the time at which it may first be exercised or the time during which it will vest.
- 10.2. Stockholder Rights. Except as provided in the Plan or an Award Agreement, no Participant shall be deemed to be the holder of, or to have any of the rights of a holder with respect to, any shares of Common Stock subject to such Award unless and until such Participant has satisfied all requirements for exercise of the Award pursuant to its terms and no adjustment shall be made for dividends (ordinary or extraordinary, whether in cash, securities or other property) or distributions of other rights for which the record date is prior to the date such Common Stock certificate is issued, except as provided in Section 11 hereof.
- 10.3. No Employment or Other Service Rights. Nothing in the Plan or any instrument executed or Award granted pursuant thereto shall confer upon any Participant any right to continue to serve the Company or an Affiliate in the capacity in effect at the time the Award was granted or shall affect the right of the Company or an Affiliate to terminate (a) the employment of an Employee with or without notice and with or without Cause or (b) the service of a Director pursuant to the By-laws of the Company or an Affiliate, and any applicable provisions of the corporate law of the state in which the Company or the Affiliate is incorporated, as the case may be.
- 10.4. Transfer; Approved Leave of Absence. For purposes of the Plan, no termination of employment by an Employee shall be deemed to result from either (a) a transfer of employment to the Company from an Affiliate or from the Company to an Affiliate, or from one Affiliate to another, or (b) an approved leave of absence for military service or sickness, or for any other purpose approved by the Company, if the Employee's right to reemployment is guaranteed either by a statute or by contract or under the policy pursuant to which the leave of absence was granted or if the Committee otherwise so provides in writing, in either case, except to the extent inconsistent with Section 409A of the Code if the applicable Award is subject thereto.
- 10.5. Withholding Obligations. To the extent provided by the terms of an Award Agreement and subject to the discretion of the Committee, the Participant may satisfy any federal, state or local tax withholding obligation relating to the exercise or acquisition of Common Stock under an Award by any of the following means (in addition to the Company's right to withhold from any compensation paid to the Participant by the Company) or by a combination of such means: (a) tendering a cash payment; (b) authorizing the Company to withhold shares of Common Stock from the shares of Common Stock otherwise issuable to the Participant as a result of the exercise or acquisition of Common Stock under the Award, provided, however, that no shares of Common Stock are withheld with a value exceeding the minimum amount of tax required to be withheld by law; or (c) delivering to the Company previously owned and unencumbered shares of Common Stock of the Company.
- 11. Adjustments Upon Changes in Stock. In the event of changes in the outstanding Common Stock or in the capital structure of the Company by reason of any stock or extraordinary cash dividend, stock split, reverse stock split, an extraordinary corporate transaction such as any recapitalization, reorganization, merger, consolidation, combination,

exchange, or other relevant change in capitalization occurring after the Grant Date of any Award, Awards granted under the Plan and any Award Agreements, the exercise price of Options and Stock Appreciation Rights, the maximum number of shares of Common Stock subject to all Awards stated in Section 4 and the maximum number of shares of Common Stock with respect to which any one person may be granted Awards during any period stated in Sections 4 and 7.4(d)(vi) will be equitably adjusted or substituted, as to the number, price or kind of a share of Common Stock or other consideration subject to such Awards to the extent necessary to preserve the economic intent of such Award. In the case of adjustments made pursuant to this Section 11, unless the Committee specifically determines that such adjustment is in the best interests of the Company or its Affiliates, the Committee shall, in the case of Incentive Stock Options, ensure that any adjustments under this Section 11 will not constitute a modification, extension or renewal of the Incentive Stock Options within the meaning of Section 424(h)(3) of the Code and in the case of Non-qualified Stock Options, ensure that any adjustments under this Section 11 will not constitute a modification of such Non-qualified Stock Options within the meaning of Section 409A of the Code. Any adjustments made under this Section 11 shall be made in a manner which does not adversely affect the exemption provided pursuant to Rule 16b-3 under the Exchange Act. Further, with respect to Awards intended to qualify as "performance-based compensation" under Section 162(m) of the Code, any adjustments or substitutions will not cause the Company to be denied a tax deduction on account of Section 162(m) of the Code. The Company shall give each Participant notice of an adjustment hereunder and, upon notice, such adjustment shall be conclusive and binding for all purposes.

- 12. Effect of Change in Control.
- 12.1. In the discretion of the Board and the Committee, any Award Agreement may provide, or the Board or the Committee may provide by amendment of any Award Agreement or otherwise, notwithstanding any provision of the Plan to the contrary, that in the event of a Change in Control, Options and/or Stock Appreciation Rights shall become immediately exercisable with respect to all or a specified portion of the shares subject to such Options or Stock Appreciation Rights, and/or the Restricted Period shall expire immediately with respect to all or a specified portion of the shares of Restricted Stock or Restricted Stock Units.
- 12.2. In addition, in the event of a Change in Control, the Committee may in its discretion and upon at least 10 days' advance notice to the affected persons, cancel any outstanding Awards and pay to the holders thereof, in cash or stock, or any combination thereof, the value of such Awards based upon the price per share of Common Stock received or to be received by other stockholders of the Company in the event. In the case of any Option or Stock Appreciation Right with an exercise price that equals or exceeds the price paid for a share of Common Stock in connection with the Change in Control, the Committee may cancel the Option or Stock Appreciation Right without the payment of consideration therefor.
- 12.3. The obligations of the Company under the Plan shall be binding upon any successor corporation or organization resulting from the merger, consolidation or other reorganization of the Company, or upon any successor corporation or organization succeeding to all or substantially all of the assets and business of the Company and its Subsidiaries, taken as a whole.
- 13. Amendment of the Plan and Awards.
- 13.1. Amendment of Plan. The Board at any time, and from time to time, may amend or terminate the Plan. However, except as provided in Section 11 relating to adjustments upon changes in Common Stock and Section 13.3, no amendment shall be effective unless approved by the stockholders of the Company to the extent stockholder approval is necessary to satisfy any Applicable Laws. At the time of such amendment, the Board shall determine, upon advice from counsel, whether such amendment will be contingent on stockholder approval.
- 13.2. Stockholder Approval. The Board may, in its sole discretion, submit any other amendment to the Plan for stockholder approval, including, but not limited to, amendments to the Plan intended to satisfy the requirements of Section 162(m) of the Code and the regulations thereunder regarding the exclusion of performance-based compensation from the limit on corporate deductibility of compensation paid to certain executive officers.
- 13.3. Contemplated Amendments. It is expressly contemplated that the Board may amend the Plan in any respect the Board deems necessary or advisable to provide eligible Employees, Consultants and Directors with the maximum benefits provided or to be provided under the provisions of the Code and the regulations promulgated thereunder relating to Incentive Stock Options or to the nonqualified deferred compensation provisions of Section 409A of the Code and/or to bring the Plan and/or Awards granted under it into compliance therewith.

- 13.4. No Impairment of Rights. Rights under any Award granted before amendment of the Plan shall not be impaired by any amendment of the Plan unless (a) the Company requests the consent of the Participant and (b) the Participant consents in writing.
- 13.5. Amendment of Awards. The Committee at any time, and from time to time, may amend the terms of any one or more Awards; provided, however, that the Committee may not affect any amendment which would otherwise constitute an impairment of the rights under any Award unless (a) the Company requests the consent of the Participant and (b) the Participant consents in writing.
- 14. General Provisions.
- 14.1. Forfeiture Events. The Committee may specify in an Award Agreement that the Participant's rights, payments and benefits with respect to an Award shall be subject to reduction, cancellation, forfeiture or recoupment upon the occurrence of certain events, in addition to applicable vesting conditions of an Award. Such events may include, without limitation, breach of non-competition, non-solicitation, confidentiality, or other restrictive covenants that are contained in the Award Agreement or otherwise applicable to the Participant, a termination of the Participant's Continuous Service for Cause, or other conduct by the Participant that is detrimental to the business or reputation of the Company and/or its Affiliates.
- 14.2. Clawback. Notwithstanding any other provisions in this Plan, any Award which is subject to recovery under any law, government regulation or stock exchange listing requirement, will be subject to such deductions and clawback as may be required to be made pursuant to such law, government regulation or stock exchange listing requirement (or any policy adopted by the Company pursuant to any such law, government regulation or stock exchange listing requirement).
- 14.3. Other Compensation Arrangements. Nothing contained in this Plan shall prevent the Board from adopting other or additional compensation arrangements, subject to stockholder approval if such approval is required; and such arrangements may be either generally applicable or applicable only in specific cases.
- 14.4. Sub-plans. The Committee may from time to time establish sub-plans under the Plan for purposes of satisfying blue sky, securities, tax or other laws of various jurisdictions in which the Company intends to grant Awards. Any sub-plans shall contain such limitations and other terms and conditions as the Committee determines are necessary or desirable. All sub-plans shall be deemed a part of the Plan, but each sub-plan shall apply only to the Participants in the jurisdiction for which the sub-plan was designed.
- 14.5. Deferral of Awards. The Committee may establish one or more programs under the Plan to permit selected Participants the opportunity to elect to defer receipt of consideration upon exercise of an Award, satisfaction of performance criteria, or other event that absent the election would entitle the Participant to payment or receipt of shares of Common Stock or other consideration under an Award. The Committee may establish the election procedures, the timing of such elections, the mechanisms for payments of, and accrual of interest or other earnings, if any, on amounts, shares or other consideration so deferred, and such other terms, conditions, rules and procedures that the Committee deems advisable for the administration of any such deferral program.
- 14.6. Unfunded Plan. The Plan shall be unfunded. Neither the Company, the Board nor the Committee shall be required to establish any special or separate fund or to segregate any assets to assure the performance of its obligations under the Plan.
- 14.7. Recapitalizations. Each Award Agreement shall contain provisions required to reflect the provisions of Section 11.
- 14.8. Delivery. Upon exercise of a right granted under this Plan, the Company shall issue Common Stock or pay any amounts due within a reasonable period of time thereafter. Subject to any statutory or regulatory obligations the Company may otherwise have, for purposes of this Plan, thirty (30) days shall be considered a reasonable period of time.
- 14.9. No Fractional Shares. No fractional shares of Common Stock shall be issued or delivered pursuant to the Plan. The Committee shall determine whether cash, additional Awards or other securities or property shall be issued or paid in lieu of fractional shares of Common Stock or whether any fractional shares should be rounded, forfeited or otherwise eliminated.
- 14.10. Other Provisions. The Award Agreements authorized under the Plan may contain such other provisions not inconsistent with this Plan, including, without limitation, restrictions upon the exercise of the Awards, as the Committee may deem advisable.

- 14.11. Section 409A. The Plan is intended to comply with Section 409A of the Code to the extent subject thereto, and, accordingly, to the maximum extent permitted, the Plan shall be interpreted and administered to be in compliance therewith. Any payments described in the Plan that are due within the "short-term deferral period" as defined in Section 409A of the Code shall not be treated as deferred compensation unless Applicable Laws require otherwise. Notwithstanding anything to the contrary in the Plan, to the extent required to avoid accelerated taxation and tax penalties under Section 409A of the Code, amounts that would otherwise be payable and benefits that would otherwise be provided pursuant to the Plan during the six (6) month period immediately following the Participant's termination of Continuous Service shall instead be paid on the first payroll date after the six-month anniversary of the Participant's separation from service (or the Participant's death, if earlier). Notwithstanding the foregoing, neither the Company nor the Committee shall have any obligation to take any action to prevent the assessment of any excise tax or penalty on any Participant under Section 409A of the Code and neither the Company nor the Committee will have any liability to any Participant for such tax or penalty.
- 14.12. Disqualifying Dispositions. Any Participant who shall make a "disposition" (as defined in Section 424 of the Code) of all or any portion of shares of Common Stock acquired upon exercise of an Incentive Stock Option within two years from the Grant Date of such Incentive Stock Option or within one year after the issuance of the shares of Common Stock acquired upon exercise of such Incentive Stock Option (a "Disqualifying Disposition") shall be required to immediately advise the Company in writing as to the occurrence of the sale and the price realized upon the sale of such shares of Common Stock.
- 14.13. Section 16. It is the intent of the Company that the Plan satisfy, and be interpreted in a manner that satisfies, the applicable requirements of Rule 16b-3 as promulgated under Section 16 of the Exchange Act so that Participants will be entitled to the benefit of Rule 16b-3, or any other rule promulgated under Section 16 of the Exchange Act, and will not be subject to short-swing liability under Section 16 of the Exchange Act. Accordingly, if the operation of any provision of the Plan would conflict with the intent expressed in this Section 14.13, such provision to the extent possible shall be interpreted and/or deemed amended so as to avoid such conflict.
- 14.14. Section 162(m). To the extent the Committee issues any Award that is intended to be exempt from the deduction limitation of Section 162(m) of the Code, the Committee may, without stockholder or grantee approval, amend the Plan or the relevant Award Agreement retroactively or prospectively to the extent it determines necessary in order to comply with any subsequent clarification of Section 162(m) of the Code required to preserve the Company's federal income tax deduction for compensation paid pursuant to any such Award.
- 14.15. Beneficiary Designation. Each Participant under the Plan may from time to time name any beneficiary or beneficiaries by whom any right under the Plan is to be exercised in case of such Participant's death. Each designation will revoke all prior designations by the same Participant, shall be in a form reasonably prescribed by the Committee and shall be effective only when filed by the Participant in writing with the Company during the Participant's lifetime.
- 14.16. Expenses. The costs of administering the Plan shall be paid by the Company.
- 14.17. Severability. If any of the provisions of the Plan or any Award Agreement is held to be invalid, illegal or unenforceable, whether in whole or in part, such provision shall be deemed modified to the extent, but only to the extent, of such invalidity, illegality or unenforceability and the remaining provisions shall not be affected thereby.
- 14.18. Plan Headings. The headings in the Plan are for purposes of convenience only and are not intended to define or limit the construction of the provisions hereof.
- 14.19. Non-Uniform Treatment. The Committee's determinations under the Plan need not be uniform and may be made by it selectively among persons who are eligible to receive, or actually receive, Awards. Without limiting the generality of the foregoing, the Committee shall be entitled to make non-uniform and selective determinations, amendments and adjustments, and to enter into non-uniform and selective Award Agreements.
- 15. Effective Date of Plan. The Plan shall become effective as of the Effective Date, but no Award shall be exercised (or, in the case of a stock Award, shall be granted) unless and until the Plan has been approved by the stockholders of the Company, which approval shall be within twelve (12) months before or after the date the Plan is adopted by the Board.
- 16. Termination or Suspension of the Plan. The Plan shall terminate automatically on September 14, 2027. No Award shall be granted pursuant to the Plan after such date, but Awards theretofore granted may extend beyond that date. The Board may suspend or terminate the Plan at any earlier date pursuant to Section 13.1 hereof. No Awards may be granted under the Plan while the Plan is suspended or after it is terminated. Unless the Company determines to

submit Section 7.4 of the Plan and the definition of "Performance Goal" and "Performance Criteria" to the Company's stockholders at the first stockholder meeting that occurs in the fifth year following the year in which the Plan was last approved by stockholders (or any earlier meeting designated by the Board), in accordance with the requirements of Section 162(m) of the Code, and such stockholder approval is obtained, then no further Performance Compensation Awards shall be made to Covered Employees under Section 7.4 after the date of such annual meeting, but the Plan may continue in effect for Awards to Participants not in accordance with Section 162(m) of the Code.

17. Choice of Law. The law of the State of Nevada shall govern all questions concerning the construction, validity and interpretation of this Plan, without regard to such state's conflict of law rules.

As adopted by the Board of Directors of Victory Energy Corporation on September 14, 2017.

As approved by the stockholders of Victory Energy Corporation on ______, 2017.

ANNEX D

Unaudited Pro Forma Financial Statements

The following unaudited pro forma consolidated financial statements are based upon the historical consolidated financial statements of Victory Energy Corporation (the "Company"), adjusted to give effect to the divestiture (the "Divestiture") of all of the Company's partnership interests in Aurora Energy Partners, a Texas general partnership, pursuant to the terms and conditions of a divestiture agreement, dated August 21, 2017, between Navitus Energy Group and the Company, as amended. These unaudited pro forma consolidated financial statements are derived from, and should be read in conjunction with, the financial statements contained herein and the Company's Annual Report on Form 10-K for the year ended December 31, 2016 filed with the Securities and Exchange Commission on March 31, 2017 and the Quarterly Report on Form 10-Q for the interim period ended June 30, 2017 filed with the Securities and Exchange Commission on August 11, 2017.

The unaudited pro forma consolidated balance sheet gives effect to the proposed Divestiture as if it had occurred on June 30, 2017. The unaudited pro forma consolidated statements of operations for the six months ended June 30, 2017 and the year ended December 31, 2016 give effect to the proposed asset sale as if it had occurred on January 1, 2016. The pro forma adjustments related to the Divestiture are based on available information and assumptions that management believes are (1) directly attributable to the Divestiture; (2) factually supportable; and (3) with respect to the unaudited pro forma consolidated statements of operations, expected to have a continuing impact on consolidated operating results. Certain of the most significant assumptions are set forth under the Notes to Unaudited Pro Forma Consolidated Financial Statements.

The Company has included the following unaudited pro forma consolidated financial information solely for the purpose of providing stockholders with information that may be useful for purposes of considering and evaluating the proposal to approve the Divestiture. The unaudited pro forma consolidated financial information is not necessarily indicative of the results of operations or financial position that might have been achieved for the dates or periods indicated, nor is it indicative of the results of operations or financial position that may occur in the future.

VICTORY ENERGY CORPORATION UNAUDITED PRO FORMA CONSOLIDATED BALANCE SHEETS

	June 30, 2017	Adjustment	ts Pro Forma
ASSETS			
Current Assets			
Cash and cash equivalents	\$47,157	\$ —	\$47,157
Accounts receivable	28,972	(28,972)	a —
Prepaid expenses	26,715	_	26,715
Total current assets	102,844	(28,972)	73,872
Fixed Assets			

Firmitian and againment	43,622		43,622
Furniture and equipment Accumulated depreciation	(40,211)	_	(40,211)
Total furniture and fixtures, net		_	
•	3,411	— (2.705.557)	3,411
Oil and gas properties, net of impairment (successful efforts method)	2,795,557	(2,795,557)	
Accumulated depletion, depreciation and amortization	(2,214,709)	2,214,709	c —
Total oil and gas properties, net	580,848	(580,848)	
Other Assets	120 455	(11.010)	1 100 140
Management fee receivable - affiliate	139,455	(11,313)	d 128,142
Total Assets	\$826,558	\$(621,133)	\$205,425
LIABILITIES AND STOCKHOLDERS' DEFICIT			
Current Liabilities			
Accounts payable	\$570,667	\$—	\$570,667
Accrued liabilities	404,419	(55,631)	e 348,788
Accrued liabilities - related parties	1,273,616		1,273,616
Liability for unauthorized preferred stock issued	9,283		9,283
Note payable (net of unamortized deferred financing costs)	254,500	_	254,500
Note payable (net of debt discount) - affiliate	284,055	_	284,055
Asset retirement obligation	52,321	(52,321)	f —
Total current liabilities	2,848,861	(107,952)	2,740,909
Other Liabilities			
Asset retirement obligations	40,895	(40,895)	f —
Total long term liabilities	40,895	(40,895)	_
Total Liabilities	2,889,756	(148,847)	2,740,909
Stockholders' Equity (Deficit)			
Common stock, \$0.001 par value, 47,500,000 shares authorized,			
31,220,326 shares and 31,220,326 shares issued and outstanding for Jun	e 31,220		31,220
30, 2017 and December 31, 2016, respectively			
Additional paid-in capital	36,134,513		36,134,513
Accumulated deficit		(472,286)	g (47,865,585)
Total Victory Energy Corporation stockholders' deficit	(11,227,566)		(11,699,852)
Non-controlling interest	9,164,368	, (, ,)	9,164,368
Total stockholders' equity (deficit)	(2,063,198)	(472,286)	(2,535,484)
Total Liabilities and Stockholders' Deficit	\$826,558	\$(621,133)	\$205,425
	-	+ (0=1,100)	+ 200, .20

VICTORY ENERGY CORPORATION UNAUDITED PRO FORMA CONSOLIDATED STATEMENTS OF OPERATIONS For the Six Months Ended June 30, 2017

	Six Month Ended Jun 30, 2017	ts Pro Forma	
Revenues			
Oil and gas sales	\$155,980	\$(155,980)	h \$—
Gain on settlement and sale of oil and gas properties			
Total revenues	155,980	(155,980)	
Operating Expenses:			
Lease operating costs	57,740	(57,740)	i —
Exploration and dry hole cost	2,218	(2,218)	j —
Production taxes	8,491	(8,491)	k —

General and administrative	1,117,488	(46,169)	1	1,071,319
Depreciation, depletion, amortization, and accretion	62,300	(49,720)	m	12,580
Total operating expenses	1,248,237	(164,338)		1,083,899
Loss from operations	(1,092,257)	8,358		(1,083,899)
Other Income (Expense):				
Management fee income	1,899	(1,899)	n	_
Interest expense	(187,991)	_		(187,991)
Total other income and expense	(186,092)	(1,899)		(187,991)
Loss before Tax Benefit	(1,278,349)	6,459		(1,271,890)
Tax benefit		_		_
Net loss	(1,278,349)	6,459		(1,271,890)
Less: Net loss attributable to non-controlling interest	(25,798)	25,798	O	_
Net loss attributable to Victory Energy Corporation	\$(1,252,551)	\$(19,339)	9	\$(1,271,890)
Weighted average shares, basic and diluted	31,220,326			31,220,326
Net loss per share, basic and diluted	\$(0.040)	\$—	9	\$(0.040)

VICTORY ENERGY CORPORATION UNAUDITED PRO FORMA CONSOLIDATED STATEMENTS OF OPERATIONS For the Year Ended December 31, 2016

	Year Ended December 31, 2016	Adjustment	iS	Pro Forma
Revenues				
Oil and gas sales	\$287,179	\$(287,179)	h :	\$ —
Gain on settlement and sale of oil and gas properties	153,624	(153,624)	p	_
Total revenues	440,803	(440,803)		_
Operating Expenses:				
Lease operating costs	105,245	(105,245)	i	
Exploration and dry hole cost	3,000	(3,000)	j	
Production taxes	14,690	(14,690)	k	
General and administrative	1,961,314	(194,088)	1	1,767,226
Depreciation, depletion, amortization, and accretion	135,009	(128,545)	m	6,464
Total operating expenses	2,219,258	(445,568)		1,773,690
Loss from operations	(1,778,455)	4,765		(1,773,690)
Other Income (Expense):				
Management fee income	5,972	(5,972)	n	
Interest expense	(134,116)			(134,116)
Total other income and expense	(128,144)	(5,972)		(134,116)
Loss before Tax Benefit	(1,906,599)	(1,207)		(1,907,806)
Tax benefit				
Net loss	(1,906,599)	(1,207)		(1,907,806)
Less: Net loss attributable to non-controlling interest	(54,975	54,975	o	
Net loss attributable to Victory Energy Corporation	\$(1,851,624)	\$(56,182)	:	\$(1,907,806)
Weighted average shares, basic and diluted	31,220,326			31,220,326
Net loss per share, basic and diluted	\$(0.060)	\$:	\$(0.060)

VICTORY ENERGY CORPORATION NOTES TO UNAUDITED PRO FORMA CONSOLIDATED FINANCIAL STATEMENTS

1. Planned Divestiture

On August 21, 2017, Victory Energy Corporation, a Nevada corporation (the "Company"), entered into a divestiture agreement (the "Divestiture Agreement") with Navitus Energy Group ("Navitus"), and on September 14, 2017, the parties entered into Amendment No. 1 to the Divestiture Agreement. Pursuant to the Divestiture Agreement, as amended, the Company agreed to divest and transfer its 50% ownership interest in Aurora Energy Partners, a Texas general partnership ("Aurora"), to Navitus (the "Divestiture"), which currently owns the remaining 50% interest, in consideration for a release from Navitus of all of the Company's obligations under the second amended partnership agreement, dated October 1, 2011, between the Company and Navitus, including, without limitation, obligations to return to Navitus investors their accumulated deferred capital, deferred interest and related allocations of equity. The Company also agreed to (i) issue 166,549,134 shares of the Company's common stock (or 4,382,872 shares following the planned 1-for-38 reverse stock split) to Navitus and (ii) pay off or otherwise satisfy all indebtedness and other material liabilities of Aurora at or prior to closing of the Divestiture Agreement.

2. Unaudited Pro Forma Adjustments

The unaudited pro forma consolidated balance sheet gives effect to the proposed Divestiture as if it had occurred on June 30, 2017. The unaudited pro forma consolidated statements of operations for the six months ended June 30, 2017 and the year ended December 31, 2016 give effect to the proposed Divestiture as if it had occurred on January 1, 2016. The preparation of the unaudited pro forma consolidated financial information is based on financial statements prepared in accordance with accounting principles generally accepted in the United States. The pro forma adjustments reflected in the accompanying unaudited pro forma consolidated financial information are based on available information and assumptions that management believes are (1) directly attributable to the Divestiture; (2) factually supportable; and (3) with respect to the unaudited pro forma consolidated statements of operations, expected to have a continuing impact on consolidated operating results. Actual results may differ from those estimates.

The unaudited pro forma consolidated financial information is provided for illustrative purposes only and does not

purport to represent what the actual results of operations would have been had the Divestiture occurred on the respective dates assumed, nor is it necessarily indicative of future operating results. The unaudited pro forma consolidated financial information and the accompanying unaudited notes should be read in conjunction with our consolidated financial statements and notes thereto included in this proxy statement.

The following pro forma adjustments related to the proposed Divestiture are included in the unaudited pro forma consolidated balance sheets and/or the unaudited pro forma consolidated statements of operations.

- (a) Represents accounts receivable related to the oil and gas properties divested pursuant to the Divestiture Agreement.
- (b) Represents the oil and gas properties, net of impairment, divested pursuant to the Divestiture Agreement.
- (c) Represents accumulated depletion related to the oil and gas properties divested pursuant to the Divestiture Agreement.
- (d) Represents a related party receivable, due to Aurora, related to the oil and gas properties divested pursuant to the Divestiture Agreement.
- (e) Represents production taxes and royalties related to the oil and gas properties divested pursuant to the Divestiture Agreement.
- (f) Represents asset retirement obligation related to the oil and gas properties divested pursuant to the Divestiture Agreement.
- (g) The overall adjustment to accumulated deficit represents the net effect of the oil and gas properties and related receivables, liabilities, revenues and expenses divested pursuant to the Divestiture Agreement.
- (h) Represents oil and gas sales related to the oil and gas properties divested pursuant to the Divestiture Agreement.
- (i) Represents lease operating costs related to the oil and gas properties divested pursuant to the Divestiture Agreement.

- (j) Represents exploration and dry hole costs related to the oil and gas properties divested pursuant to the Divestiture Agreement.
- (k) Represents production taxes related to the oil and gas properties divested pursuant to the Divestiture Agreement.
- (l) Represents general and administrative expense related to the oil and gas properties divested pursuant to the Divestiture Agreement.
- (m) Represents depletion and accretion related to the oil and gas properties divested pursuant to the Divestiture Agreement.
- (n) Represents management fee income of the Company that would not have been earned if the Company's interested in Aurora was divested as of January 1, 2016.
- (o) Represents net losses attributable to non-controlling interest which would not exist if the Company's interested in Aurora was divested as of January 1, 2016.
- (p) Represents gain on settlement and sale of oil and gas properties related to the oil and gas properties divested pursuant to the Divestiture Agreement.

ANNEX E

Financial Statements

The Company's audited financial statements for the years ended December 31, 2016 and 2015 and its unaudited interim period financial statements for the periods ended June 30, 2017 and 2016 are included below.

VICTORY ENERGY CORPORATION AND SUBSIDIARIES AUDITED CONSOLIDATED FINANCIAL STATEMENTS YEARS ENDED DECMEBER 31, 2016 AND 2015

Report of Independent Registered Public Accounting Firm

To the Board of Directors and Stockholders of Victory Energy Corporation

We have audited the accompanying consolidated balance sheets of Victory Energy Corporation (the Company) as of December 31, 2016 and 2015, and the related consolidated statements of operations, stockholders' equity, and cash flows for each of the years in the two-year period ended December 31, 2016. These consolidated financial statements are the responsibility of the Company's management. Our responsibility is to express an opinion on these consolidated financial statements based on our audits.

We conducted our audits in accordance with the standards of the Public Company Accounting Oversight Board (United States). Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the consolidated financial statements are free of material misstatement. An audit includes examining, on a test basis, evidence supporting the amounts and disclosures in the consolidated financial statements. An audit also includes assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall consolidated financial statement presentation. We believe that our audits provide a reasonable basis for our opinion.

In our opinion, the consolidated financial statements referred to above present fairly, in all material respects, the consolidated financial position of Victory Energy Corporation as of December 31, 2016 and 2015, and the results of its operations and its cash flows for each of the years in the two-year period ended December 31, 2016 in conformity with accounting principles generally accepted in the United States of America.

The accompanying consolidated financial statements have been prepared assuming the Company will continue as a going concern. The Company has experienced recurring losses since its inception and has an accumulated deficit. These conditions raise substantial doubt regarding the Company's ability to continue as a going concern. Management's plans in regard to these matters are described in Note 1 to the consolidated financial statements. The consolidated financial statements do not include any adjustments to reflect the possible future effects on the recoverability and classification of assets or the amounts and classification of liabilities that may result from the outcome of this uncertainty.

WEAVER AND TIDWELL, L.L.P. March 31, 2017 Houston, Texas

VICTORY ENERGY CORPORATION AND SUBSIDIARIES CONSOLIDATED BALANCE SHEETS December 31, 2016 and 2015

ASSETS Current Assets 12/31/2016 12/31/2015

Cash and cash equivalents	\$56,456	\$2,384
Accounts receivable - less allowance for doubtful accounts of \$0 and \$200,000 at December		•
31, 2016 and 2015, respectively	44,379	37,690
Prepaid expenses	9,951	8,734
Total current assets	110,786	48,808
Fixed Assets	.,	-,
Furniture and equipment	46,883	46,883
Accumulated depreciation	•	(24,429)
Total furniture and fixtures, net	15,990	22,454
Oil and gas properties, net of impairment (successful efforts method)	2,787,986	3,033,279
Accumulated depletion, depreciation and amortization		(2,274,188)
Total oil and gas properties, net	621,343	759,091
Management fee receivable - affiliate	137,556	131,584
	4005.655	Φ061.02 7
Total Assets	\$885,675	\$961,937
LIABILITIES AND STOCKHOLDERS' DEFICIT		
Current Liabilities	ф. 43 0. 550	Φ1 501 7 64
Accounts payable	\$420,559	\$1,591,764
Accrued liabilities	746,491	534,619
Accrued liabilities - related parties	1,489,973	805,179
Liability for unauthorized preferred stock issued	9,283	9,283
Note payable (net of unamortized deferred financing costs)	564,263	632,940
Asset retirement obligation	76,850	14,403
Total current liabilities	3,307,419	3,588,188
Other Liabilities		
Asset retirement obligations	7,141	94,768
Total long term liabilities	7,141	94,768
Total liabilities	\$3,314,560	\$3,682,956
Stockholders' Equity (Deficit)		
Common stock, \$0.001 par value, 47,500,000 shares authorized, 31,220,326 shares and	\$31,220	\$31,220
31,220,326 shares issued and outstanding at December 31, 2016 and 2015, respectively		
Additional paid-in capital		35,708,746
Accumulated deficit		(44,289,126)
Total Victory Energy Corporation stockholders' deficit		(8,549,160)
Non-controlling interest	7,885,166	5,828,141
Total stockholders' equity (deficit)	(2,428,885)	(2,721,019)
Total Liabilities and Stockholders' Deficit	\$885,675	\$961,937

The accompanying notes are an integral part of these consolidated financial statements.

VICTORY ENERGY CORPORATION AND SUBSIDIARIES CONSOLIDATED STATEMENTS OF OPERATIONS

For the years ended December 31, 2016 and 2015

	12/31/2016	12/31/2015
Revenues		
Oil and gas sales	\$287,179	\$650,648
Gain on settlement and sale of oil and gas properties	153,624	637,248

1,287,896
159,800
2,513
32,704
14 4,389,788
867,048
637,121
58 6,088,974
155) (4,801,078)
8,028
6) (112,468)
4) (104,440)
599) (4,905,518)
_
599) (4,905,518)
) (728,218)
,624) \$(4,177,300)
326 29,803,421
) \$(0.14)

The accompanying notes are an integral part of these consolidated financial statements.

VICTORY ENERGY CORPORATION AND SUBSIDIARIES CONSOLIDATED STATEMENTS OF CASH FLOWS

For the years ended December 31, 2016 and 2015

12/31/2016	12/31/2015)
CASH FLOWS FROM OPERATING ACTIVITIES		
Net loss \$(1,906,599)	9) \$(4,905,51	8)
Adjustments to reconcile net loss to net cash used in operating activities		
Accretion and revisions of asset retirement obligations 2,670	66,172	
Amortization of debt discount and financing warrants 40,823	40,823	
Depletion, depreciation, and amortization 132,339	570,337	
Gain on settlement of asset retirement obligation (27,850) (3,721)
Gain on sale of oil and gas properties (125,774) —	
Gain from legal settlement agreement —	(637,248)
Impairment of oil and natural gas properties —	867,048	
Stock based compensation 86,733	567,112	
Stock grants in exchange for services —	169,210	
Change in operating assets and liabilities		
Accounts receivable (6,689) 3,875	
Management fee receivable - affiliate (5,972)) (7,217)
Prepaid expense (1,217) 13,112	

Accounts payable Accrued liabilities - related parties Accrued liabilities Net cash used in operating activities	(1,112,210 684,794 211,872 (2,027,080	•)
CASH FLOWS FROM INVESTING ACTIVITIES: Lease purchases, drilling capital expenditures	(18,442) (1,058,704	`
Proceeds from sale of assets	97,094	(1,036,704)
Net cash provided used by investing activities	78,652	(1,058,704)
CASH FLOWS FROM FINANCING ACTIVITIES:			
Non-controlling interest contributions	2,112,000	2,917,000	
Principal payments on debt financing	(109,500) (120,000)
Net cash provided by financing activities	2,002,500	2,797,000	
Net Change in Cash and Cash Equivalents	54,072	(557)
Beginning Cash and Cash Equivalents	2,384	2,941	
Ending Cash and Cash Equivalents	\$56,456	\$2,384	
Supplemental cash flow information:			
Cash paid for: Interest	\$46,056	\$40,053	
Non-cash investing and financing activities:			
Interest - Accrued interest on Roger's settlement	\$88,060	\$ —	
Accrued capital expenditures	\$230,661	\$293,304	
The accompanying notes are an integral part of these consolidated finance	cial statements	S.	

VICTORY ENERGY CORPORATION AND SUBSIDARIES CONSOLIDATED STATEMENTS OF STOCKHOLDER'S EQUITY

For the years ended December 31, 2016 and 2015

	Common St Par Value Number	tock \$0.001 Amount	Additional Paid In Capital	Accumulated Deficit	Non- controlling Interest	Total Equity (Deficit)
Balance, December 31, 2014	29,202,826	\$29,203	\$34,974,441	\$(40,111,826)	\$3,639,359	\$(1,468,823)
Contributions from noncontrolling interest owners		_	_	_	2,917,000	2,917,000
Stock awards granted	2,017,500	\$2,017	506,722	_		508,739
Stock based compensation			58,373	_		58,373
Stock in exchange for services			169,210			169,210
Net loss			_	(4,177,300	(728,218	(4,905,518)
Balance December 31, 2015	31,220,326	\$31,220	\$35,708,746	\$(44,289,126)	\$5,828,141	\$(2,721,019)
Contributions from noncontrolling interest owners		_	_	_	2,112,000	2,112,000
Stock based compensation			86,733	_	_	86,733
Net loss			_	(1,851,624	(54,975	(1,906,599)
Balance December 31, 2016	31,220,326	\$31,220	\$35,795,479	\$(46,140,750)	\$7,885,166	\$(2,428,885)

The accompanying notes are an integral part of these consolidated financial statements.

Victory Energy Corporation and Subsidiaries Notes to the Consolidated Financial Statements

Note 1 – Organization and Summary of Significant Accounting Policies:

Victory Energy Corporation ("Victory" or "the Company") is an independent, growth oriented oil and natural gas company engaged in the acquisition, exploration and production of oil and natural gas properties, through its partnership with Aurora Energy Partners ("Aurora"). In this report, "the Company" refers to the consolidated accounts and presentation of Victory and Aurora, with the equity of non-controlling interests stated separately. The Company is engaged in the exploration, acquisition, development, and production of domestic oil and natural gas properties. Current operations are primarily located onshore in Texas and New Mexico. The Company was organized under the laws of the State of Nevada on January 7, 1982. The Company is authorized to issue 47,500,000 shares of \$0.001 par value common stock, and has 31,220,326 shares of common stock outstanding as of December 31, 2016. Our corporate headquarters are located at 3355 Bee Caves Rd. Ste. 608, Austin, Texas.

A summary of significant accounting policies followed in the preparation of the accompanying consolidated financial statements is set forth below.

Basis of Presentation and Consolidation:

Victory is the managing partner of Aurora, and holds a 50% partnership interest in Aurora. Aurora, a subsidiary of the Company, is consolidated with Victory for financial statement reporting purposes, as the terms of the partnership agreement that govern the operations of Aurora give Victory effective control of the partnership. The consolidated financial statements include the accounts of Victory and the accounts of Aurora. The Company's management, in considering accounting policies pertaining to consolidation, has reviewed the relevant accounting literature. The Company follows that literature, in assessing whether the rights of the non-controlling interests should overcome the presumption of consolidation when a majority voting or controlling interest in its investee "is a matter of judgment that depends on facts and circumstances". In applying the circumstances and contractual provisions of the partnership agreement, management determined that the non-controlling rights do not, individually or in the aggregate, provide for the non-controlling interest to "effectively participate in significant decisions that would be expected to be made in the ordinary course of business." The rights of the non-controlling interest are protective in nature. All intercompany balances have been eliminated in consolidation. Certain reclassifications of prior year balances have been made to confirm such amounts to current year classifications. These reclassifications have no impact on net income. Non-controlling Interests:

The Navitus Energy Group ("Navitus") is a partner with Victory in Aurora. The two partners each own a 50% interest in Aurora. Victory is the Managing partner and has contractual authority to manage the business affairs of Aurora. The Navitus Energy Group currently has four partners. They are James Capital Consulting, LLC ("JCC"), James Capital Energy, LLC ("JCE"), Rodinia Partners, LLC and Navitus Partners, LLC. Although this partnership has been in place since January 2008, its members and other elements have changed since that time.

The non-controlling interest in Aurora is held by Navitus, a Texas general partnership. As of December 31, 2016, \$7,885,166 was recorded as the equity of the non-controlling interest in our consolidated balance sheet representing Navitus' third-party investment in Aurora, with losses attributable to non-controlling interests of \$54,975 for the year ended December 31, 2016. As of December 31, 2015, \$5,828,141 was recorded as the equity of the non-controlling interest in our consolidated balance sheet representing Navitus' third-party investment in Aurora, with losses attributable to the non-controlling interests of \$728,218 for the year ended December 31, 2015. A total of \$150,000 of previously designated capital contributions by Navitus were redesignated as temporary advances in December 31, 2014 and are included in the accrued liabilities - related parties total as of December 31, 2016 and December 31, 2015. Use of Estimates:

The preparation of our consolidated financial statements in conformity with U.S. Generally Accepted Accounting Principles ("GAAP") requires our management to make estimates and assumptions that affect the reported amounts of assets and liabilities, the disclosure of contingent assets and liabilities at the date of the consolidated financial statements and the reported amounts of revenues and expenses during the reporting period. Actual results could differ from those estimates. Estimates are used primarily when accounting for depreciation, depletion, and amortization ("DD&A") expense, property costs, estimated future net cash flows from proved reserves, cost to abandon and impaired

oil and natural gas properties, taxes, accruals of capitalized costs, operating costs and production revenue, general and administrative costs and interest, purchase price allocation on properties acquired, various common stock, warrants and option transactions, and loss contingencies.

Oil and Natural Gas Properties:

We account for investments in oil and natural gas properties using the successful efforts method of accounting. Under this method of accounting, only successful exploration drilling costs that directly result in the discovery of proved reserves are capitalized. Unsuccessful exploration drilling costs that do not result in an asset with future economic benefit are expensed. All development costs are capitalized because the purpose of development activities is considered to be building a producing system of wells, and related equipment facilities, rather than searching for oil and natural gas. Items charged to expense generally include geological and geophysical costs. Capitalized costs for producing wells leasehold costs of proved properties are amortized on a unit-of-production basis over the remaining life of proved developed and total proved reserves, respectively.

We depreciate other property and equipment using the straight-line method based on estimated useful lives ranging from five to ten years.

We review our proved oil and gas properties for impairment whenever events and circumstances indicate that a decline in the recoverability of their carrying value may have occurred. We estimate the expected undiscounted future cash flows of our oil and gas properties and compare such undiscounted future cash flows to the carrying amount of the oil and gas properties to determine if the carrying amount is recoverable. If the carrying amount exceeds the estimated undiscounted future cash flows, we will adjust the carrying amount of the oil and gas properties to fair value. The factors used to determine fair value are subject to our judgment and expertise and include, but are not limited to, recent sales prices of comparable properties, the present value of future cash flows, net estimated operating and development costs using estimates of proved reserves, future commodity pricing, future production estimates, anticipated capital expenditures, and various discount rates commensurate with the risk and current market conditions associated with realizing the expected cash flows projected. Because of the uncertainty inherent in these factors, we cannot predict when or if future impairment charges for proved properties will be recorded.

The assessment of unproved properties to determine any possible impairment requires significant judgment. We assess our unproved properties to determine any possible impairment on a property-by-property basis based on remaining lease terms, drilling results or future plans to develop acreage. Due to the uncertainty inherent in these factors, we cannot predict the amount of impairment charges that may be recorded in the future.

The Company recorded impairment expense of \$0 and \$867,048 for 2016 and 2015 based on the analysis discussed above.

Asset Retirement Obligations:

The Company records the estimate of the fair value of liabilities related to future asset retirement obligations ("ARO") in the period the obligation is incurred. Asset retirement obligations relate to the removal of facilities and tangible equipment at the end of an oil and natural gas property's useful life. The application of this rule requires the use of management's estimates with respect to future abandonment costs, inflation, market risk premiums, useful life and cost of capital and required government regulations. U.S. GAAP requires that the estimate of our asset retirement obligations does not give consideration to the value the related assets could have to other parties.

The following table is a reconciliation of the ARO liability for the twelve months ended December 31, 2016 and 2015.

	Years Ended		
	December 31,		
	2016	2015	
Asset retirement obligation at beginning of period	\$109,171	\$44,214	
Liabilities incurred	_	2,506	
Revisions to previous estimates and sales of properties	_	60,832	
Liabilities on properties sold or settled	(27,850	(3,721)	
Accretion expense	2,670	\$5,340	
Asset retirement obligation at end of period	\$83,991	\$109,171	

Other Property and Equipment:

Our office equipment in Austin, Texas is being depreciated on the straight-line method over the estimated useful life of five to seven years.

Cash and Cash Equivalents:

The Company considers all liquid investments with original maturities of three months or less from the date of purchase that are readily convertible into cash to be cash equivalents. The Company had no cash equivalents at December 31, 2016 and 2015.

Accounts Receivable:

Our accounts receivable are primarily from purchasers of natural gas and oil and exploration and production companies which own an interest in properties we operate.

Allowance for Doubtful Accounts:

The Company recognizes an allowance for doubtful accounts to ensure trade receivables are not overstated due to uncollectibility. Allowance for doubtful accounts are maintained for all customers based on a variety of factors, including the length of time receivables are past due, macroeconomic conditions, significant one-time events and historical experience. An additional allowance for individual accounts is recorded when we become aware of a customer's inability to meet its financial obligations, such as in the case of bankruptcy filings or deterioration in the customer's operating results or financial position. If circumstances related to customers change, estimates of the recoverability of receivables would be further adjusted. As of December 31, 2016 and 2015, the Company has deemed \$0 and \$200,000, respectively from the sale of oil and gas properties associated with the Jones County prospect, to be doubtful and thus, has recorded this amount as an allowance for doubtful accounts.

Fair Value:

At December 31, 2016 and 2015, the carrying value of the Company's financial instruments such as prepaid expenses and payables approximated their fair values based on the short-term maturities of these instruments. The carrying value of other liabilities approximated their fair values because the underlying interest rates approximate market rates at the balance sheet dates. Management believes that due to the Company's current credit worthiness, the fair value of debt could be less than the book value; however, due to current market conditions and available information, the fair value of such debt is not readily determinable. Financial Accounting Standard Board ("FASB") ASC Topic 820 established a hierarchical disclosure framework associated with the level of pricing observability utilized in measuring fair value. This framework defined three levels of inputs to the fair value measurement process and requires that each fair value measurement be assigned to a level corresponding to the lowest level input that is significant to the fair value measurement in its entirety. The three broad levels of inputs defined by FASB ASC Topic 820 hierarchy are as follows:

Level 1 - quoted prices (unadjusted) in active markets for identical assets or liabilities that the reporting entity has the ability to access at the measurement date;

Level 2 - inputs other than quoted prices included within Level 1 that are observable for the asset or liability, either directly or indirectly. If the asset or liability has a specified (contractual) term, a Level 2 input must be observable for substantially the full term of the asset or liability; and

Level 3 - unobservable inputs for the asset or liability. These unobservable inputs reflect the entity's own assumptions about the assumptions that market participants would use in pricing the asset or liability and are developed based on the best information available in the circumstances (which might include the reporting entity's own data).

The initial measurement of asset retirement obligations is calculated using discounted cash flow techniques and based on internal estimates of future retirement costs associated with proved oil and gas properties. Inputs used in the calculation of asset retirement obligations include plugging costs and reserve lives, which are considered Level 3 inputs. A reconciliation of Victory's asset retirement obligations is presented in Note 1.

During 2015, proved oil and gas properties with a carrying value of \$1,640,147 were written down, based upon engineering estimates, to their fair value of \$759,091 as a result of \$867,048 in impairment charges. Of this impairment amount, \$303,312 was taken against the Eagle Ford properties, \$297,212 was taken against the Adams Baggett properties, and \$99,682 was taken against the Fairway properties. In addition, the Company has written off the entire balance associated with undeveloped properties, or \$166,842.

Revenue Recognition:

The Company uses the sales method of accounting for oil and natural gas revenues. Under this method, revenues are recognized based on actual volumes of gas and oil sold to purchasers. The volumes sold may differ from the volumes to which the Company is entitled based on our interests in the properties. Differences between volumes sold and entitled volumes create oil and natural gas imbalances which are generally reflected as adjustments to reported proved oil and natural gas reserves and future cash flows in their supplemental oil and natural gas disclosures. If their excess takes of natural gas or oil exceed their estimated remaining proved reserves for a property, a natural gas or oil imbalance liability is recorded in the Consolidated Balance Sheets.

Concentrations:

There is a ready market for the sale of crude oil and natural gas. During 2016 and 2015, our gas field and our producing wells sold their respective gas and oil production to one purchaser for each field or well. However, because alternate purchasers of oil and natural gas are readily available at similar prices, we believe that the loss of any of our purchasers would not have a material adverse effect on our financial results. A majority of the Company's production and reserves are from the Eagle Ford property in South Texas and the Permian Basin of West Texas.

Earnings per Share:

Basic earnings per share are computed using the weighted average number of common shares outstanding at December 31, 2016 and December 31, 2015, respectively. The weighted average number of common shares outstanding was 31,220,326 at December 31, 2016. Diluted earnings per share reflect the potential dilutive effects of common stock equivalents such as options, warrants and convertible securities. Given the historical and projected future losses of the Company, all potentially dilutive common stock equivalents are considered anti-dilutive. The following table outlines outstanding common stock shares and common stock equivalents.

	Years Ended December 31,	
	2016	2015
Common Stock Shares Outstanding	31,220,326	31,220,326
Common Stock Equivalents Outstanding		
Warrants	11,392,386	8,622,486
Stock Options	1,055,000	1,430,000
Unconverted Class B Shares	137,932	137,932
Total Common Stock Equivalents Outstanding	12,585,318	10,190,418

Income Taxes:

The Company accounts for income taxes in accordance with ASC 740 "Income Taxes" which requires an asset and liability approach for financial accounting and reporting of income taxes. Deferred income taxes reflect the impact of temporary differences between the amount of assets and liabilities for financial reporting purposes and such amounts as measured by tax laws and regulations. Deferred tax assets include tax loss and credit carry forwards and are reduced by a valuation allowance if, based on available evidence, it is more likely than not that some portion or all of the deferred tax assets will not be realized.

Stock-Based Compensation:

The Company applies ASC 718, "Compensation-Stock Compensation" to account for the issuance of options and warrants to employees, key partners, directors, officers and Navitus investors. The standard requires all share-based payments, including employee stock options, warrants and restricted stock, be measured at the fair value of the award and expensed over the requisite service period (generally the vesting period). The fair value of options and warrants granted to employees, directors and officers is estimated at the date of grant using the Black-Scholes option pricing model by using the historical volatility of the Company's stock price. The calculation also takes into account the common stock fair market value at the grant date, the exercise price, the expected term of the common stock option or warrant, the dividend yield and the risk-free interest rate.

The Company from time to time may issue stock options, warrants and restricted stock to acquire goods or services from third parties. Restricted stock, options or warrants issued to third parties are recorded on the basis of their fair value, which is measured as of the date issued. The options or warrants are valued using the Black-Scholes option

pricing model on the basis of the market price of the underlying equity instrument on the "valuation date," which for options and warrants related to contracts that have substantial disincentives to non-performance, is the date of the contract, and for all other contracts is the vesting date. Expense related to the options and warrants is recognized on a straight-line basis over the shorter of the period over which services are to be received or the vesting period and is included in general and administrative expenses in the accompanying consolidated statements of operations. The Company recognized stock-based director's compensation expense from warrants and stock awards granted to directors for services of \$0 and \$508,739, for the years ended December 31, 2016 and 2015, respectively. The Company recognized stock-based incentive compensation expense from stock options granted to officers and employees of the Company of \$86,733 and \$58,373 for the twelve months ended December 31, 2016 and 2015, respectively.

Going Concern:

The accompanying consolidated financial statements have been prepared assuming the Company will continue as a going concern, which contemplates the realization of assets and satisfaction of liabilities in the normal course of business. As presented in the consolidated financial statements, the Company has incurred a net loss of \$1,906,599 and \$4,905,518 during the years ended December 31, 2016 and 2015, respectively. Non-cash expenses and allowances were significant during the years ended December 31, 2016 and December 31, 2015, and the net cash used in operating activities, or negative cash flows from operating activities, were \$2,027,080 and \$1,738,853, respectively.

The cash proceeds from new contributions to the Aurora partnership by Navitus, and loans from affiliates have allowed the Company to continue operations and invest in new oil and natural gas properties. See Note 4. Management anticipates that operating losses will continue in the near term until new wells are drilled, successfully completed and incremental production increases revenue. On a year to date basis, as of December 31, 2016 the Company has invested \$18,442 in the drilling of wells and \$0 in the acquisition of oil and gas properties. The Company remains in active discussions with Navitus and others related to longer term financing required for our capital expenditures planned for 2017. Without additional outside investment from the sale of equity securities and/or debt financing, our capital expenditures and overhead expenses must be reduced to a level commensurate with available cash flows.

The accompanying consolidated financial statements are prepared as if the Company will continue as a going concern. The consolidated financial statements do not contain adjustments, including adjustments to recorded assets and liabilities, which might be necessary if the Company were unable to continue as a going concern.

Note 2 - Recent accounting pronouncements

In January 2017, the FASB issued ASU 2017-01, Business Combinations (Topic 805): Clarifying the Definition of a Business , which changes the definition of a business to assist entities with evaluating when a set of transferred assets and activities is deemed to be a business. Determining whether a transferred set constitutes a business is important because the accounting for a business combination differs from that of an asset acquisition. The definition of a business also affects the accounting for dispositions. Under the new standard, when substantially all of the fair value of assets acquired is concentrated in a single asset, or a group of similar assets, the assets acquired would not represent a business and business combination accounting would not be required. The new standard may result in more transactions being accounted for as asset acquisitions rather than business combinations. The standard is effective for interim and annual periods beginning after December 15, 2017 and shall be applied prospectively. Early adoption is permitted. The Company has elected to early adopt ASU 2017-01 on January 1, 2017 and will apply the new guidance to applicable transactions occurring after that date.

In March 2016, the FASB issued guidance regarding the simplification of employee share-based payment transactions, including the income tax consequences, classification of awards as either equity or liabilities, and classification on the statement of cash flows. We adopted this guidance in the second quarter of 2016 as permitted by the guidance. Adoption of this guidance did not impact our financial statements, except for the simplification in accounting for income taxes using a modified retrospective approach. Upon adoption, we recorded a related deferred tax asset for previously unrecognized excess tax benefits of \$37 million. As we consider it more likely than not that the deferred tax asset will not be realized, we recorded a full valuation allowance of \$37 million, resulting in no net effect on our consolidated statement of operations. We elected to continue our current policy of estimating forfeitures.

In February 2016, the FASB issued guidance regarding the accounting for leases. The guidance requires recognition of most leases on the balance sheet. The guidance requires lessees and lessors to recognize and measure leases at the beginning of the earliest period presented using a modified retrospective approach. The guidance is effective for interim and annual periods beginning after December 15, 2018. We are currently evaluating the impact of this guidance on our consolidated financial statements.

In January 2016, the FASB issued guidance regarding several broad topics related to the recognition and measurement of financial assets and liabilities. The guidance is effective for interim and annual periods beginning after December 15, 2017. We do not expect this guidance to have a material impact on our consolidated financial statements. In April 2015, the FASB issued ASU 2015-03, "Interest—Imputation of Interest" (Subtopic 835-30): "Simplifying the Presentation of Debt Issuance Costs." Entities that have historically presented debt issuance costs as an asset, related to a recognized debt liability, will be required to present those costs as a direct deduction from the carrying amount of that debt liability. The ASU does not change the recognition, measurement, or subsequent measurement guidance for debt issuance costs. In August 2015, the FASB issued ASU 2015-15, "Interest—Imputation of Interest" (Subtopic 835-30), which addresses the presentation or subsequent measurement of debt issuance costs related to line-of-credit arrangements, given the absence of authoritative guidance within ASU 2015-03 for debt issuance costs related to line-of-credit arrangements. The amendments are effective for interim and annual reporting periods beginning after December 15, 2015. Therefore, the Company adopted ASU 2015-03 beginning January 1, 2016. Changes to the balance sheet have been applied on a retrospective basis. This resulted in the reclassification of debt issuance costs of \$6,237 associated with our Credit Agreement from Other Assets to Current Note Payable in the Consolidated Balance Sheet as of year ended December 31, 2016 and 2015.

In May 2014, the FASB issued guidance regarding the accounting for revenue from contracts with customers. In April 2016, May 2016 and December 2016, the FASB issued additional guidance, addressed implementation issues and provided technical corrections. The guidance may be applied retrospectively or using a modified retrospective approach to adjust retained earnings (deficit). The guidance is effective for interim and annual periods beginning after December 15, 2017. We are currently evaluating the impact of this guidance on our consolidated financial statements. Note 3 – Oil and natural gas properties

Oil and natural gas properties are comprised of the following:

	December 31,	
	2016	2015
Proved property	\$9,695,367	\$9,940,660
Unproved property	1,375,940	1,375,940
Total oil and natural gas properties, at cost	11,071,307	11,316,600
Less: accumulated impairment	(8,283,321)	(8,283,321)
Oil and natural gas properties, net of impairment	2,787,986	3,033,279
Less: accumulated depletion	(2,166,643)	(2,274,188)
Oil and natural gas properties, net	\$621,343	\$759,091

Depletion, depreciation, and amortization expense related to oil and natural gas properties for the years ended December 31, 2016 and 2015 was \$125,744 and \$637,121, respectively. During the years ended December 31, 2016 and 2015, the Company recorded impairment losses of \$0 and \$867,048, respectively. As a result of the impairment charges incurred for the year ended December 31, 2015, the Company's unproved property asset base has zero net book value as of December 31, 2016 and December 31, 2015.

In 2016, as part of the legal settlement of the Trilogy lawsuit, the Company assigned its interests in the seven wells. Additionally, the Company sold its interest in the Morgan #1 well. As a result, a gain was recorded in the amount of \$64,824. See Note 5 for additional information.

Note 4 - Acquisitions and Dispositions

During February of 2015, Victory entered into a letter of intent ("LOI") and subsequently into (a) the Pre-Merger Collaboration Agreement (the "Collaboration Agreement") with Lucas Energy Inc. ("Lucas"), Navitus and AEP Assets, LLC ("AEP"), a wholly-owned subsidiary of Aurora; and (b) the Pre-Merger Loan and Funding Agreement (the "Loan

Agreement") with Lucas. During March of 2015 the parties entered into Amendment No. 1 to the Pre-Merger Collaboration which amendments affected thereby are included in the discussion of the Collaboration Agreement below. Payments of \$195,928 and \$317,027 were made by Aurora, on behalf of Victory, to Earthstone Energy/Oak Valley Resources and Penn Virginia, respectively, pursuant to the Pre-Merger Collaboration Agreement for costs related to the two Earthstone Energy/Oak Valley Resources and the five Penn Virginia operated Eagle Ford wells, respectively.

The initial draw, and any other amounts borrowed by Lucas under the Loan Agreement were evidenced by a Secured Subordinated Delayed Draw Term Note issued by Lucas in favor of Victory, which was in an initial amount of \$250,000 (the "Draw Note"). Borrowings evidenced by the Draw Note accrued interest at 0.5% per annum, with accrued interest payable in one lump sum on maturity. The maturity date of the Draw Note was February 26, 2016. A total of \$600,000 was paid to Lucas under the Draw Note.

Subsequent to March 31, 2015, the Company terminated the LOI and notified Lucas pursuant to the Loan Agreement, that it would not extend any further credit to Lucas under the Loan Agreement. Merger related direct costs, as well as cost related to reserving amounts receivable on advances made under the Collaboration Agreement totaled approximately \$1,326,850 and are included in General and Administrative expenses for the twelve months ended December 31, 2015. There were zero associated costs incurred during the twelve months ended December 31, 2016. Further, the Company entered into (1) a Settlement Agreement and Mutual Release (the "Lucas Settlement Agreement") with Lucas, (2) a Settlement Agreement and Mutual Release (the "Rogers Settlement Agreement") with Louise H. Rogers, ("Rogers"), and (3) a Compromise Settlement Agreement and Mutual General Release, effective as of September 25, 2015 (the "Earthstone Settlement Agreement", and, together with the Lucas Settlement Agreement and the Rogers Settlement Agreement, the "Settlement Agreements") with Earthstone Operating, LLC, Earthstone Energy, Inc., Oak Valley Resources, LLC, Oak Valley Operating LLC and Sabine River Energy, LLC (collectively, "Earthstone"), Lucas, AEP, and Aurora.

Lucas Settlement Agreement

The Company and Lucas agreed to terminate any and all obligations between the parties arising under the LOI and the Collaboration Agreement. The Company and Lucas further agreed that the Company would retain ownership and control over five Penn Virginia well-bores previously assigned by Lucas to the Company (the "Penn Virginia Well-Bores"), as well as the obligations to pay the expenses associated with such Penn Virginia Well-Bores effective after August 1, 2014. Under the terms of the Lucas Settlement Agreement, Lucas agreed to assign to the Company all of Lucas' rights in a certain oil and gas property located in the same field as the Penn Virginia Well-Bores (the "Additional Penn Virginia Property"), including the rights to all revenues from all wells on some properties. Rogers Settlement and Amended Rogers Settlement Agreements

The Company and Rogers agreed, among other things, (i) to terminate the contingent promissory note in the principal amount of \$250,000 payable to Rogers that was issued by Victory in connection with the entry by Lucas and the Company into the Collaboration Agreement, (ii) that the Company would pay Rogers, on or before July 15, 2015, \$258,125, and (iii) that Rogers' legal counsel will hold the assignment of the Additional Penn Virginia Property and the Settlement Shares in escrow until such time as the payment of \$258,125 is made by the Company to the Rogers. Failure of the Company to make the payment of \$258,125 on or before July 15, 2015, would result in the Company being in default under the Rogers Settlement Agreement and default interest on the amount due would begin to accrue at a per diem rate of \$129.0625. Additionally, the Company acknowledged in the Amendment its obligation to pay Rogers' attorney's fees in the amount of \$22,500. The Company has not made any payments to Rogers pursuant to the Rogers Settlement Agreement and as a result the additional Penn Virginia Property was returned to Lucas in September 2015. The full amount due under the Roger's obligation including accrued interest at December 31, 2016 totals \$349,916 and is included in accrued liabilities on the consolidated balance sheet.

Earthstone Settlement Agreement

The Company assigned to Earthstone certain oil and gas interests in the wells which were previously transferred to the Company by Lucas in February 2015. The Company and Earthstone agreed to release each other from any and all claims, demands and causes of action which either party had against the other prior to the effective date of the Earthstone Settlement Agreement, whether known or unknown, except in connection with the breach, enforcement or interpretation of the Earthstone Settlement Agreement. Lucas and Earthstone similarly agreed to release each other from such claims pursuant to the terms of the Earthstone Settlement Agreement. The Company

charged \$195,928 related to the Earthstone Settlement Agreement to General and Administrative expenses as a cost of the merger termination during the twelve-month period ended December 31, 2015.

Note 5 - Gain from Settlement Agreement and Sale of Oil and Gas Properties

During the year ended December 31, 2016, the Company sold undeveloped acreage in Howard County, Texas for \$88,800 which was recorded as a gain on sale of oil and gas properties, as there was no remaining book value associated with this property.

During the year ended December 31, 2015, Aurora entered into a Settlement Agreement and Release (the "Settlement Agreement and Release") to settle the outstanding litigation between Aurora and Trilogy in the case styled Trilogy Operating, Inc. v. Aurora Energy Partners, which was pending in Howard County, Texas (the "Litigation"). Pursuant to the Settlement Agreement and Release, Aurora agreed to assign any and all of its interests in four specified wells located in Glasscock and Howard Counties, those being Wagga Wagga #2, Homar #1, Ballarat '185' #1 and BOA North #5 (collectively, the "Obligation Wells"). The Company has not historically included any production or reserve information in its financial or operational reporting in any of its prior filings for these Obligation Wells. A separate, but related, lawsuit between Trilogy Operating, Inc. and Aurora Energy Partner dated January 6, 2016 alleged causes of action for a suit on a sworn account, breach of contract and a suit to foreclose on liens regarding the drilling and completion of seven wells. On May 2, 2016, a Joint Motion to Dismiss with Prejudice was granted by the court resulting in the assignment of these seven wells, BOA 12 #1, BOA 12 #3, BOA 12 #4 North, Darwin #1, Darwin #2, Darwin #3 and Wagga Wagga #1. The liabilities associated with these wells exceeded the asset value by \$64,824 which was recorded as a gain on legal settlement during the year ended December 31, 2016. The Company recorded these costs, billed to it by the operator, in 2014 to oil and gas property acquisitions. In accordance with the Company's impairment policy these costs were charged to impairment expense in the Company's Consolidated Statement of Operations for the year ending December 31, 2014. Due to continuing litigation the related joint interest payable balance to the operator remained outstanding until the settlement on November 21, 2015. This settlement included the reversal or cancellation of all related outstanding joint interest billings payable to the operator. The Company therefore recorded a \$637,248 non-cash gain on the settlement of this matter in the Company's Consolidated Statement of Operations for the year ending December 31, 2015.

Note 6- Liability for Unauthorized Preferred Stock Issued

During the year ended December 31, 2006, the Company authorized the issuance of 10,000,000 shares of Preferred Stock, convertible at the shareholder's option to common stock at the rate of 100 shares of common stock for every share of preferred stock. During the year ended December 31, 2006, the Company issued 715,517 shares of preferred stock for cash of \$246,950. The Company subsequently issued additional preferred stock and had several preferred shareholders convert their shares into common stock during the years ended December 31, 2009, 2008, and 2007. The Company's legal counsel determined that the preferred shares had not been duly authorized by the State of Nevada. Since the Company had issued and received consideration for the preferred stock, notwithstanding that the stock was not legally authorized, the Company has presented the preferred stock as a liability in the consolidated balance sheets. The Company has offered to settle the debt with the remaining holders of the unauthorized preferred stock by honoring the terms of conversion of two shares of preferred stock into 100 shares of common stock. The Company intends to cancel the preferred stock once all remaining preferred stockholders have converted. There were 68,966 and 68,966 shares of unconverted preferred stock outstanding at December 31, 2016 and 2015, respectively. The Company needs approximately 138,000 common shares in order to settle the outstanding debt as stated below.

The remaining liability for the unconverted preferred stock is based on the original cash tendered and consisted of the following as of:

December 31, 2016 2015

Liability for unauthorized preferred stock \$9,283 \$9,283

Note 7 - Revolving Credit Agreement

On February 20, 2014, Aurora, as borrower, entered a credit agreement (the "Credit Agreement") with Texas Capital Bank ("the Lender"), Guarantors on the Credit Agreement are Victory and Navitus, the two partners of Aurora, Pursuant to the Credit Agreement, the Lender agreed to extend credit to Aurora in the form of (a) one or more revolving credit loans (each such loan, a "Loan") and (b) the issuance of standby letters of credit, of up to an aggregate principal amount at any one time not to exceed the lesser of (i) \$25,000,000 or (ii) the borrowing base in effect from time to time (the "Commitment"). The initial borrowing base on February 20, 2014 was set at \$1,450,000. The borrowing base is determined by the Lender, in its sole discretion, based on customary lending practices, review of the oil and natural gas properties included in the borrowing base, financial review of Aurora, the Company and Navitus and such other factors as may be deemed relevant by the Lender. The borrowing base is re-determined (i) on or about June 30 of each year based on the previous December 31 reserve report prepared by an independent reserve engineer, and (ii) on or about August 31 of each year based on the previous June 30 reserve report prepared by Aurora's internal reserve engineers or an independent reserve engineer and certified by an officer of Aurora. The Credit Agreement will mature on February 20, 2017. Amounts borrowed under the Credit Agreement will bear interest at rates equal to the lesser of (i) the maximum rate of interest which may be charged or received by the Lender in accordance with applicable Texas law and (ii) the interest rate per annum publicly announced from time to time by the Lender as the prime rate in effect at its principal office plus the applicable margin. The applicable margin is, (i) with respect to Loans, one percent (1.00%) per annum, (ii) with respect to letter of credit fees, two percent (2.00%) per annum and (iii) with respect to commitment fees, one-half of one percent (0.50%) per annum. Loans made under the Credit Agreement are secured by (i) a first priority lien in the oil and gas properties of Aurora, the Company and Navitus, and (ii) a first priority security interest in substantially all of the assets of Aurora and its subsidiaries, if any, as well as in 100% of the partnership interests in Aurora held by the Company and Navitus. Loans made under the Credit Agreement to Aurora are fully guaranteed by the Company and Navitus.

The Credit Agreement contains various affirmative and negative covenants. These covenants, among other things, limit additional indebtedness, additional liens and transactions with affiliates. Among the covenants contained in the Credit Agreement are financial covenants that Aurora will maintain a minimum EBITDAX to Cash Interest Ratio of 3.5 to 1.0 and a minimum Current Ratio of not less than 1.0 to 1.0. The Current Ratio is defined under the covenants to include, as a current asset, the revolving credit availability.

On April 13, 2015, the Company received the annual Borrowing Base Adjustment called for under the terms of the Credit Agreement, which called for a decrease in the borrowing base of \$300,000 payable by May 13, 2015, and an increase in the monthly reduction amount to \$10,000 commencing as of June 1, 2015. Additionally, the Lender notified Aurora that, based on the Lender's redetermination of Aurora's borrowing base, the monthly reduction amount under the Credit Agreement was increased, commencing on June 1, 2015, from \$0 to \$10,000. Pursuant to this increase in the monthly reduction amount, Aurora's borrowing base will be automatically reduced by \$10,000 on the first day of each calendar month beginning on June 2015 until the Lender's next periodic borrowing base redetermination. The Company made one payment in the amount of \$10,000 in June 2015.

On May 13, 2015, Aurora informed the Lender it would not make the required \$300,000 payment but was submitting the newly acquired five Eagle Ford wells as additional collateral to be considered and its willingness to execute mortgages regarding the properties to meet the Deficiency.

On August 21, 2015, the Company executed a Forbearance Agreement whereby the Lender would forbear all existing events of default which includes all payments under the previously mentioned Borrowing Base Deficiency payments not yet paid under the April 13, 2015 Redetermination Date notification, as well as the late interest payments for June, July and August 2015, violations of Aurora financial covenants for the three months ended March 31, 2015, and June 30, 2015, and default notice for the late filing of March 31, 2015 financial reports. On August 26, 2015, the Company paid the Lender \$76,081 to cover a portion of the deficiency payment, as well as a Forbearance document fee and Lender's legal expenses, as required by the Forbearance Agreement, and the aforementioned Forbearance Agreement went into effect for the \$260,000 remaining borrowing base deficiency payment. On August 31, 2015, the Forbearance Agreement terminated pursuant to its terms. The Company made a \$50,000 principal payment to the lender on October 14, 2015.

On December 5, 2016, the Company entered into a new Forbearance Agreement to the Credit Agreement. Pursuant to the Forbearance Agreement, the Lender has agreed to forbear from exercising any of its rights and remedies under the Credit Agreement until February 20, 2017 with respect to the historical events of default.

The Forbearance Period was amended and extended on March 2, 2017 and will end on the first to occur of the following: (i) the expiration of the amended Forbearance Period on August 20, 2017, (ii) a breach by Aurora or any Guarantor of any of the conditions, covenants, representations and/or warranties set forth in the Forbearance Agreement, (iii) the occurrence of any new event of default under the Credit Agreement, (iv) the occurrence or threat of the occurrence of any enforcement action against Aurora or any Guarantor by any of their creditors which, in Lender's reasonable judgment, would materially interfere with the operation of Aurora's or the Guarantor's business or the Lender's ability to collect on the obligations due under the Credit Agreement, (v) the institution of any bankruptcy proceeding relating to Aurora or any Guarantor, or (vi) the initiation by Aurora or any Guarantor of any judicial, administrative or arbitration proceedings against the Lender. The Lender's agreement to forbear from exercising its rights and remedies as a result of the Existing Events of Default is subject to and conditioned upon the following: (i) the payment by Aurora to the Lender of at least \$20,000 on or before the last business day of each calendar week occurring hereafter and (ii) the delivery by Aurora of such other documents, instruments and certificates as reasonably requested by Lender. The foregoing description of the Forbearance Agreement is a summary only and is qualified in its entirety by reference to the complete text of the Forbearance Agreement. Since the execution of the extended Forbearance Agreement, the Company has paid the Lender \$190,500. The balance owed on the Credit Agreement as of December 31, 2016 was \$570,500.

As of December 31, 2016 and 2015, the Company was out of compliance with the Current Ratio and with the EBITDAX to Cash Interest Ratio due to its reduced revenue streams from price and production declines and continued high general and administrative expenses. Therefore, the Company is in technical default of the Credit Agreement and related agreements. The Company has not yet been advised by the Lender of any additional actions the Lender plans to take.

Amortization of debt financing costs on this debt was \$40,823 and \$40,823 for the twelve months ended December 31, 2016 and December 31, 2015, respectively. Interest expense related to the Credit Agreement was \$46,056 and 40,053 for the twelve months ended December 31, 2016, and December 31, 2015, respectively. Note 8 – Income Taxes

There was no provision for (benefit of) income taxes for the years ended December 31, 2016 and 2015, after the application of ASC 740 "Income Taxes."

The Internal Revenue Code of 1986, as amended, imposes substantial restrictions on the utilization of net operating losses in the event of an "ownership change" of a corporation. Accordingly, a company's ability to use net operating losses may be limited as prescribed under Internal Revenue Code Section 382 ("IRC Section 382"). Events which may cause limitations in the amount of the net operating losses that the Company may use in any one year include, but are not limited to, a cumulative ownership change of more than 50% over a three-year period. There have been transactions that have changed the Company's ownership structure since inception that may have resulted in one or more ownership changes as defined by the IRC Section 382. The Company's stock issuance arising from convertible debt in 2012 has resulted in a limitation of net operating loss carry forward for the Company of \$13,807,335 over a 20-year period.

At December 31, 2016, the Company had available Federal operating loss carry forwards to reduce future taxable income. Additional Federal net operating loss carry forward of \$1,601,551 for 2016 would make available approximately \$21,775,380 as of December 31, 2016. The Federal net operating loss carry forwards begin to expire in 2028. Capital loss carryovers may only be used to offset capital gains.

Given the Company's history of net operating losses, management has determined that it is more likely than not the Company will not be able to realize the tax benefit of the net operating loss carry forwards. ASC 740 requires that a valuation allowance be established when it is more likely than not that all or a portion of deferred tax assets will not be realized. Accordingly, the Company has recorded a full valuation allowance against its net deferred tax assets at December 31, 2016 and 2015, respectively. Upon the attainment of taxable income by the Company, management will assess the likelihood of realizing the deferred tax benefit associated with the use of the net operating loss carry forwards and will recognize a deferred tax asset at that time.

Significant components of the Company's deferred income tax assets are as follows:

	December 31,	December 31,
	2016	2015
Net operating loss carry forward	\$7,403,629	\$6,859,102
Depreciation and accretion	3,106	7,222
Equity based expenses	86,734	1,920,230
Impairment losses on oil and gas properties	_	1,559,951
Deferred taxes	7,493,469	10,346,505
Valuation allowance	(7,493,469)	(10,346,505)
Net Deferred Income Tax Assets	\$ —	\$ —

Reconciliation of the effective income tax rate to the U.S. statutory rate is as follows:

	12/31/2	2016	12/31/2	2015
Net operating loss	34	%	34	%
Meals and entertainment	1	%	15	%
Debt discount accretion	0.11	%	0.10	%
Net operating loss reduction due to IRC 382	_			%
Change in valuation allowance	33.89	%	33.75	%
Effective income tax rate		%		%

ASC 740 provides guidance which addresses the determination of whether tax benefits claimed or expected to be claimed on a tax return should be recorded in the consolidated financial statements. Under the current accounting guidelines, the Company may recognize the tax benefit from an uncertain tax position only if it is more likely than not that the tax position will be sustained on examination by the taxing authorities, based on the technical merits of the position. The tax benefits recognized in the consolidated financial statements from such a position should be measured based on the largest benefit that has a greater than fifty percent likelihood of being realized upon ultimate settlement. As of December 31, 2016 and 2015 the Company does not have a liability for unrecognized tax benefits.

The Company has elected to include interest and penalties related to uncertain tax positions as a component of income tax expense. To date, no penalties or interest has been accrued.

Tax years 2011 forward are open and subject to examination by the Federal taxing authority. The Company is not currently under examination and it has not been notified of a pending examination.

Note 9 – Stockholders' Equity

Long-Term Incentive Plan

On February 24, 2014, the Board of Directors (the "Board") of the Victory Energy Corporation (the "Company") approved and adopted the Victory Energy Corporation 2014 Long Term Incentive Plan (the "LTIP") for the employees, directors and consultants of the Company and its affiliates. The LTIP provides for the grant of all or any of the following components: (1) stock options, (2) restricted stock, (3) other stock-based awards, (4) performance awards and (5) dividends and dividend equivalents. Subject to adjustment in accordance with the LTIP, the maximum aggregate number of shares of the common stock of the Company, par value \$0.001 per share (the "Common Stock") that may be issued with respect to awards under the LTIP is fifteen percent (15%) of the outstanding shares of Common Stock at the end of the preceding calendar quarter, of which the maximum number of such shares that may be issued as incentive stock options, as defined in Section 422(b) of the Internal Revenue Code of 1986 is two million (2,000,000) shares of Common Stock, Common Stock withheld to satisfy exercise prices or tax withholding obligations will be available for delivery pursuant to other awards. The LTIP will be administered by the Board, until such time as a compensation committee of the Board is established (the "Compensation Committee"), at which time the LTIP will be administered by the Compensation Committee. The total number of shares of common stock initially available for issuance under the LTIP was 4,591,174. As of December 31, 2016, 3,367,500 shares of unrestricted common stock and 595,000 options were issued under the LTIP. The maximum contractual term is five years. As of December 31, 2016, 628,674 shares of common stock are available for issuance under the LTIP.

Stock Based Compensation

The Company estimates the fair value of employee stock options and warrants granted using the Black-Scholes Option Pricing Model. Key assumptions used to estimate the fair value of warrants and stock options include the exercise price of the award, the fair value of the Company's common stock on the date of grant, the expected warrant or option term, the risk free interest rate at the date of grant, the expected volatility and the expected annual dividend yield on the Company's common stock.

During the year ended December 31, 2016, the Company did not grant stock awards to directors, officers, or employees. During the year ended December 31, 2015, the Company granted 2,017,500 stock awards to directors, officers, and employee's at fair value of the stock on the date of issuance of \$508,739.

Note 10 – Warrants for Stock

At December 31, 2016 and 2015 warrants outstanding for common stock of the Company were as follows:

	Number of Shares Underlying Warrants	Weighted Average Exercise Price
Balance at January 1, 2016	8,622,486	\$ 0.48
Granted	2,162,000	_
Exercised	_	_
Canceled	(96,400)	0.50
Balance at December 31, 2016	10,688,086	\$ 0.48

	Number of Shares Underlying Warrants	Weighted Average Exercise Price
Balance at January 1, 2015	5,937,386	\$ 0.66
Granted	2,917,000	0.29
Exercised	_	_
Canceled	(231,900)	2.18
Balance at December 31, 2015	8,622,486	\$ 0.48

During the year ended December 31, 2016, the Company granted 2,112,000 warrants for \$2,112,000 in capital contributions through Navitus Partners, LLC valued with the Black Scholes pricing model. The Company also granted 50,000 warrants in exchange for services during the year ended December 31, 2016.

The following table summarizes information about underlying outstanding warrants for common stock of the Company outstanding and exercisable as of December 31, 2016:

	Warrants Ou	itstanding		Warrants Ex	ercisable
	Number of	Weighted	Weighted Average	Number of	Weighted
Range of Exercise Prices	Shares	Average	Remaining	Shares	Average
Range of Exercise Prices	Underlying	Exercise	Contractual Life (in years)	Underlying	Exercise
	Warrants	Price		Warrants	Price
\$12.50 - \$17.50	104,845	\$ 12.50	6.79	104,845	\$ 12.50
\$0.13 - \$2.50	10,583,241	\$ 0.30	2.88	10,583,241	\$ 0.30
	10,688,086			10,688,086	

The following table summarizes information about underlying outstanding warrants for common stock of the Company outstanding and exercisable as of December 31, 2015:

	Warrants (Outstanding		Warrants Ex	kercisable
	Number of	Weighted	1	Number	Weighted
Range of Exercise	Shares	Average	Weighted Average Remaining Contractual	of Shares	Average
Prices	Underlying	g Exercise	Life (in years)	Underlying	Exercise
	Warrants	Price		Warrants	Price
\$12.50 - \$17.50	104,845	\$ 13.03	6.62	104,845	\$ 13.03
0.25 - 2.50	8,517,641	\$ 0.39	3.17	8,517,641	\$ 0.31
	8,622,486			8,622,486	

These common stock purchase warrants do not trade in an active securities market, and as such, we estimate the fair value of these warrants using the Black-Scholes Option Pricing Model using the following assumptions:

	2016	2015
Risk free interest rates	1.25% - 1.72%	0.77% - 1.73%
Expected life	5 years	5 years
Estimated volatility	422.9% - 667.5%	629.8% - 788.7%
Dividend yield	%	%

Expected volatility is based primarily on historical volatility. Historical volatility was computed using daily pricing observations for recent periods that correspond to the expected term of the warrants. We believe this method produces an estimate that is representative of our expectations of future volatility over the expected term of these warrants. We currently have no reason to believe future volatility over the expected term of these warrants is likely to differ materially from historical volatility. The expected term is based on the remaining term of the warrants. The risk-free interest rate is based on U.S. Treasury securities.

At December 31, 2016 and 2015 the aggregate intrinsic value of the warrants outstanding and exercisable was \$50,580 and \$5,295, respectively. The intrinsic value of a warrant is the amount by which the market value of the underlying warrant exercise price exceeds the market price of the stock at December 31 of each year. Note 11 – Stock Options

The following table summarizes stock option activity in the Company's stock-based compensation plans for the year ended December 31, 2016. All options issued were non-qualified stock options.

	Number of Options	Weighted Average Exercise Price	Aggr Intrin	sic	Number of Options Exercisable	Weighted Average Fair Value At Date of
						Grant
Outstanding at December 31, 2014	460,000	\$ 0.43	\$		174,167	\$ 0.59
Granted at Fair Value	1,000,000	\$ 0.27	\$	—	483,333	\$ 0.27
Exercised	_	\$ —	\$	—	_	\$ —
Forfeited	(30,000)	\$ 0.50	\$		(30,000)	\$ 0.50
Outstanding at December 31, 2015	1,430,000	\$ 0.31	\$		627,500	\$ 0.34
Granted at Fair Value		\$ —	\$		_	\$ —

Exercised	_	\$ —	\$ _	_	\$ —
Canceled	(375,000)	\$ 0.27	\$ 	(375,000) \$ 0.27
Outstanding at December 31, 2016	1,055,000		\$ 	252,500	

The intrinsic value of a stock option is the amount by which the market value of the underlying stock exceeds the (1) exercise price of the option at December 31, 2016. If the exercise price exceeds the market value, there is no intrinsic value.

During the year ended December 31, 2016, the Company did not grant employee stock options or stock options for consulting services.

The fair value of the stock option grants are amortized over the respective vesting period using the straight-line method and assuming no forfeitures and cancellations.

Compensation expense related to stock options included in general and administrative expense in the accompanying consolidated statements of operations for the years ended December 31, 2016 and December 31, 2015, was \$86,733, and \$169,210, respectively.

Stock options are granted at the fair market value of the Company's common stock on the date of grant. Options granted to officers and other employees vest immediately or over 36 months as provided in the option agreements at the date of grant.

The fair value of each option granted in 2016 and 2015 was estimated using the Black-Scholes Option Pricing Model. The following assumptions were used to compute the weighted average fair value of options granted during the periods presented.

	2016	2015	
Expected term of option	3 years	3 years	
Risk free interest rates	1.12	% 1.52	%
Estimated volatility	593.3	606.3	
Dividend yield		% —	%

The following table summarizes information about stock options outstanding at December 31, 2016:

Range of Exercise Prices	Number of Options	Weighted Average Remaining Contractual Life (Years)	Exercise	Aggreg Intrins Value	IC	Number Exercisable		Intri	regate nsic e (1)
\$0.27 - \$1.00	1,055,000	2.13	\$ 0.28	\$ -		1,055,000	\$ 0.28	\$	_

The intrinsic value of a stock option is the amount by which the market value of the underlying stock exceeds the (1) exercise price of the option at December 31, 2016. If the exercise price exceeds the market value, there is no intrinsic value.

The following table summarizes information about options outstanding at December 31, 2015:

Range of Exercise Prices	Options	Weighted Average Remaining Contractual Life (Years)	Exercise	Intrin	SIC	Number Exercisable		Aggre Intrina Value	sic
0.27 - 1.00	1,430,000	2.13	\$ 0.31	\$	_	241,667	\$ 0.34	\$	

A summary of the Company's non-vested stock options at December 31, 2016 and December 31, 2015 and changes during the years are presented below.

Non-Vested Stock Options	Options	Weighted Average Grant Date Fair Value
Non-Vested at December 31, 2015	1,188,333	\$ 0.28
Granted	_	\$ —
Vested	(813,333	\$ 0.27
Forfeited	(375,000)\$ 0.29
Non-Vested at December 31, 2016		\$ 0.28

Note 12 – Commitments and Contingencies

Leases

Rent expense for the years ended December 31, 2016 and 2015 was \$30,000 and \$29,250, respectively. The Company's office space is leased on a month-to-month basis, and therefore future annual minimum payments under non-cancellable operating leases are \$0 and \$0 for the years ending December 31, 2015 and 2016, respectively. Partnership Distributions

Under terms of the Second Amended Partnership Agreement of Aurora, Navitus earns a net profits interest respective to its 50% partnership interest. In addition, Navitus is entitled to a respective proportion of proceeds from the sale of Aurora assets. Any distributions of the net profits interest or asset sale proceeds to partners are at the discretion of Victory, as managing partner, together with 100% of the partnership interests. The accumulated net deficits of Navitus, along with historical contributions, net of paid distributions, are reported as non-controlling interests in the equity section of the consolidated balance sheets.

Under the terms of Aurora's Seconded Amended Partnership Agreement, Navitus Partners, LLC, the fourth partner of the Navitus Energy Group, admitted under the Navitus Private Placement Memorandum (the "Navitus PPM"), earns a preferred return distribution of 10% based upon capital contributions to Aurora used by Victory to acquire or develop oil and gas prospects or related enterprises on behalf of Aurora. The preferred return distribution is in addition to and does not reduce any net profits or asset sale proceeds interests distributions other than the 2% management fee receivable balance as of the year ended December 31, 2016 and any future management fees earned. The table below summarizes the net profit distributions, proceeds of asset sales and preferred return distributions earned by Navitus Energy Group during the years ended December 31, 2016 and 2015, respectively.

Novitus Energy Group Distribution Formed	Year Ended
Navitus Energy Group Distribution Earned	December 31,
	201@015
Aurora Net Profits Interests	\$-\$78,963
Proceeds from the Sale of Aurora Assets	

Preferred Distributions Due to Navitus Partners, LLC — 656,256 Total Distributions Earned By Navitus Energy Group \$—\$735,219

There were no net profit distributions, proceeds of asset sales and preferred return distributions paid to Navitus Energy Group during the years ended December 31, 2016 and 2015, respectively.

Navitus Partners, LLC, a partner in Navitus, also receives warrants for Victory's common stock, allocated as 50,000 warrants for every Unit purchased under the Navitus PPM (equivalent of one warrant for every \$1.00 invested), exercisable under the terms of Aurora's Second Amended Partnership Agreement and the Navitus PPM. Since August 23, 2012, \$7,332,900 of capital contributions have resulted in issuance of 9,444,900 common stock warrants (1,089,900 in 2012, 2,186,000 in 2013, 1,140,000 in 2014, 2,917,000 in 2015, and 2,112,000 in 2016). Litigation

Legal Cases Settled

Cause No. 08-04-07047-CV; Oz Gas Corporation v. Remuda Operating Company, et al. v. Victory Energy Corporation.; In the 112th District Court of Crockett County, Texas.

Plaintiff Oz Gas Corporation ("Oz") filed a lawsuit in April 2008 against various parties for bad faith trespass, among other claims, regarding the drilling of two wells on lands that Oz claims title to. On November 18, 2009, Victory Energy Corporation intervened in the lawsuit to protect its 50% interest in one of the named wells in the lawsuit (that being the 155-2 well located on the Adams Baggett Ranch in Crockett County, Texas).

This case was mediated, with no settlement reached. It went to trial February 8-9, 2012. The Court found in favor of Oz and rendered verdict against Victory and the other Defendants, jointly and severally. Victory appealed this case to the 8th Court of Appeals in El Paso, Texas where the Court of Appeals affirmed the verdict of the District Court and Victory filed a Motion for Rehearing, which was denied. Victory filed a Petition for Review in the Supreme Court of Texas on December 15, 2014 which was denied. Victory filed a Motion for Rehearing with the Supreme Court which was denied.

Oz filed Interrogatories and Request for Production in Aid of Judgment which have been answered by Victory. A Settlement and Forbearance Agreement was entered into on March 22, 2016 between the parties wherein no further post-judgment discovery or collection efforts will be made by Oz, for \$140,000 net of a \$14,000 payment received by the Oz receiver (see next following Cause No. C-1-CV-16-001610), with monthly payments of \$7,500 commencing April, 15, 2016. The remaining balance of \$65,000 as of December 31, 2016 is included in Accounts Payable on the Consolidated Balance Sheet.

Cause No. C-1-CV-16-001610; Oz Gas Corporation v. Victory Energy Corporation; In the County Court at Law No. 1 of Travis County, Texas.

Plaintiff Oz Gas Corporation ("Oz") filed an Application for Turnover Relief in Travis County, Texas on February 19, 2016. This order was granted and Thomas L. Kolker was appointed as Receiver to assist in the collection of non-exempt assets. Victory itself has not been placed into Receivership. Victory filed its Motion to Vacate the Turnover that was heard and denied by the trial court. Oz has since filed an Amended Application for Turnover Relief and Appointment of a Receiver to be heard March 10, 2016. Victory filed its Notice of Appeal March 4, 2016. A Settlement and Forbearance Agreement was entered into on March 22, 2016 as described above.

Cause No. D-1-GN-13-000044; Aurora Energy Partners and Victory Energy Corporation v. Crooked Oaks, LLC; In the 261st District Court of Travis County, Texas.

Victory Energy Corporation sued Crooked Oaks, LLC a/k/a Crooked Oak, LLC for breach of a purchase and sale agreement dated May 7, 2012 in which Victory sold certain assets to Crooked Oaks, LLC for \$400,000 of which only \$200,000 has been paid as of December 31, 2014. The lawsuit seeks to recover the remaining balance owed of \$200,000 from Crooked Oaks, LLC in addition to attorney's fees and all costs of court. Crooked Oaks, LLC has asserted a counterclaim for rescission of the underlying contract.

Victory and Crooked Oaks attended a mediation on February 10, 2016 where it was determined that Crooked Oaks was insolvent and since that date the case has been dismissed with prejudice.

Cause No. 50198; Trilogy Operating, Inc. v. Aurora Energy Partners; In the 118th Judicial District Court of Howard County, Texas.

This lawsuit was filed on January 9, 2015. This lawsuit alleges causes of action for declaratory judgment, breach of contract, and suit to quiet title regarding the drilling and completion of four wells. The parties entered into a

Settlement Agreement and Release on November 20, 2015 and an Agreed Order to Dismiss with Prejudice was granted on November 24, 2015.

Cause No. 50,916; Trilogy Operating Inc. v. Aurora Energy Partners; In the 118th Judicial District Court of Howard County, Texas.

This lawsuit was filed on January 6, 2016. This lawsuit alleges causes of action for a suit on a sworn account, breach of contract and a suit to foreclose on liens regarding the drilling and completion of seven wells. Aurora filed an answer on January 29, 2016. Trilogy filed a Motion for Partial Summary Judgment on March 23, 2016.

The parties entered into a Settlement Agreement and Release on April 26, 2016, effective April 1, 2016 to dismiss the lawsuit with prejudices. The Joint Motion to Dismiss with Prejudice was granted by the court May 2, 2016. In conjunction with the Joint Motion to Dismiss, Aurora assigned Trilogy all its interests in the seven wells and related oil and gas leases.

Cause No. 2015-05280; TELA Garwood Limited, LP. v. Aurora Energy Partners, Victory Energy Corporation, Kenneth Hill, David McCall, Robert Miranda, Robert Grenley, Ronald Zamber, and Patrick Barry; In the 164th District Court of Harris County, Texas.

This lawsuit was filed on January 30, 2015 and supplemented on March 4, 2015. This lawsuit alleges breach of contract regarding a Purchase and Sale Agreement that TELA Garwood Limited, LP and Aurora Energy Partners entered into on June 30, 2014. A first closing was held on June 30, 2014 and a purchase price adjustment payment was made on July 31, 2014. Between these two dates Aurora paid TELA approximately \$3,050,133. A second closing was to take place in September, however several title defects were found to exist. The title defects could not be cured and a purchase price reduction could not be agreed upon by the parties in relation to the title defects, therefore, the second closing never took place. Aurora and Victory filed an answer and counterclaim in this case. Both parties filed opposing motions for summary judgment which were heard on April 14, 2016. The Court granted Aurora's partial motions for summary judgment dismissing claims against Aurora/Victory's officers and directors, including Kenny Hill, David McCall, Robert Grenley, Ronald Zamber, Patrick Barry, and Fred Smith. The Court denied the remaining summary judgment issues of both parties. On June 2, 2016 Aurora/Victory filed a second Motion for Partial Summary Judgment on some discrete contract interpretation issues. The Court denied this motion on September 2, 2016. On December 9, 2016, Aurora/Victory and TELA entered into a Mutual Release and Settlement Agreement in which Aurora agreed to pay TELA \$320,000 (which is recorded in Accounts Payable as of December 31, 2016) and in turn each Party agreed to release the other Party from any matter relating to the PSA, the litigation or any claims that were or could have been brought in the litigation. In accordance with the Mutual Release and Settlement Agreement, Aurora made the full payment on February 1, 2017.

Cause No. 10-09-07213; Perry Howell, et al. v. Charles Gary Garlitz, et al.; In the 112th District Court of Crockett County, Texas.

The above referenced lawsuit was filed on or about September 6, 2010. This lawsuit alleges that Cambrian Management, Ltd. and Victory were trespassers on their land, and that they, along with other Defendants, drilled a well (115 #8) on land belonging to Plaintiffs. Plaintiffs claim trespass and unjust enrichment by certain Defendants because of the drilling of the 115 #8 well.

The Court placed this case on the Dismissal Docket asking any party to show cause as to why it should maintain this case on the docket on July 8, 2016. No party came forward stating why the case should be maintained and the Court entered and Order of Dismissal on August 9, 2016.

Legal Cases Pending

Cause No. CV-47,230; James Capital Energy, LLC and Victory Energy Corporation v. Jim Dial, et al.; In the 142nd District Court of Midland County, Texas.

This is a lawsuit filed on or about January 19, 2010 by James Capital Energy, LLC and Victory Energy Corporation against numerous parties for fraud, fraudulent inducement, negligent misrepresentation, breach of contract, breach of fiduciary duty, trespass, conversion and a few other related causes of action. This lawsuit stems from an investment Victory entered into for the purchase of six wells on the Adams Baggett Ranch with the right of first refusal on option acreage.

On December 9, 2010, Victory was granted an interlocutory Default Judgment against Defendants Jim Dial, 1st Texas Natural Gas Company, Inc., Universal Energy Resources, Inc., Grifco International, Inc., and Precision Drilling & Exploration, Inc. The total judgment amounted to approximately \$17,183,987.

Victory has added a few more parties to this lawsuit. Discovery is ongoing in this case and no trial date has been set at this time.

Victory believes they will be victorious against all the remaining Defendants in this case.

On October 20, 2011 Defendant Remuda filed a Motion to Consolidate and a Counterclaim against Victory. Remuda is seeking to consolidate this case with two other cases wherein Remuda is the named Defendant. An objection to this motion was filed and the cases have not been consolidated. Additionally, we do not believe that the counterclaim made by Remuda has any legal merit.

Note 13 – Related Party Transactions

During the years ended December 31, 2016 and 2015, we incurred a total of \$324,803 and \$411,059, respectively in legal fees with The McCall Firm. David McCall, our general counsel and a director, is a partner in The McCall Firm. The fees are attributable to litigation involving the Company's oil and natural gas operations in Texas. As of December 31, 2016 and 2015, the Company owed The McCall Firm approximately \$503,377, and \$371,826, respectively, for these professional services.

During the year ended December 31, 2016 temporary capital advances totaling \$130,000 were made by Navitus Energy Group Partnership and are recorded in Accrued Liabilities - related parties.

During the year ended December 31, 2015, a member of management made a\$29,553 temporary advance to the Company, a member of the board of directors made a \$15,000 temporary advance to the Company, and temporary capital advances totaling\$388,800 had been made by Navitus Energy Group Partnership. All the above amounts are recorded in Accrued Liabilities - related parties as of December 31, 2016 and 2015.

Note 14 - Subsequent Events

During the period of January 1, 2017 through and March 31, 2017, additional contributions of \$660,000 were received from Navitus, resulting in the issuance of an additional 560,000 common stock warrants for the purchase of shares of common stock of the Company.

On February 1, 2017, the Company entered into a securities purchase agreement (the "Securities Purchase Agreement") with Visionary Private Equity Group I, LP, a Missouri limited partnership (the "Investor"), pursuant to which the Investor agreed to purchase a unit comprised of (i) \$320,000 principal amount of 12% unsecured six-month promissory note with a maturity date of the earlier of six months from the date of the note or the date the Company consummates a material business combination transaction (the "Note"), and (ii) a common stock purchase warrant to purchase 5,203,252 shares of the Company's common stock, par value \$0.001 per shares (the "Common Stock") at an exercise price of \$0.0923 per share (the "Warrant" and together with the Note, the "Unit"). The sale by the Company to the Investor of the Unit, pursuant to the Securities Purchase Agreement is referred to herein as the "Private Placement." See Form 8-K filed on February 7, 2017.

On March 2, 2017, the Company extended the maturity date of the Credit Agreement to August 20, 2017. See Footnote 7 - Revolving Credit Agreement.

Supplementary Financial Information on Oil and Natural Gas Exploration, Development and Production Activities (Unaudited)

The following disclosures provide unaudited information required by ASC 932, "Extractive Activities – Oil and Gas" on oil and natural gas producing activities. These disclosures include non-controlling interests in Aurora which is managed and owned 50% by Victory.

Results of operations from oil and natural gas producing activities (Successful Efforts Method)

The Company's oil and natural gas properties are located within the United States. The Company currently has no operations in foreign jurisdictions. Results of operations from oil and natural gas producing activities are summarized below for the years ended December 31:

	Years Ended December		
	31,		
	2016	2015	
Revenues	\$287,179	\$650,648	
Costs incurred:			
Exploration and dry hole costs	3,000	2,513	

Lease operating costs and production taxes	119,935	192,504
Impairment of oil and natural gas reserves		867,048
Depletion, depreciation and accretion	125,744	637,121
Totals, costs incurred	248,679	1,699,186
Pre-tax income(loss) from producing activities	38,500	(1,048,538)
Results income (loss) from of oil and natural gas producing activities (excluding overhead,	\$38,500	\$(1,048,538)
income taxes, and interest costs)	,	1 () /

Costs incurred in oil and natural gas property acquisition, exploration and development activities are summarized below for the years ended December 31:

	Years End	led December
	2016	2015
Property acquisition and developmental costs:		
Development	\$18,442	\$1,058,704
Property Acquisition		_
Undrilled Leaseholds	_	_
Asset retirement obligations	_	2,506
Totals costs incurred	\$18,442	\$1,061,210

Oil and natural gas reserves

Proved reserves are estimated quantities of oil and natural gas which geological and engineering data demonstrate with reasonable certainty to be recoverable in future years from known reservoirs under existing economic and operating conditions. Proved developed reserves are proved reserves that can reasonably be expected to be recovered through existing wells with existing equipment and operating methods.

Proved oil and natural gas reserve quantities at December 31, 2016 and 2015 and the related discounted future net cash flows are based on estimates prepared by independent petroleum engineers. Such estimates have been prepared in accordance with guidelines established by the Securities and Exchange Commission.

Standardized measure of discounted future net cash flows relating to proven oil and gas reserves (SMOG) The following information has been prepared in accordance with the Financial Accounting Standards Board pronouncements and the regulations of the Securities and Exchange Commission, which require the standardized measure of discounted future cash flows based on sales prices, costs and statutory interest rates. The standardized measure of oil and gas producing activities is the present value of estimated future cash inflow from proved oil and natural gas reserves, less future development, abandonment, production and income tax expenses, discounted to reflect timing of future cash flows.

The Company's proved oil and natural gas reserves for the years ended December 31, 2016 and December 31, 2015 are shown below:

	Years Ended December 31,	
Volumes	2016	2015
Natural gas:	(Mcfs)	
Proved developed and undeveloped reserves (mcf):		
Beginning of year	178,750	600,000
Purchase (sale) of natural gas properties in place	(6,168)—
Discoveries and extensions		
Revisions	(30,794)	(410,362)

Production (30,038)(37,568) Proved reserves, at end of year (a) 111,750 178,750

> Years Ended December 31, 2016 2015

Oil: (Bbls)

Proved developed and undeveloped reserves:

Beginning of year 41,380 20,700
Purchase (sale) of oil producing properties in place (1,107)—
Discoveries and extensions - 30,720
Revisions (6,872)2,112
Production (6,871)(12,152)
Proved reserves, at end of year (a) 26,530 41,380

Includes 55,875 Mcf and 13,265 bbl and 89,375 Mcf and 20,690 bbl for the twelve months ended December 31, (a) 2016 and 2015, respectively of proved reserves attributable to a consolidated subsidiary in which there is a 50% non-controlling interest.

	Years Ended	
	December	r 31,
Values	2016	2015
Future cash inflows	1,272,950	2,345,940
Future costs:		
Production	(641,527)(964,520)
Development		(87,650)
Future cash flows	631,423	1,293,770
10% annual discount for estimated timing of cash flow	(159,123)(421,640)
Standardized measure of discounted cash flow (a)	472.300	872.130

(a) Includes 472,300 and 872,130 for the twelve months ended December 31, 2016 and 2015, respectively, of discounted cash flows attributable to a consolidated subsidiary in which there is a 50% non-controlling interest.

Using the SEC adjusted guidelines in place for 2016, the gas and oil prices for this analysis were set at the average price received on the "first-day-of-the-month" for 2016, for appropriate differentials. The "benchmark" prices are \$42.75 per barrel and \$2.49 per Mcf. The average quarterly price received for natural gas for 2015 ranged from \$1.72 /Mcf to \$3.95 /Mcf . The average quarterly price received oil for 2016 ranged from \$31.95/bbl to \$42.56/bbl.

Future income taxes are based on year-end statutory rates, adjusted for tax basis of oil and natural gas properties and availability of applicable tax assets, such as net operating losses. A discount factor of 10% was used to reflect the timing of future net cash flows.

The standardized measure of discounted future net cash flows is not intended to represent the replacement cost or fair market value of the Company's oil and natural gas properties. An estimate of fair value may also take into account, among other things, the recovery of reserves not presently classified as proved, anticipated future changes in prices and costs, and may require a discount factor more representative of the time value of money and the risks inherent in

reserve estimates.

Changes in standardized measure

Included within standardized measure are reserves purchased in place. The purchase of reserves in place includes undeveloped reserves which were acquired at minimal value that have been estimated by independent reserve engineers to be recoverable through existing wells utilizing equipment and operating methods available to the Company and that are expected to be developed in the near term based on an approved plan of development contingent on available capital.

Changes in the standardized measure of future net cash flows relating to proved oil and natural gas reserves for the years ended December 31 is summarized below:

	For the year December 2016	
Balance, January 1	872,130	964,000
Net change in price and production costs (a)	(220,100)	(235,300)
Sale of oil and gas, net	(173,400)	(455,600)
Extensions/Discoveries	12,500	_
Purchase of reserves in place	_	531,700
Sales of reserves in place	(15,500)	
Revisions of previous estimates	(153,900)	(803,900)
Accretion of discount	87,200	146,500
Net change in taxes	_	500,600
Net change in timing and other	63,400	224,300
Balance, December 31	472,300	872,130

⁽a) Includes 472,300 and 872,130 for the twelve months ended December 31, 2016 and 2015, respectively of future net cash flows attributable to a consolidated subsidiary in which there is a 50% non-controlling interest.

VICTORY ENERGY CORPORATION AND SUBSIDIARY UNAUDITED CONDENSED CONSOLIDATED FINANCIAL STATEMENTS THREE AND SIX MONTHS ENDED JUNE 30, 2017 AND 2016

VICTORY ENERGY CORPORATION AND SUBSIDIARY CONDENSED CONSOLIDATED BALANCE SHEETS

(Unaudited)	
ASSETS	
Current Assets	
Cash and cash equivalents \$47,157 \$56,456	
Accounts receivable 28,972 44,379	
Prepaid expenses 26,715 9,951	
Total current assets 102,844 110,786	
Fixed Assets	
Furniture and equipment 43,622 46,883	
Accumulated depreciation (40,211) (30,893)
Total furniture and fixtures, net 3,411 15,990	
Oil and gas properties, net of impairment (successful efforts method) 2,795,557 2,787,986	
Accumulated depletion, depreciation and amortization (2,214,709) (2,166,643)
Total oil and gas properties, net 580,848 621,343	
Other Assets	
Management fee receivable - affiliate 139,455 137,556	
Total Assets \$826,558 \$885,675	
LIABILITIES AND STOCKHOLDERS' DEFICIT	
Current Liabilities	
Accounts payable \$570,667 \$420,559	
Accrued liabilities 404,419 746,491	
Accrued liabilities - related parties 1,273,616 1,489,973	
Liability for unauthorized preferred stock issued 9,283 9,283	
Note payable (net of unamortized deferred financing costs) 254,500 564,263	
Note payable (net of debt discount) - affiliate 284,055 —	
Asset retirement obligation 52,321 76,850	
Total current liabilities 2,848,861 3,307,419	
Other Liabilities	
Asset retirement obligations 40,895 7,141	
Total long term liabilities 40,895 7,141	
Total Liabilities 2,889,756 3,314,560	
Stockholders' Equity (Deficit)	
Common stock, \$0.001 par value, 47,500,000 shares authorized, 31,220,326 shares and	
31,220,326 shares issued and outstanding for June 30, 2017 and December 31, 31,220 31,220	
2016, respectively	
Additional paid-in capital 36,134,513 35,795,479	
Accumulated deficit (47,393,299) (46,140,750)	
Total Victory Energy Corporation stockholders' deficit (11,227,56) (10,314,05)	1)
Non-controlling interest 9,164,368 7,885,166	

Total stockholders' equity (deficit)
Total Liabilities and Stockholders' Deficit

(2,063,198) (2,428,885) \$826,558 \$885,675

The accompanying notes are an integral part of these condensed consolidated financial statements. VICTORY ENERGY CORPORATION AND SUBSIDIARY CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS (Unaudited)

	For the Three Months Ended		d	For the Six Months Ended		
	June 30,			June 30,		
	2017	2016		2017		2016
Revenues						
Oil and gas sales	\$70,680	\$ 79,185		\$155,980		\$145,178
Gain on settlement and sale of oil and gas properties		64,824		_		64,824
Total revenues	70,680	144,009		155,980		210,002
Operating Expenses:						
Lease operating costs	31,836	25,774		57,740		63,127
Exploration and dry hole cost	315			2,218		_
Production taxes	3,806	3,782		8,491		7,055
General and administrative	513,826	623,995		1,117,488		1,068,156
Depreciation, depletion, amortization, and accretion	37,259	28,267		62,300		76,286
Total operating expenses	587,042	681,818		1,248,237		1,214,624
Loss from operations	(516,362)	(537,809)	(1,092,257)	(1,004,622)
Other Income (Expense):						
Management fee income	822	1,305		1,899		2,641
Interest expense	(98,633)	(33,124)	(187,991)	(66,313)
Total other income and expense	(97,811)	(31,819)	(186,092)	(63,672)
Loss before Tax Benefit	(614,173)	(569,628)	(1,278,349)	(1,068,294)
Tax benefit	_	_		_		
Net loss	(614,173)	\$ (569,628)	(1,278,349)	(1,068,294)
Less: Net loss attributable to non-controlling interest	(28,883)	(26,154)	(25,798)	(83,196)
Net loss attributable to Victory Energy Corporation	\$(585,290)	\$ (543,474)	\$(1,252,551)	\$(985,098)
Weighted average shares, basic and diluted	31,220,326	31,220,326		31,220,326		31,220,326
Net loss per share, basic and diluted	\$(0.02)	\$ (0.02)	\$(0.04)	\$(0.03)

The accompanying notes are an integral part of these condensed consolidated financial statements.

VICTORY ENERGY CORPORATION AND SUBSIDIARY CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOW (Unaudited)

For the Six Months Ended

June 30,

2017 2016

CASH FLOWS FROM OPERATING ACTIVITIES

Net loss \$(1,278,349) \$(1,068,294)

Adjustments to reconcile net loss to net cash used in operating activities			
Accretion of asset retirement obligations	1,655	1,597	
Amortization of debt discount	124,055		
Amortization of deferred financing costs	6,237	20,412	
Gain on settlement and sale of oil and gas properties		(64,824)
Depletion, depreciation, and amortization	60,645	74,689	-
Stock based compensation	179,034	57,409	
Change in operating assets and liabilities			
Accounts receivable	15,407	(20,274)
Management fee receivable - affiliate	(1,899) (2,641)
Prepaid expense	(16,764) 3,492	
Accounts payable	150,107	15,599	
Accrued liabilities - related parties	(216,357) 215,683	
Accrued liabilities	(357,789) (140,319)
Accrued interest note payable - affiliate	15,719		
Net cash used in operating activities	(1,318,299) (907,471)
CASH FLOWS FROM INVESTING ACTIVITIES:			
Lease purchases, drilling capital expenditures		(18,442)
Proceeds from sale of assets		8,294	
Net cash used in investing activities		(10,148)
CASH FLOWS FROM FINANCING ACTIVITIES:			
Non-controlling interest contributions	1,305,000	972,000	
Debt financing proceeds - affiliate	320,000		
Principal payments of debt financing	(316,000) —	
Net cash provided by financing activities	1,309,000	972,000	
Net Change in Cash and Cash Equivalents	(9,299) 54,381	
Beginning Cash and Cash Equivalents	56,456	2,384	
Ending Cash and Cash Equivalents	\$47,157	\$56,765	
Supplemental cash flow information:			
Cash paid for:			
Interest	\$18,362	\$22,412	
Non-cash investing and financing activities:			
Interest - accrued interest and amortization of debt discount	\$163,392	\$ —	
Accrued capital expenditures	\$173,568	\$305,983	
Revisions to asset retirement obligations	\$7,570	\$ —	

The accompanying notes are an integral part of these condensed consolidated financial statements.

Victory Energy Corporation and Subsidiary

Notes to the Condensed Consolidated Financial Statements

(Unaudited)

These consolidated financial statements have been prepared by Victory Energy Corporation ("Victory" or the "Company") without audit, pursuant to the rules and regulations of the Securities and Exchange Commission ("SEC"). They reflect all adjustments that are, in the opinion of management, necessary for a fair presentation of the results for the interim periods, on a basis consistent with the annual audited financial statements. All such adjustments are of a normal recurring nature. Certain information, accounting policies, and footnote disclosures normally included in financial statements prepared in accordance with accounting principles generally accepted in the United States ("GAAP") have been omitted pursuant to such rules and regulations, although the Company believes that the disclosures are adequate to make the information presented not misleading. This Quarterly Report on Form 10-Q should be read along with Victory's Annual Report on Form 10-K for the fiscal year ended December 31, 2016, which

contains a summary of the Company's significant accounting policies and other disclosures.

Note 1 – Organization and Summary of Significant Accounting Policies

Victory is an independent, growth-oriented oil and natural gas company engaged in the acquisition, exploration, development, and production of domestic oil and natural gas properties, through its partnership with Aurora Energy Partners ("Aurora"). In this report, "the Company", "we" and "our" refers to the consolidated accounts and presentation of Victory and Aurora, with the equity of non-controlling interests stated separately. Current operations are primarily located onshore in Texas and New Mexico. The Company was organized under the laws of the State of Nevada on January 7, 1982. The Company is authorized to issue 47,500,000 shares of \$0.001 par value common stock, and has 31,220,326 shares of common stock outstanding as of June 30, 2017. Our corporate headquarters are located at 3355 Bee Caves Rd. Ste. 608, Austin, Texas.

A summary of significant accounting policies followed in the preparation of the accompanying consolidated financial statements is set forth below.

Basis of Presentation and Consolidation:

Victory is the managing partner of Aurora, and holds a fifty percent (50%) partnership interest in Aurora. Aurora, a subsidiary of the Company, is consolidated with Victory for financial statement reporting purposes, as the terms of the partnership agreement that govern the operations of Aurora give Victory effective control of the partnership. The consolidated financial statements include the accounts of Victory and the accounts of Aurora. The Company's management, in considering accounting policies pertaining to consolidation, has reviewed the relevant accounting literature. The Company follows the relevant accounting literature in assessing whether the rights of the non-controlling interests should overcome the presumption of consolidation when a majority voting or controlling interest in its investee "is a matter of judgment that depends on facts and circumstances". In applying the circumstances and contractual provisions of the partnership agreement, management determined that the non-controlling rights do not, individually or in the aggregate, provide for the non-controlling interest to "effectively participate in significant decisions that would be expected to be made in the ordinary course of business." The rights of the non-controlling interest are protective in nature. All intercompany balances have been eliminated in consolidation. Certain reclassifications of prior year balances have been made to confirm such amounts to current year classifications. The reclassifications have no prior impact on net income.

Non-controlling Interests:

The Navitus Energy Group ("Navitus"), a Texas general partnership, is a partner with Victory in Aurora. The two partners each own a fifty percent (50%) interest in Aurora. Victory is the Managing partner and has contractual authority to manage the business affairs of Aurora. Navitus currently has four partners. They are James Capital Consulting, LLC ("JCC"), James Capital Energy, LLC ("JCE"), Rodinia Partners, LLC and Navitus Partners, LLC. Although this partnership has been in place since January 2008, its members and other elements have changed since that time.

The non-controlling interest in Aurora is held by Navitus. As of June 30, 2017, \$9,164,368 was recorded as the equity of the non-controlling interest in our consolidated balance sheets representing a third-party investment in Aurora, with net loss attributable to non-controlling interest of \$28,883 and \$26,154 for the three months ended June 30, 2017 and 2016, respectively, and \$25,798 and \$83,196 for the six months ended June 30, 2017 and 2016, respectively. As of December 31, 2016, \$7,885,166 was recorded as the equity of the non-controlling interest in our consolidated balance sheets representing a third-party investment in Aurora.

Use of Estimates:

The preparation of our consolidated financial statements in conformity with GAAP requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities, the disclosure of contingent assets and liabilities at the date of the consolidated financial statements and the reported amounts of revenues and expenses during the reporting period. Actual results could differ from those estimates. Estimates are used primarily when accounting for depreciation, depletion, and amortization ("DD&A") expense, property costs, estimated future net cash flows from proved reserves, assumptions related to abandonments and impairments of oil and natural gas properties, taxes, accruals of capitalized costs, operating costs and production revenue, general and administrative costs and interest, purchase price allocation on properties acquired, various common stock, warrants and option transactions, and loss contingencies.

Oil and Natural Gas Properties:

We account for investments in oil and natural gas properties using the successful efforts method of accounting. Under this method of accounting, only successful exploration drilling costs that directly result in the discovery of proved reserves are capitalized. Unsuccessful exploration drilling costs that do not result in an asset with future economic benefit are expensed. All development costs are capitalized because the purpose of development activities is considered to be building a producing system of wells, and related equipment facilities, rather than searching for oil and natural gas. Items charged to expense generally include geological and geophysical costs. Capitalized costs for producing wells and leasehold costs of proved properties are amortized on a unit-of-production basis over the remaining life of proved developed and total proved reserves, respectively.

We review our proved oil and gas properties for impairment whenever events and circumstances indicate that a decline in the recoverability of their carrying value may have occurred. We estimate the expected undiscounted future cash flows of our oil and gas properties and compare such undiscounted future cash flows to the carrying amount of the oil and gas properties to determine if the carrying amount is recoverable. If the carrying amount exceeds the estimated undiscounted future cash flows, we will adjust the carrying amount of the oil and gas properties to fair value. The factors used to determine fair value are subject to our judgment and expertise and include, but are not limited to, recent sales prices of comparable properties, the present value of future cash flows, net estimated operating and development costs using estimates of proved reserves, future commodity pricing, future production estimates, anticipated capital expenditures, and various discount rates commensurate with the risk and current market conditions associated with realizing the expected cash flows projected. Because of the uncertainty inherent in these factors, we cannot predict when or if future impairment charges for proved properties will be recorded.

The assessment of unproved properties to determine any possible impairment requires significant judgment. We assess our unproved properties to determine any possible impairment on a property-by-property basis based on remaining lease terms, drilling results or future plans to develop acreage. Due to the uncertainty inherent in these factors, we cannot predict the amount of impairment charges that may be recorded in the future.

The Company recorded no impairment expense for the six months ended June 30, 2017 and 2016, respectively based on the analysis above.

Asset Retirement Obligations:

The Company records the estimate of the fair value of liabilities related to future asset retirement obligations ("ARO") in the period the obligation is incurred. Asset retirement obligations relate to the removal of facilities and tangible equipment at the end of an oil and natural gas property's useful life. The application of this rule requires the use of management's estimates with respect to future abandonment costs, inflation, market risk premiums, useful life and cost of capital and required government regulations. GAAP requires that the estimate of our ARO does not give consideration to the value the related assets could have to other parties.

Other Property and Equipment:

Our office equipment in Austin, Texas is being depreciated on the straight-line method over the estimated useful life of three to seven years.

Cash and Cash Equivalents:

The Company considers all liquid investments with original maturities of three months or less from the date of purchase that are readily convertible into cash to be cash equivalents. The Company had no cash equivalents at June 30, 2017 and December 31, 2016.

Accounts Receivable:

Our accounts receivable are primarily from purchasers of natural gas and oil and exploration and production companies which own an interest in properties we operate.

Fair Value:

At June 30, 2017 and December 31, 2016, the carrying value of the Company's financial instruments such as prepaid expenses and payables approximated their fair values based on the short-term maturities of these instruments. The carrying value of other liabilities approximated their fair values because the underlying interest rates approximated market rates at the balance sheet dates. Management believes that due to the Company's current credit worthiness, the fair value of debt could be less than the book value; however, due to current market conditions and available information, the fair value of such debt is not readily determinable. Financial Accounting Standard Board ("FASB") Accounting Standards Codification ("ASC") Topic 820, Fair Value Measurements and Disclosures, established a hierarchical disclosure framework associated with the level of pricing observability utilized in measuring fair value.

This framework defined three levels of inputs to the fair value measurement process and requires that each fair value measurement be assigned to a level corresponding to the lowest level input that is significant to the fair value measurement in its entirety. The three broad levels of inputs defined by FASB ASC Topic 820 hierarchy are as follows:

Level 1 - quoted prices (unadjusted) in active markets for identical assets or liabilities that the reporting entity has the ability to access at the measurement date;

Level 2 - inputs other than quoted prices included within Level 1 that are observable for the asset or liability, either directly or indirectly. If the asset or liability has a specified (contractual) term, a Level 2 input must be observable for substantially the full term of the asset or liability; and

Level 3 - unobservable inputs for the asset or liability. These unobservable inputs reflect the entity's own assumptions about the assumptions that market participants would use in pricing the asset or liability and are developed based on the best information available in the circumstances (which might include the reporting entity's own data).

The initial measurement of ARO is calculated using discounted cash flow techniques and based on internal estimates of future ARO costs associated with proved oil and gas properties. Inputs used in the calculation of ARO include plugging costs and reserve lives, which are considered Level 3 inputs. A reconciliation of Victory's ARO is presented in Note 4.

Unamortized Discount:

Unamortized discount consists of value attributed to free standing equity instruments issued to the holders of affiliate note payable (see Note 6) and are amortized over the life of the related loans using a method consistent with the interest method. Amortization of debt discount totaled \$124,055 for the six months ended June 30, 2017 and is included in interest expense in the condensed consolidated statements of operations. The following table shows the discount and related accumulated amortization as of June 30, 2017 and December 31, 2016:

Original issuance discount	June 30, 2017 \$160,000		cember 2016 —
Accumulated amortization	(124,055)	_	
Unamortized discount, net	\$35,945	\$	

Revenue Recognition:

The Company uses the sales method of accounting for oil and natural gas revenues. Under this method, revenues are recognized based on actual volumes of gas and oil sold to purchasers. The volumes sold may differ from the volumes to which the Company is entitled based on our interests in the properties. Differences between volumes sold and entitled volumes create oil and natural gas imbalances which are generally reflected as adjustments to reported proved oil and natural gas reserves and future cash flows in their supplemental oil and natural gas disclosures. If their excess takes of natural gas or oil exceed their estimated remaining proved reserves for a property, a natural gas or oil imbalance liability is recorded in the Consolidated Balance Sheets.

Concentrations:

There is a ready market for the sale of crude oil and natural gas. During 2017 and 2016, our gas field and our producing wells sold their respective gas and oil production to one purchaser for each field or well. However, because alternate purchasers of oil and natural gas are readily available at similar prices, we believe that the loss of any of our purchasers would not have a material adverse effect on our financial results. A majority of the Company's production and reserves are from the Eagle Ford property in South Texas and the Permian Basin of West Texas.

Earnings (Losses) per Share:

Basic earnings per share ("EPS") is computed by dividing net income (loss) attributable to controlling interests by the weighted-average number of shares of common stock outstanding during the period. Diluted EPS takes into account the dilutive effect of potential common stock that could be issued by the Company in conjunction with stock awards that have been granted to directors and employees. In accordance with FASB ASC 260, Earnings per Share, awards of

unvested shares shall be considered outstanding as of the respective grant dates for purposes of computing diluted EPS even though their exercise is contingent upon vesting. Given the historical and projected future losses of the Company, all potentially dilutive common stock equivalents are considered anti-dilutive.

Income Taxes:

The Company accounts for income taxes in accordance with FASB ASC 740, Income Taxes, which requires an asset and liability approach for financial accounting and reporting of income taxes. Deferred income taxes reflect the impact of temporary differences between the amount of assets and liabilities for financial reporting purposes and such amounts as measured by tax laws and regulations. Deferred tax assets include tax loss and credit carry forwards and are reduced by a valuation allowance if, based on available evidence, it is more likely than not that some portion or all of the deferred tax assets will not be realized.

The realization of future tax benefits is dependent on our ability to generate taxable income within the carry forward period. Given the Company's history of net operating losses, management has determined that it is likely that the Company will not be able to realize the tax benefit of the carry forwards. ASC 740 requires that a valuation allowance be established when it is more likely than not that all or a portion of deferred tax assets will not be realized. Accordingly, the Company has a full valuation allowance against its net deferred tax assets at June 30, 2017 and December 31, 2016. Upon the attainment of taxable income by the Company, management will assess the likelihood of realizing the deferred tax benefit associated with the use of the net operating loss carry forwards and will recognize a deferred tax asset at that time.

Stock-Based Compensation:

The Company applies FASB ASC 718, Compensation-Stock Compensation, to account for the issuance of options and warrants to employees, key partners, directors, officers and Navitus investors. The standard requires all share-based payments, including employee stock options, warrants and restricted stock, be measured at the fair value of the award and expensed over the requisite service period (generally the vesting period). The fair value of options and warrants granted to employees, directors and officers is estimated at the date of grant using the Black-Scholes option pricing model by using the historical volatility of the Company's stock price. The calculation also takes into account the common stock fair market value at the grant date, the exercise price, the expected term of the common stock option or warrant, the dividend yield and the risk-free interest rate.

The Company from time to time may issue stock options, warrants and restricted stock to acquire goods or services from third-parties. Restricted stock, options or warrants issued to third parties are recorded on the basis of their fair value, which is measured as of the date issued. The options or warrants are valued using the Black-Scholes option pricing model on the basis of the market price of the underlying equity instrument on the "valuation date," which for options and warrants related to contracts that have substantial disincentives to non-performance, is the date of the contract, and for all other contracts is the vesting date. Expense related to the options and warrants is recognized on a straight-line basis over the shorter of the period over which services are to be received or the vesting period and is included in general and administrative expenses in the accompanying consolidated statements of operations.

The Company recognized stock-based compensation expense from stock awards, warrants, and stock options granted to directors, officers, and employees for services of \$8,337 and \$33,994 for the three months ended June 30, 2017 and 2016, respectively and \$179,034 and \$57,409 for the six months ended June 30, 2017 and 2016, respectively.

Recently Adopted Accounting Standards:

In January 2017, FASB issued Accounting Standards Update ("ASU") 2017-01, Business Combinations (Topic 805): Clarifying the Definition of a Business, which changes the definition of a business to assist entities with evaluating when a set of transferred assets and activities is deemed to be a business. Determining whether a transferred set constitutes a business is important because the accounting for a business combination differs from that of an asset acquisition. The definition of a business also affects the accounting for dispositions. Under ASU 2017-01, when substantially all of the fair value of assets acquired is concentrated in a single asset, or a group of similar assets, the assets acquired would not represent a business and business combination accounting would not be required. ASU 2017-01 may result in more transactions being accounted for as asset acquisitions rather than business combinations. ASU 2017-01 is effective for interim and annual periods beginning after December 15, 2017 and shall be applied prospectively. Early adoption is permitted. The Company adopted ASU 2017-01 on January 1, 2017 and will apply the new guidance to applicable transactions going forward.

In March 2016, FASB issued guidance regarding the simplification of employee share-based payment transactions, including the income tax consequences, classification of awards as either equity or liabilities, and classification on the statement of cash flows. We adopted this guidance in the second quarter of 2016 as permitted by the guidance. Adoption of this guidance did not impact our financial statements, except for the simplification in accounting for income taxes using a modified retrospective approach. Upon adoption, we recorded a related deferred tax asset for previously unrecognized excess tax benefits of \$37 million. As we consider it more likely than not that the deferred tax asset will not be realized, we recorded a full valuation allowance of \$37 million, resulting in no net effect on our consolidated statement of operations. We elected to continue our current policy of estimating forfeitures. In April 2015, FASB issued ASU 2015-03, Interest—Imputation of Interest (Subtopic 835-30): Simplifying the Presentation of Debt Issuance Costs, Entities that have historically presented debt issuance costs as an asset, related to a recognized debt liability, will be required to present those costs as a direct deduction from the carrying amount of that debt liability. ASU 2015-03 does not change the recognition, measurement, or subsequent measurement guidance for debt issuance costs. In August 2015, FASB issued ASU 2015-15, Interest—Imputation of Interest (Subtopic 835-30), which addresses the presentation or subsequent measurement of debt issuance costs related to line-of-credit arrangements, given the absence of authoritative guidance within ASU 2015-03 for debt issuance costs related to line-of-credit arrangements. The amendments are effective for interim and annual reporting periods beginning after December 15, 2015. Therefore, the Company adopted ASU 2015-03 beginning January 1, 2016. Changes to the balance sheet have been applied on a retrospective basis. This resulted in the reclassification of debt issuance costs of \$6,237 and \$40,823 associated with our Credit Agreement from Other Assets to Current Note Payable in the Consolidated Balance Sheet as of the six months ended June 30, 2017 and the year ended December 31, 2016. In February 2015, FASB issued ASU 2015-02, Consolidation (Topic 810): Amendments to the Consolidated Analysis. ASU 2015-02 amended the consolidation guidance by modifying the evaluation criteria for whether limited partnerships and similar legal entities are variable interest entities, eliminating the presumption that a general partner should consolidate a limited partnership, and affecting the consolidated analysis of reporting entities that are involved with variable interest entities. The adoption of ASU 2015-02, effective January 1, 2016, did not have a material impact on our consolidated balance sheets, statements of operations or statements of cash flows. Recently Issued Accounting Standards:

In February 2016, the FASB issued guidance regarding the accounting for leases. The guidance requires recognition of most leases on the balance sheet. The guidance requires lessees and lessors to recognize and measure leases at the beginning of the earliest period presented using a modified retrospective approach. The guidance is effective for interim and annual periods beginning after December 15, 2018. We are currently evaluating the impact of this guidance on our consolidated financial statements.

In January 2016, the FASB issued guidance regarding several broad topics related to the recognition and measurement of financial assets and liabilities. The guidance is effective for interim and annual periods beginning after December 15, 2017. We do not expect this guidance to have a material impact on our consolidated financial statements. In May 2014, the FASB issued guidance regarding the accounting for revenue from contracts with customers. In April 2016, May 2016 and December 2016, FASB issued additional guidance, addressed implementation issues and provided technical corrections. The guidance may be applied retrospectively or using a modified retrospective approach to adjust retained earnings (deficit). The guidance is effective for interim and annual periods beginning after December 15, 2017. We are currently evaluating the impact of this guidance on our consolidated financial statements. Going Concern:

The accompanying consolidated financial statements have been prepared assuming the Company will continue as a going concern, which contemplates the realization of assets and satisfaction of liabilities in the normal course of business. As presented in the consolidated financial statements, the Company has incurred a net loss of \$585,290 and \$543,474 for the three months ended June 30, 2017 and 2016, respectively, and net losses of \$1,252,551 and \$985,098 for the six months ended June 30, 2017 and 2016, respectively. The cash proceeds from new contributions to the Aurora partnership by Navitus, and loans from affiliates have allowed the Company to continue operations and invest in new oil and natural gas properties. Management anticipates that operating losses will continue in the near term until new wells are drilled, successfully completed and incremental production increases revenue. The Company has invested \$0 and \$18,442, respectively, in leases, and drilling and

completion costs, for the six months ended June 30, 2017 and 2016, respectively.

The Company remains in active discussions with Navitus and others related to longer term financing required for our capital expenditures planned for 2017. Without additional outside investment from the sale of equity securities and/or debt financing, our capital expenditures and overhead expenses must be reduced to a level commensurate with available cash flows.

The accompanying consolidated financial statements are prepared as if the Company will continue as a going concern. The consolidated financial statements do not contain adjustments, including adjustments to recorded assets and liabilities, which might be necessary if the Company were unable to continue as a going concern. Note 2 - Acquisitions

During February 2015, Victory entered into a letter of intent ("LOI") and subsequently into (a) the Pre-Merger Collaboration Agreement (the "Collaboration Agreement") with Lucas Energy Inc. ("Lucas"), Navitus and AEP Assets, LLC ("AEP"), a wholly-owned subsidiary of Aurora; and (b) the Pre-Merger Loan and Funding Agreement (the "Loan Agreement") with Lucas. During March 2015 the parties entered into Amendment No. 1 to the Pre-Merger Collaboration which amendments affected thereby are included in the discussion of the Collaboration Agreement below. Payments of \$195,928 and \$317,027 were made by Aurora, on behalf of Victory, to Earthstone Energy/Oak Valley Resources and Penn Virginia, respectively, for costs related to the two Earthstone Energy/Oak Valley Resources and the five Penn Virginia operated Eagle Ford wells, respectively.

The initial draw, and additional amounts borrowed by Lucas under the Loan Agreement were evidenced by a Secured Subordinated Delayed Draw Term Note issued by Lucas in favor of Victory, which was in an initial amount of \$250,000 (the "Draw Note"). Borrowings evidenced by the Draw Note accrued interest at one-half of one percent (0.5%) per annum, with accrued interest payable in one lump sum on maturity. The maturity date of the Draw Note was February 26, 2015. A total of \$600,000 was paid to Lucas under the Draw Note.

Subsequent to March 31, 2015, the Company terminated the LOI and notified Lucas pursuant to the Loan Agreement, that it would not extend any further credit to Lucas under the Loan Agreement. There were \$0 associated costs incurred during the six-month periods ended June 30, 2017 and 2016.

Further, the Company entered into: (1) a Settlement Agreement and Mutual Release (the "Lucas Settlement Agreement") with Lucas; (2) a Settlement Agreement and Mutual Release (the "Rogers Settlement Agreement") with Louise H. Rogers, ("Rogers"), and; (3) a Compromise Settlement Agreement and Mutual General Release, effective as of September 25, 2015 (the "Earthstone Settlement Agreement", and, together with the Lucas Settlement Agreement and the Rogers Settlement Agreement, the "Settlement Agreements") with Earthstone Operating, LLC, Earthstone Energy, Inc., Oak Valley Resources, LLC, Oak Valley Operating LLC and Sabine River Energy, LLC (collectively, "Earthstone"), Lucas, AEP, and Aurora.

Lucas Settlement Agreement

The Company and Lucas agreed to terminate any and all obligations between the parties arising under the LOI and the Collaboration Agreement. The Company and Lucas further agreed that the Company would retain ownership and control over five Penn Virginia well-bores previously assigned by Lucas to the Company (the "Penn Virginia Well-Bores"), as well as the obligations to pay the expenses associated with such Penn Virginia Well-Bores effective after August 1, 2014. Under the terms of the Lucas Settlement Agreement, Lucas agreed to assign to the Company all of Lucas' rights in a certain oil and gas property located in the same field as the Penn Virginia Well-Bores (the "Additional Penn Virginia Property"), including the rights to all revenues from all wells on some properties. Rogers Settlement and Amended Rogers Settlement Agreements

The Company and Rogers agreed, among other things: (i) to terminate the contingent promissory note in the principal amount of \$250,000 payable to Rogers that was issued by Victory in connection with the entry by Lucas and the Company into the Collaboration Agreement; (ii) that the Company would pay Rogers, on or before July 15, 2015, \$258,125; and (iii) that Rogers' legal counsel will hold the assignment of the Additional Penn Virginia Property and the Settlement Shares in escrow until such time as the payment of \$258,125 is made by the Company to the Rogers. Failure of the Company to make the payment of \$258,125 on or before July 15, 2015, would result in the Company being in default under the Rogers Settlement Agreement and default interest on the amount due would begin to accrue at a per diem rate of \$129.0625. Additionally, the Company acknowledged in the Amendment its obligation to pay Rogers' attorney's fees in the amount of \$25,000. The Company has not made any payments to Rogers pursuant to the Rogers Settlement Agreement and as a result the additional Penn Virginia Property was returned to Lucas in September 2015. The full amount due under the Roger's obligation including accrued interest at June 30, 2017

totals \$348,788 and is included in accrued liabilities on the consolidated balance sheet. Note 3 – Oil and natural gas properties, net of accumulated impairment (under successful efforts accounting) Oil and natural gas properties are comprised of the following:

	June 30, 2017	December
	Julie 30, 2017	31, 2016
Proved property	\$9,702,938	\$9,695,367
Unproved property	\$1,375,940	\$1,375,940
Total oil and natural gas properties, at cost	\$11,078,878	\$11,071,307
Less: accumulated impairment	\$(8,283,321)	\$(8,283,321)
Oil and natural gas properties, net of impairment	\$2,795,557	\$2,787,986
Less: accumulated depletion	(2,214,709)	(2,166,643)
Oil and natural gas properties, net	\$580,848	\$621,343

Depletion and accretion expense for the three months ended June 30, 2017 and 2016 was \$26,295 and \$26,651, respectively, and \$49,720 and \$73,054 for the six months ended June 30, 2017 and 2016, respectively. During the three and six months ended June 30, 2017 and 2016, the Company recorded no impairment losses.

Note 4 – Asset Retirement Obligations

The following table is a reconciliation of the ARO liability as of and for the six months ended June 30, 2017 and the twelve months ended December 31, 2016.

	June 30, 2017	December	
	June 30, 2017	31, 2016	
Asset retirement obligation at beginning of period	\$ 83,991	\$109,171	
Liabilities incurred on properties acquired and developed		_	
Revisions to previous estimates	7,570	_	
Liabilities on properties sold or settled		(27,850)
Accretion expense	1,655	2,670	
Asset retirement obligation at end of period	\$ 93,216	\$83,991	

Note 5 – Revolving Credit Agreement

On February 20, 2014, Aurora, as borrower, entered into a credit agreement (the "Credit Agreement") with Texas Capital Bank ("the Lender"). Guarantors on the Credit Agreement are Victory and Navitus, the two partners of Aurora. Pursuant to the Credit Agreement, the Lender agreed to extend credit to Aurora in the form of: (a) one or more revolving credit loans (each such loan, a "Loan"); and (b) the issuance of standby letters of credit, of up to an aggregate principal amount at any one time not to exceed the lesser of: (i) \$25,000,000; or (ii) the borrowing base in effect from time to time (the "Commitment"). The initial borrowing base on February 20, 2014 was set at \$1,450,000. The borrowing base is determined by the Lender, in its sole discretion, based on customary lending practices, review of the oil and natural gas properties included in the borrowing base, financial review of Aurora, the Company and Navitus and such other factors as may be deemed relevant by the Lender. The borrowing base is re-determined: (i) on or about June 30 of each year based on the previous December 31 reserve report prepared by an independent reserve engineer; and (ii) on or about August 31 of each year based on the previous June 30 reserve report prepared by Aurora's internal reserve engineers or an independent reserve engineer and certified by an officer of Aurora. The Credit Agreement will mature on February 20, 2017. Amounts borrowed under the Credit Agreement will bear interest at rates equal to the lesser of: (i) the maximum rate of interest which may be charged or received by the Lender in accordance with applicable Texas law; and (ii) the interest rate per annum publicly announced from time to time by the Lender as the prime rate in effect at its principal office plus the applicable margin. The applicable margin is: (i) with respect to Loans, one percent (1.00%) per annum; (ii) with respect to letter of credit fees, two percent (2.00%) per annum; and (iii) with respect to commitment fees, one-half of one percent (0.50%) per annum. Loans made under the Credit

Agreement are secured by: (i) a first priority lien in the oil and gas properties of Aurora, the Company and Navitus; and (ii) a first priority security interest in substantially all of the assets of Aurora and its subsidiaries, if any, as well as in all (100%) of the partnership interests in Aurora held by the Company and Navitus. Loans made under the Credit Agreement to Aurora are fully guaranteed by the Company and Navitus.

The Credit Agreement contains various affirmative and negative covenants. These covenants, among other things, limit additional indebtedness, additional liens and transactions with affiliates. Among the covenants contained in the Credit Agreement are financial covenants that Aurora will maintain a minimum earnings before interest, taxes, depreciation, depletion, amortization, and exploration expenses ("EBITDAX") to Cash Interest Ratio of 3.5 to 1.0 and a minimum Current Ratio of not less than 1.0 to 1.0. The Current Ratio is defined under the covenants to include, as a current asset, the revolving credit availability.

On April 13, 2015, the Company received the annual Borrowing Base Adjustment called for under the terms of the Credit Agreement, which called for a decrease in the borrowing base of \$300,000 payable by May 13, 2015, and an increase in the monthly reduction amount to \$10,000 commencing as of June 1, 2015. Additionally, the Lender notified Aurora that, based on the Lender's redetermination of Aurora's borrowing base, the monthly reduction amount under the Credit Agreement will be increased, commencing on June 1, 2015, from \$0 to \$10,000. Pursuant to this increase in the monthly reduction amount, Aurora's borrowing base will be automatically reduced by \$10,000 on the first day of each calendar month beginning in June 2015 until the Lender's next periodic borrowing base redetermination. The Company made one payment in the amount of \$10,000 in June 2015.

On May 13, 2015, Aurora informed the Lender it would not make the required \$300,000 payment but was submitting the newly acquired five Eagle Ford wells as additional collateral to be considered and its willingness to execute mortgages regarding the properties to meet the Deficiency.

On August 21, 2015, the Company executed a Forbearance Agreement whereby the Lender would forbear all existing events of default which includes all payments under the previously mentioned Borrowing Base Deficiency payments not yet paid under the April 13, 2015 Redetermination Date notification, as well as the late interest payments for June, July and August 2015, violations of Aurora financial covenants for the three months ended March 31, 2015, and June 30, 2015, and default notice for the late filing of March 31, 2015 financial reports. On August 26, 2015, the Company paid the Lender \$76,081 to cover a portion of the deficiency payment, as well as a Forbearance document fee and Lender's legal expenses, as required by the Forbearance Agreement, and the aforementioned Forbearance Agreement went into effect for the \$260,000 remaining borrowing base deficiency payment. On August 31, 2015, the Forbearance Agreement terminated pursuant to its terms. The Company did not make the above payment and has been in continuous contact with its lender regarding its plan of payment of the \$260,000 as well as the remaining credit facility balance. The Company made a \$50,000 principle payment to the Lender on October 14, 2015 as part of that plan.

On December 5, 2016, the Company entered into a new Forbearance Agreement to the Credit Agreement. Pursuant to the Forbearance Agreement, the Lender agreed to forbear from exercising any of its rights and remedies under the Credit Agreement until February 20, 2017 with respect to the historical events of default.

The Forbearance Period was amended and extended on March 2, 2017 and will end on the first to occur of the following: (i) the expiration of the amended Forbearance Period on August 20, 2017; (ii) a breach by Aurora or any Guarantor of any of the conditions, covenants, representations and/or warranties set forth in the Forbearance Agreement; (iii) the occurrence of any new event of default under the Credit Agreement; (iv) the occurrence or threat of the occurrence of any enforcement action against Aurora or any Guarantor by any of their creditors which, in Lender's reasonable judgment, would materially interfere with the operation of Aurora's or the Guarantor's business or the Lender's ability to collect on the obligations due under the Credit Agreement; (v) the institution of any bankruptcy proceeding relating to Aurora or any Guarantor; or (vi) the initiation by Aurora or any Guarantor of any judicial, administrative or arbitration proceedings against the Lender. The Lender's agreement to forbear from exercising its rights and remedies as a result of the Existing Events of Default is subject to and conditioned upon the following: (i) the payment by Aurora to the Lender of at least \$20,000 on or before the last business day of each calendar week occurring hereafter; and (ii) the delivery by Aurora of such other documents, instruments and certificates as reasonably requested by Lender. The foregoing description of the Forbearance Agreement is a summary only and is qualified in its entirety by reference to the complete text of the Forbearance Agreement. Since the execution of the extended Forbearance Agreement, the Company has paid the Lender \$316,000. The balance owed on the Credit

Agreement was \$254,500 and \$564,263 as of June 30, 2017 and December 31, 2016, respectively.

As of June 30, 2017 and 2016, the Company was out of compliance with the Current Ratio and with the EBITDAX to Cash Interest Ratio due to its reduced revenue streams from price and production declines and continued high general and administrative expenses. Therefore, the Company is in technical default of the Credit Agreement and related agreements. The Company has not yet been advised by the Lender of any additional actions the Lender plans to take. Amortization of debt financing costs on this debt was \$6,237 and \$20,412 for the six months ended June 30, 2017 and 2016, respectively. Interest expense was \$18,308 and \$22,412 for the six months ended June 30, 2017, and 2016, respectively.

Note 6 – Related Party Transactions

David McCall, our general counsel and a director, is a partner in The McCall Firm. Fees related to his services are attributable to litigation involving the Company's oil and natural gas operations in Texas. As of June 30, 2017 and December 31, 2016, the Company owed The McCall Firm \$527,098, and \$503,377, respectively.

On February 1, 2017, the Company entered into a securities purchase agreement (the "Securities Purchase Agreement") with Visionary Private Equity Group I, LP, a Missouri limited partnership (the "Investor"), pursuant to which the Investor agreed to purchase a unit comprised of: (i) \$320,000 principal amount of twelve percent (12%) unsecured six-month promissory note with a maturity date of the earlier of six months from the date of the note or the date the Company consummates a material business combination transaction (the "Note"); and (ii) a common stock purchase warrant to purchase 5,203,252 shares of the Company's common stock, par value \$0.001 per shares (the "Common Stock") at an exercise price of \$0.0923 per share (the "Warrant" and together with the Note, the "Unit"). Visionary PE GP I, LLC is the general partner of the Investor. Mr. Ronald Zamber, one of the Company's directors, is the Managing Director and Chairman of Visionary Private Equity Group I, LP, and the Manager of Visionary PE GP I, LLC. The sale by the Company to the Investor of the Unit, pursuant to the Securities Purchase Agreement is referred to herein as the "Private Placement." For more information, see the Company's Current Report on Form 8-K filed on February 7, 2017.

During the six months ended June 30, 2017, temporary capital advances totaling \$135,000 had been made by Visionary Private Equity Group I, LP. Mr. Ronald Zamber, one of the Company's directors, is the Managing Director and Chairman of Visionary Private Equity Group I, LP. These amounts are recorded in Accrued Liabilities - related parties.

Note 7 – Shareholders' Equity

Common stock

The Company estimates the fair value of employee stock options and warrants granted using the Black-Scholes Option Pricing Model. Key assumptions used to estimate the fair value of warrants and stock options include the exercise price of the award, the fair value of the Company's common stock on the date of grant, the expected warrant or option term, the risk free interest rate at the date of grant, the expected volatility and the expected annual dividend yield on the Company's common stock.

During the six months ended June 30, 2017 the Company issued 2,640,000 warrants to directors, officers and employees for 2016 services with an exercise price of \$0.06. Also during the six months ended June 30, 2017 the Company issued 1,170,000 warrants to purchase shares of common stock to Navitus at exercise prices ranging from \$0.06 - \$0.09. During the six months ended June 30, 2016 the Company issued 302,000 warrants to Navitus with an exercise price ranging from \$0.15 - \$0.21. These warrants to purchase shares of common stock were issued in consideration of capital contributions to Aurora pursuant to the Company's capital contribution agreement with Aurora. The warrants vest immediately and the Company valued the common stock warrants using the Black Scholes Option Pricing Model. The values for the three months ended June 30, 2017 and 2016 were \$24,929 and \$98,540, respectively, and were \$75,342 and \$154,374, for the six months ended June 30, 2017 and 2016, respectively.

Note 8 - Commitments and Contingencies

Contingencies

Liabilities and other contingencies are recognized upon determination of an exposure, which when analyzed indicates that it is both probable that an asset has been impaired or that a liability has been incurred and that the amount of such loss is reasonably estimable.

Volatility of Oil and Natural Gas Prices

Our revenues, future rate of growth, results of operations, financial condition and ability to borrow funds or obtain additional capital, as well as the carrying value of our properties, are substantially dependent upon prevailing prices of oil and natural gas.

Litigation

Legal Cases Settled

Cause No. 08-04-07047-CV; Oz Gas Corporation v. Remuda Operating Company, et al. v. Victory Energy Corporation.; In the 112th District Court of Crockett County, Texas.

Plaintiff Oz Gas Corporation ("Oz") filed a lawsuit in April 2008 against various parties for bad faith trespass, among other claims, regarding the drilling of two wells on lands that Oz claims title to. On November 18, 2009, Victory Energy Corporation intervened in the lawsuit to protect its fifty percent (50%) interest in one of the named wells in the lawsuit (that being the 155-2 well located on the Adams Baggett Ranch in Crockett County, Texas).

This case was mediated, with no settlement reached. It went to trial February 8-9, 2012. The Court found in favor of Oz and rendered verdict against Victory and the other Defendants, jointly and severally. Victory appealed this case to the 8th Court of Appeals in El Paso, Texas where the Court of Appeals affirmed the verdict of the District Court and Victory filed a Motion for Rehearing, which was denied. Victory filed a Petition for Review in the Supreme Court of Texas on December 15, 2014, which was denied. Victory filed a Motion for Rehearing with the Supreme Court which was denied. Oz then filed Interrogatories and Request for Production in Aid of Judgment, which were answered by Victory.

A Settlement and Forbearance Agreement was entered into on March 22, 2016, between the parties wherein no further post-judgment discovery or collection efforts will be made by Oz, for \$140,000 net of a \$14,000 payment received by the Oz receiver (see next following Cause No. C-1-CV-16-001610), with monthly payments of \$7,500 commencing April 15, 2016. The balance as of June 30, 2017 was \$12,500 and is included in Accrued Liabilities on the balance sheet.

Cause No. C-1-CV-16-001610; Oz Gas Corporation v. Victory Energy Corporation; In the County Court at Law No. 1 of Travis County, Texas.

Plaintiff Oz Gas Corporation ("Oz") filed an Application for Turnover Relief in Travis County, Texas on February 19, 2016. This order was granted and Thomas L. Kolker was appointed as Receiver to assist in the collection of non-exempt assets. Victory itself has not been placed into Receivership. Victory filed its Motion to Vacate the Turnover that was heard and denied by the trial court. Oz has since filed an Amended Application for Turnover Relief and Appointment of a Receiver to be heard March 10, 2016. Victory filed its Notice of Appeal March 4, 2016. A Settlement and Forbearance Agreement was entered into on March 22, 2016 as described above.

Cause No. D-1-GN-13-000044; Aurora Energy Partners and Victory Energy Corporation v. Crooked Oaks, LLC; In the 261st District Court of Travis County, Texas.

Victory Energy Corporation sued Crooked Oaks, LLC a/k/a Crooked Oak, LLC for breach of a purchase and sale agreement dated May 7, 2012 in which Victory sold certain assets to Crooked Oaks, LLC for \$400,000 of which only \$200,000 has been paid as of December 31, 2014. The lawsuit seeks to recover the remaining balance owed of \$200,000 from Crooked Oaks, LLC in addition to attorney's fees and all costs of court. Crooked Oaks, LLC has asserted a counterclaim for rescission of the underlying contract.

Victory and Crooked Oaks attended a mediation on February 10, 2016 where it was determined that Crooked Oaks was insolvent and since that date the case has been dismissed with prejudice.

Cause No. 50916; Trilogy Operating Inc. v. Aurora Energy Partners; In the 118th Judicial District Court of Howard County, Texas.

This lawsuit was filed on January 6, 2016. This lawsuit alleges causes of action for a suit on a sworn account, breach of contract and a suit to foreclose on liens regarding the drilling and completion of seven wells. Aurora filed an answer on January 29, 2016. Trilogy filed a Motion for Partial Summary Judgment on March 23, 2016.

The parties entered into a Settlement Agreement and Release on April 26, 2016, effective April 1, 2016 to dismiss the lawsuit with prejudice. The court granted the Joint Motion to Dismiss with Prejudice on May 2, 2016. In conjunction with the Joint Motion to Dismiss, Aurora assigned Trilogy all of its interests in the seven wells and related oil and gas leases.

Cause No. 2015-05280; TELA Garwood Limited, LP. v. Aurora Energy Partners, Victory Energy Corporation, Kenneth Hill, David McCall, Robert Miranda, Robert Grenley, Ronald Zamber, and Patrick Barry; In the 164th

District Court of Harris County, Texas.

This lawsuit was filed on January 30, 2015 and supplemented on March 4, 2015. This lawsuit alleges breach of contract regarding a Purchase and Sale Agreement that TELA Garwood Limited, LP and Aurora Energy Partners entered into on June 30, 2014. A first closing was held on June 30, 2014 and a purchase price adjustment payment was made on July 31, 2014. Between these two dates, Aurora paid TELA approximately \$3,050,133. A second closing was to take place in September, however several title defects were found to exist. The title defects could not be cured and a purchase price reduction could not be agreed upon by the parties in relation to the title defects, therefore, the second closing never took place. Aurora and Victory filed an answer and counterclaim in this case. Both parties filed opposing motions for summary judgment which were heard on April 14, 2016. The Court granted Aurora's partial motions for summary judgment dismissing claims against Aurora/Victory's officers and directors, including Kenny Hill, David McCall, Robert Grenley, Ronald Zamber, Patrick Barry, and Fred Smith. The Court denied the remaining summary judgment issues of both parties. On June 2, 2016 Aurora/Victory filed a second Motion for Partial Summary Judgment on some discrete contract interpretation issues. The Court denied this motion on September 2, 2016. On December 9, 2016, Aurora/Victory and TELA entered into a Mutual Release and Settlement Agreement in which Aurora agreed to pay TELA \$320,000 and in turn each Party agreed to release the other Party from any matter relating to the Purchase and Sale Agreement, the litigation or any claims that were or could have been brought in the litigation. In accordance with the Mutual Release and Settlement Agreement, Aurora made the full payment on February 1, 2017.

Cause No. 10-09-07213; Perry Howell, et al. v. Charles Gary Garlitz, et al.; In the 112th District Court of Crockett County, Texas.

The above referenced lawsuit was filed on or about September 6, 2010. This lawsuit alleges that Cambrian Management, Ltd. and Victory were trespassers on their land, and that they, along with other Defendants, drilled a well (115 #8) on land belonging to Plaintiffs. Plaintiffs claim trespass and unjust enrichment by certain Defendants because of the drilling of the 115 #8 well.

The Court placed this case on the Dismissal Docket asking any party to show cause as to why it should maintain this case on the docket on July 8, 2016. No party came forward stating why the case should be maintained and the Court entered and Order of Dismissal on August 9, 2016.

Legal Cases Pending

Cause No. CV-47230; James Capital Energy, LLC and Victory Energy Corporation v. Jim Dial, et al.; In the 142nd District Court of Midland County, Texas.

This is a lawsuit filed on or about January 19, 2010, by James Capital Energy, LLC and Victory Energy Corporation against numerous parties for fraud, fraudulent inducement, negligent misrepresentation, breach of contract, breach of fiduciary duty, trespass, conversion and a few other related causes of action. This lawsuit stems from an investment Victory made involving the purchase of six wells on the Adams Baggett Ranch with the right of first refusal on option acreage.

On December 9, 2010, Victory was granted an interlocutory Default Judgment against Defendants Jim Dial, 1st Texas Natural Gas Company, Inc., Universal Energy Resources, Inc., Grifco International, Inc., and Precision Drilling & Exploration, Inc. The total judgment amounted to approximately \$17,183,987.

Victory has added a few more parties to this lawsuit. Discovery is ongoing in this case and no trial date has been set at this time.

Victory believes it will prevail against all the remaining Defendants in this case.

On October 20, 2011, Defendant Remuda filed a Motion to Consolidate and a Counterclaim against Victory. Remuda is seeking to consolidate this case with two other cases wherein Remuda is the named Defendant. An objection to this motion was filed and the cases have not been consolidated. Additionally, we do not believe that the counterclaim made by Remuda has any legal merit.

Note 9 - Subsequent Events

During the period of July 1, 2017 through August 11, 2017, additional capital contributions from an affiliate of \$200,000 were received.

On July 25, 2017, Plaintiff Penn Virginia Oil and Gas, L.P. filed a lawsuit in the 295th Judicial Court of Harris County, Texas, of which Victory Energy Corporation is a defendant. This lawsuit alleges breach of contract and seeks to foreclose on its liens on the Eagle Ford wells in conjunction with collecting monetary damages.

PROXY CARD

VICTORY ENERGY CORPORATION C/O TRANSFER ONLINE, INC. 512 SE SALMON ST. PORTLAND, OR 97214

VOTE BY INTERNET - www.proxyvote.com

Use the Internet to transmit your voting instructions and for electronic delivery of information up until 11:59 p.m. Eastern Time the day before the cut-off date or meeting date. Have your proxy card in hand when you access the web site and follow the instructions to obtain your records and to create an electronic voting instruction form.

ELECTRONIC DELIVERY OF FUTURE PROXY MATERIALS

If you would like to reduce the costs incurred by our company in mailing proxy materials, you can consent to receiving all future proxy statements, proxy cards and annual reports electronically via e-mail or the Internet. To sign up for electronic delivery, please follow the instructions above to vote using the Internet and, when prompted, indicate that you agree to receive or access proxy materials electronically in future years.

VOTE BY PHONE - 1-800-690-6903

Use any touch-tone telephone to transmit your voting instructions up until 11:59 p.m. Eastern Time the day before the cut-off date or meeting date. Have your proxy card in hand when you call and then follow the instructions.

VOTE BY MAIL

Mark, sign and date your proxy card and return it in the postage-paid envelope we have provided or return it to Vote Processing, c/o Broadridge, 51 Mercedes Way, Edgewood, NY 11717.

TO VOTE, MARK BLOCKS BELOW IN BLUE OR BLACK INK AS FOLLOWS:

E33236-S63138

KEEP THIS PORTION FOR YOUR RECORDS

THIS PROXY CARD IS VALID ONLY WHEN SIGNED AND DATED.

DETACH AND RETURN THIS PORTION ONLY

Signature (Joint Owners)

Date

VICTORY ENERGY CORPORATION	For Withhold For All The Board of Directors recommends you vote FOR the All All Except following:
4.	
Election of Directors	
Nominees:	OS) DICADDO A CALAC
01) RONALD W. ZAMBER	· ·
nominee(s) on the line below.	or any individual nominee(s), mark "For All Except" and write the number(s) of the
!!!	
2)KEVIN DELEON 06) JUL	JO C. HERRERA
	C EILERTSEN
4)ROBERT GRENLEY	
-	nends you vote FOR proposals 1, 2, 3 and 5.
For Against Abstain	
For Against Abstain	
1.	
* *	restatement of the Company's Amended and Restated Articles of Incorporation to,
	e the number of authorized shares of the common stock,
	he Company from 47,500,000 shares to 300,000,000 shares and increase the number of
_	ed stock, \$0.001 par value per share, of the Company from 2,500,000 shares to lement a 1-for-38 reverse split of the Company's outstanding common stock.
!!	lement a 1-101-36 reverse spirt of the Company's outstanding common stock.
!	
3.	
5.	
To approve the Victory Energy	Corporation 2017 Equity Incentive Plan.
To approve the adjournment of	the special meeting for any purpose, including to solicit additional proxies if there are
insufficient votes at the time of	the special meeting to approve the proposals described above.
!	
!	
!!	
!!	
2.	may properly come before the meeting or any adjournment thereof.
	ll of the Company's partnership interests in Aurora Energy Partners, pursuant to the
	titure agreement, dated August 21, 2017, between the Company and Navitus Energy
Group, as amended.	
Place sion avestly as your name	as(a) amasu(a) hansan Whan sizning as attenuary arranton administraton an athen
	ne(s) appear(s) hereon. When signing as attorney, executor, administrator, or other as such. Joint owners should each sign personally. All holders must sign. If a
• 1	as such. Joint owners should each sight personany. An holders must sigh. If a use sigh in full corporate or partnership name by authorized officer.
Signature [PLEASE SIGN WI7	
Date	·· · — · _J



Important Notice Regarding the Availability of Proxy Materials for the Special Meeting: The Notice and Proxy Statement is available at www.proxyvote.com.

E33237-S63138

VICTORY ENERGY CORPORATION

Special Meeting of Stockholders November 14, 2017 10:30 AM This proxy is solicited by the Board of Directors

The undersigned stockholder of VICTORY ENERGY CORPORATION (the "Company") hereby acknowledges receipt of the Notice of Special Meeting of Stockholders and Proxy Statement, each dated September 28, 2017 and hereby appoints Company Secretary, Kenneth Hill, with full power of substitution, as proxy to vote all shares of the Company's common stock held by the undersigned at the Special Meeting of Stockholders of the Company to be held on November 14, 2017, and at any adjournments thereof, upon the proposals set forth in this proxy and described in the Notice of Special Meeting and Proxy Statement, and in his discretion with respect to such other matters as may be properly brought before the meeting or any adjournments thereof.

If this proxy is properly executed and returned, this proxy will be voted for the specifications made on the reverse side or if no direction is made, this proxy will be voted "FOR" the nominees for directors set forth on the reverse side as item 4, and "FOR" items 1, 2, 3 and 5.

Continued and to be signed on reverse side